Examining the Effects of Text Message and Competition-Based Interventions on Physical Activity, Self-Efficacy, Self-Regulation, Social Support, Incentive Motivation, and Mental Well-being among University Students

by

Chenyu Zou

A dissertation submitted to the Graduate Faculty of Auburn University in partial fulfillment of the requirements for the Degree of Doctor of Philosophy

> Auburn, Alabama August 5, 2023

Keywords: Physical activity, university student, text message, fitness competition, Social Cognitive Theory

Copyright 2023 by Chenyu Zou

Approved by

Brent I. Fox, Chair, Professor of Health Outcomes Research and Policy Salisa Westrick, Professor of Health Outcomes Research and Policy Kimberly Garza, Associate Professor of Health Outcomes Research and Policy William M. Murrah III, Associate Professor of Educational Foundations, Leadership, and Technology Joshua C. Hollingsworth, Assistant Professor of Pharmacology, Edward Via College of Osteopathic Medicine Auburn Campus

Abstract

Background: Compelling evidence highlights the importance of physical activity and its numerous benefits for both physical and mental well-being. However, physical activity levels remain low, particularly among university students. Several interventions have been designed and tested to target low physical activity levels among this audience, including text message interventions and fitness competitions.

Objective: The objectives of this study are: 1) To conduct a systematic review and metaanalysis of text message interventions targeting physical activity among university students from published literature, and 2) To assess and compare the effectiveness of text message interventions and a competition-based intervention in improving physical activity among university students, and evaluate participants' acceptability and experiences.

Methods: A structured, electronic search was conducted in October 2022, using PsycINFO, MEDLINE, SPORTDiscuss, CINAHL through the EBSCO interface and Web of Science, to select experimental studies focusing on text message interventions for physical activity. Studies in the same study design were included in several meta-analyses for different outcomes.

Second, students in three health professional programs were recruited in a three-group quasi-experimental study. The three groups included a control group ("Control"), a competition group ("DFFC), and a competition + text message group ("DFFC + text"). During the eight weeks of competition and four weeks of follow-up, participants were expected to complete two surveys (week-0 and week-8) and report their weekly moderate-to-vigorous physical activity (MVPA) minutes. Outcomes include Physical Activity, Mental Well-being, Self-Efficacy,

ii

Incentive Motivation, Social Support, and Self-Regulation. One-way ANOVA, Repeated Measurement ANOVA, Paired t-test, and mediation models were used to test the effects of the two interventions. The relationship between outcomes variables were assessed by path analysis.

Results: Among 20 studies included in the systematic review, 15 (75%) were randomized controlled trials (RCTs), and 10 (50%) studies focused on university students younger than 30 or 35 years old. In the university setting, texts were commonly sent on a daily basis for 2-6 weeks and were incorporated into various physical activity programs. The metaanalyses of RCTs indicated that text message interventions resulted in significantly greater total metabolic equivalent of task (MET) (n=3, SMD=0.67, p<0.001). However, for the analysis of intervention effects on Body Mass Index, statistically significant differences were not observed in RCTs (n=3, SMD=-0.15, p=0.08).

Weekly MVPA minutes in the DFFC Group and DFFC + Text Group were significantly different compared to the control group (week 0-8), but there was no difference in MVPA minutes between the DFFC Group and DFFC + Text Group. From week-4 to week-8, participants in the DFFC + Text Group demonstrated a statistically significant increase in weekly MVPA minutes compared with week-1. Results of Repeated Measures ANOVA indicated that from week-1 to week-8, the effect of the competition was significant on weekly MVPA (p<0.01) between the Control Group and two intervention groups, and the influence of texting depended on time (p=0.01). A statistically significant difference in post-intervention Incentive Motivation was observed among the three groups. The effect of interventions was mediated by Self-Efficacy and Incentive Motivation have a medium direct and significant effect on Physical Activity. Both interventions were found to be satisfactory and acceptable by participants.

Conclusions: Positive effects of text message interventions on university students' Physical Activity have been observed in the literature for MET but not for other outcomes. Text message interventions with well-controlled designs are needed to further examine the associations and identify characteristics of effective texts among university students. In the quasi-experimental study, both competition-based and text message interventions were found to be feasible and acceptable in promoting physical activity among university students. Future work should focus on developing multimodal interventions grounded in behavior theories, while also investigating the long-term effects and generalizability through robust study designs.

Acknowledgments

When I was planning to write this acknowledgement statement, I came across a tweet from Mr. Hugh Kearns, an internationally recognized public speaker, educator, and researcher, "Every day I am in awe of international students. Doing a PhD in your native language and home country is hard. Doing it in a non-native language, in a different country and different culture is so much harder. You are not stupid. You are legends." This heartfelt message resonates with many, garnering 12,600 likes as of June 20, 2023, including one from me. I am deeply touched by the understanding and empathy extended towards international students; in the meantime, I feel extremely fortunate to have received dedicated support from so many people along the way.

First and foremost, I am sincerely grateful to my advisor, Dr. Brent I. Fox, for his invaluable guidance and unwavering support. It is a great honor and privilege that I was given the opportunity to work under his supervision. I would like to thank him for his patience, knowledge, expertise, empathy, and great sense of humor. Next, I would also like to extend my appreciation to my dissertation committee members, Dr. Salisa Westrick, Dr. Kimberly Garza, Dr. William M. Murrah, and Dr. Joshua C. Hollingsworth for their valuable comments which helped in designing and developing this project, and to my University Reader, Dr. Robin Gosdin Farrell, for her time and efforts on my dissertation.

I want to thank Auburn University, Harrison College of Pharmacy, and the Department of Health Outcomes Research and Policy. I would like to extend my gratitude to all faculty, staff members, and students who supported me throughout my PhD journey. Also, I would like to thank Dr. Jingjing Qian and Mrs. Adelia Grabowsky for their help in the systematic review and meta-analysis. Likewise, I want to thank Dr. Surachat Ngorsuraches and Dr. Matthew Loop for

v

their advice on statistical analysis. I would also like to thank Dr. Yi Zhao for serving as the second reviewer and second coder. Overall, I thank everyone who participated in this study.

I must also thank my family for their constant love and encouragement. I wanted to thank my parents for their exceptional support in enabling me, their only child, to study abroad. I haven't been able to go back home for about three years, but their support in my pursuit of my dreams remains steadfast. I would like to acknowledge my boyfriend, Fan Wu. He is always supportive, willing to listen, and encourages me when I am struggling. Also, I must thank my beloved pets, Paper and Diudiu, for their antics and noise they made during those late nights when I worked on my dissertation.

Table of Contents

Abstractii
Acknowledgmentsv
Table of Contents vii
List of Tables xii
List of Figuresxiv
List of Abbreviationsxvi
Chapter 1. Introduction1
1.1 Background of the Problem1
1.1.1 Physical Activity1
1.1.2 Physical Activity Interventions
1.1.3 Deans' Fit Family Challenge
1.2 Problem Statement and Purpose of the Study
1.3 Specific Aims
1.4 Study Significance
1.5 Innovation
1.6 Organization of Dissertation10
Chapter 2. Literature Review
2.1 The Problem: Lack of Physical Activity 11
2.1.1 Definitions of Physical Activity and Recommended Physical Activity Levels
2.1.2 Benefits of Physical Activity

2.1.3 Risks and Costs of Physical Inactivity	
2.1.4 Physical Inactivity Among the U.S. Population	19
2.1.5 Physical Inactivity Among University Students	
2.2 A Potential Solution: Physical Activity Interventions	25
2.2.1 Competitions to Increase Physical Activity	
2.2.2 Mobile Text-Message Interventions	
2.2.3 Website-Delivered Interventions	
2.2.4 Community Physical Activity Campaigns	
2.2.5 University Campus-based Physical Activity programs	30
2.3 Motivation and Barriers of Physical Activity among University Students	
2.3.1 Motivation of Physical Activity among University Students	
2.3.2 Barriers to Physical Activity among University Students	
2.4 Social Cognitive Theory Framework	
2.5 Summary of the Literature Review	
Chapter 3. Methods	40
3.1 Specific Aim 1	40
3.1.7 Potential Limitations and Alternative Strategies	46
3.2 Specific Aim 2	47
3.2.1 Research Design / Experimental Approach	48
3.2.2 Participants and Settings	49

3.2.3 Interventions	51
3.2.4 Measures and Data Collection	55
3.2.5 Statistical Analysis	61
3.2.6 Expected Findings	65
3.2.7 Potential Limitations and Alternative Strategies	65
Chapter 4. Results	67
4.1 Aim 1: Systematic Review and Meta-Analysis	67
4.1.1 Systematic Review	67
4.1.2 Meta-analysis	
4.2 Aim 2.1	
4.2.1 Study Enrollment	
4.2.2 Descriptive Results	
4.2.3 Path Analysis for Pretest Survey Data	85
4.2.4 Effects of the DFFC and Text Message Interventions on PA	
4.2.5 Effects of the DFFC and Text Message Interventions on Mental Well-bein	ng, Incentive
Motivation, Social Support, Self-Efficacy, and Self-Regulation.	
4.3 Aim 2.2	
4.3.1. Feedback and Satisfaction in DFFC	
4.3.2 Feedback and Satisfaction of Text Message Interventions	101
4.4 Summary	

Chapter 5. Discussion	
5.1 Aim 1	
5.1.1 Discussion of Systematic Review and Meta-Analysis Findings	
5.1.2 Quality and Bias Assessment	
5.1.3 Implications and Future Directions	108
5.1.4 Strengths	
5.1.5 Limitations	
5.2 Aim 2	
5.2.1 Effects, Experience, and Satisfaction with the DFFC	
5.2.2 Effects, Experience, and Satisfaction with Text Message	
5.2.3 Roles of Incentive Motivation and Self-Efficacy	
5.2.4 Implications and Future Directions	
5.2.5 Strengths	
5.2.6 Limitations	
5.3 Implications and Conclusions	
References	
Appendix 1. 1st DFFC Email for Students	
Appendix 2. 2nd DFFC Email for Students	151
Appendix 3. Sign-up Surveys for Control Group	155
Appendix 4. 3rd DFFC Email for Students	
Appendix 5. Sign-up Surveys for DFFC and DFFC + Text Group	

Appendix 6. 20 Texts for DFFC + Text Group	. 170
Appendix 7. Week-0 Survey for All Three Groups	. 172
Appendix 8. Week-8 Survey for Control Group	. 184
Appendix 9. Week-8 Survey for DFFC Group	. 192
Appendix 10. Week-8 Survey for DFFC + Text Group	. 200
Appendix 11. Week -0 Contact Information Survey	. 209
Appendix 12. Week -8 Contact Information Survey	. 210
Appendix 13. Referral List for Mental Wellbeing	.211

List of Tables

Table 2.1. Different Aerobic Physical Activities	12
Table 2.2. Levels of Physical Activity	13
Table 2.3. Examples of Community Events in the U.S	30
Table 2.4. Examples of Campus-based Physical Activity Programs	31
Table 2.5. Barriers to Physical Activity Among University Students	33
Table 2.6. Social Cognitive Theory Constructs	35
Table 3.1. Population, Intervention, Comparator, Outcome, and Study Design (PICOS) Cr	iteria
for the Systematic Reviews and Meta-Analysis	41
Table 3.2. Standardized Article Screening Form for Systematic Review	43
Table 3.3. The SCT Components of the Deans' Fit Family Challenge	53
Table 3.4. Text Message Schedule and Examples Based on SCT Components	55
Table 3.5. Outcomes, Measurement, and Data Collection	56
Table 3.6. ChallengeRunner Trackers and Activities	57
Table 3.7. Summary of Data Analyses Plan	64
Table 4.1. Characteristics of Included Studies	69
Table 4.2. Categories of Outcomes for 20 Included Studies	75
Table 4.3. Article Quality Assessment	80
Table 4.4. Response Rate for Two Surveys	83
Table 4.5. Participants' Baseline Characteristics	85
Table 4.6. Pre-test Model Fit Information	86
Table 4.7. Standardized Total and Direct Effects of Pre-Test Model	87
Table 4.8. Standardized Indirect Effects of Pre-Test Model	

Table 4.9. Repeated Measures ANOVA for Between Subjects Effects (DFFC and text, week 1-8)
Table 4.10. Repeated Measures ANOVA for Within Subject Effects (Time, week 1-8)90
Table 4.11. Repeated Measures ANOVA for Between Subjects Effects (DFFC and text, week 1-
12)91
Table 4.12. Repeated Measures ANOVA for Within Subject Effects (Time, week 1-12)91
Table 4.13. Weekly MVPA Minutes Weeks 0-12 93
Table 4.14. Scores for Mental Wellbeing and Social Cognitive Outcomes After Intervention94
Table 4.15. One-way ANOVA with Multiple Comparisons for Mental Wellbeing and Social
Cognitive Outcomes95
Table 4.16. Excerpts from Open-Ended Questions 99
Table 5.1. Effective Strategies for Text Messaging University Students for PA Purposes

List of Figures

Figure 1.1 Number of DFFC Participants 2019-2021
Figure 1.2 Average Activity Minutes Per Participant
Figure 2.1 Relationship of Moderate-to-Vigorous Physical Activity to All-Cause Mortality 15
Figure 2.2. Percent of U.S. Adults Meeting Aerobic and Muscle Strengthening Guidelines 2011-
2019
Figure 2.3. Percent of Alabama Adults Meeting Aerobic and Muscle Strengthening Guidelines
2011-2019
Figure 2.4. Percent of U.S. Adults Who Engage in No Leisure-Time Physical Activity 2011-
2020
Figure 2.5. Percent of Alabama Adults Who Engage in No Leisure-Time Physical Activity 2011-
2020
Figure 2.6. Prevalence of Self-Reported Physical Inactivity Among US Adults by State and
Territory, BRFSS, 2017-2020
Figure 2.7. Social Cognitive Theory
Figure 3.1. Example of Symmetrical Funnel Plot (Fixed-Effect Meta-Analysis Model)46
Figure 3.2. Example of Funnel Plot for the Random-Effects Meta-Analysis
Figure 3.3. Timeline of Interventions
Figure 3.4. Process of Collecting MVPA Minutes Via the ChallengeRunner App
Figure 3.5 Path Analysis of the Effect of Interventions on Weekly MVPA Minutes, Mental
Wellbeing, Self-Efficacy, Self-Regulation, Incentive Motivation, and Social Support62
Figure 4.1. Study Identification and Selection following the PRISMA Guidelines67
Figure 4.2. Overall Characteristics of 20 Articles in Systematic Review

Figure 4.3. Impact of Text Message Interventions on weekly total MET (RCTs, n=3)	82
Figure 4.4. Impact of Text Message Interventions on BMI (RCTs, n=3)	83
Figure 4.5. Path Analysis of the Effect of Framing on Self-Regulation, Incentive Motivation,	
Self-Efficacy, Social Support, and Physical Activity (Pre-test Model)	86
Figure 4.6. Weekly MVPA Minutes Weeks 0-12	89
Figure 4.7. Mediation Model for Mental Wellbeing	96
Figure 4.8. Mediation Model for Self-Efficacy	96
Figure 4.9. Mediation Model for Incentive Motivation	96
Figure 4.10. Mediation Model for Self-Regulation	96
Figure 4.11. Mediation Model for Social Support	96
Figure 4.12. Motivators of Exercise Among DFFC Participants	97
Figure 4.13. Barriers of Exercise Among DFFC Participants	98
Figure 4.14. Overall Satisfaction of DFFC1	.01
Figure 4.15. Intentions to Participant in 2023 DFFC1	.01
Figure 4.16. Acceptability of Text Message Interventions1	.02
Figure 4.17. Overall Satisfaction of Text Message Interventions	.02

List of Abbreviations

ACSM	American College of Sports Medicine
AUCON	College of Nursing
BRFSS	Behavioral Risk Factor Surveillance System
CDC	Centers for Disease Control and Prevention
CINAHL	Cumulative Index to Nursing and Allied Health Literature
CI	Confidence Interval
DFFC	Deans' Fit Family Challenge
НСОР	Harrison College of Pharmacy
IPAQ	International Physical Activity Questionnaire
MEDLINE	National Library of Medicine's Bibliographic Database/Article Index
MET	Metabolic Equivalent of Task
MMAT	Mixed Methods Appraisal Tool
MPA	Moderate-Intensity Physical Activity
MVPA	Moderate-to-Vigorous Physical Activity
NWU	Nightly-Week-U
OSPAQ	Occupational Sitting and Physical Activity Questionnaire
PA	Physical Activity
PASB-Q	Physical Activity and Sedentary Behavior Questionnaire
PAR	Physical Activity Recall
PICOS	Population, Intervention, Comparator, Outcome, and Study Design
PRISMA	Preferred Reporting Items for Systematic reviews and Meta-Analyses

PTSD	Post-Traumatic Stress Disorder
RMSEA	Root Mean Square Error of Approximation
SCT	Social Cognitive Theory
SLIPA	Sedentary and Light Intensity Physical Activity
SMD	Standardized Mean Difference
SMS	Short Message Service
U.S.	the United States
VCOM	Edward Via College of Osteopathic Medicine
VPA	Vigorous-Intensity Physical Activity
WHO	World Health Organization
WHO-5	World Health Organization Five Wellbeing Index

Chapter 1. Introduction

1.1 Background of the Problem

1.1.1 Physical Activity

Physical activity (PA) was defined as the "best buy" in public health by the epidemiologist Jerry Morris in 1994.¹ It is estimated that one in ten premature deaths can be prevented by getting enough physical activity.² The health benefits of physical activity are well documented, including a reduced risk of several chronic conditions such as cardiovascular disease, stroke, arthritis, hypertension, type 2 diabetes, and cancers.³⁻⁹ By engaging in regular physical activity, significant improvement can be achieved in weight management, muscular fitness, cognitive function, immune system function, mental well-being, quality of life, and life expectancy.^{2,10-15} As Dr. Ruth Petersen, the Director of the Division of Nutrition, Physical Activity, and Obesity from the Centers for Disease Control and Prevention (CDC) says, "if you could package physical activity into a pill, it would be the most effective drug on the market".²

Based on the second edition of the Physical Activity Guidelines for Americans, adults are recommended to perform at least 150 minutes of moderate-intensity or 75 minutes of vigorousintensity physical activity per week, or "an equivalent combination of moderate and vigorousintensity aerobic activity," as well as muscle-strengthening activities two or more days in a week.¹⁶ According to the CDC, physical inactivity in adults refers to the absence of participation in any form of leisure-time physical activities (e.g., running, walking for exercise, or gardening) during the past month.¹⁷ The overall prevalence of physical inactivity is 25.3% in United States (U.S.) population, and it is 30.7% in Alabamian adults.¹⁵ Globally, 40%-50% of university students are physically inactive.¹⁸⁻²⁴

1.1.2 Physical Activity Interventions

In light of the numerous benefits of physical activity and the prevalence of physical inactivity, effective interventions are needed to encourage the adoption and maintenance of active lifestyles in the U.S. population, particularly for university students who are at the age where sedentary behaviors are adopted.²⁵ Evidence suggests that various intervention strategies have been successful in promoting physical activity among different populations.²⁵⁻³¹ Fitness competitions, which use rankings, leader boards, status markers, incentives, and other competitive strategies, are widely implemented in working places, schools, and long-term care facilities.³²⁻³⁵ Text messaging is well suited for public health interventions because of its popularity, enormous reach, low cost, and effectiveness in promoting health outcomes.³⁶⁻³⁹ Through media coverage and promotion, educational programs, social support, and policy initiatives, community-wide campaigns are effective in increasing physical activity.^{31,40-42} Institutions of higher education, such as universities and colleges, are ideal for implementing health promotion programs to engage large numbers of participants.^{24,43,44}

1.1.3 Deans' Fit Family Challenge

To encourage students to be physically active, Auburn University holds the Deans' Fit Family Challenge (DFFC) each fall semester between the Harrison College of Pharmacy (HCOP), College of Nursing (AUCON), and Edward Via College of Osteopathic Medicine – Auburn (VCOM-Auburn). The DFFC is a 9-week annual health and wellness competition based on two metrics: cumulative minutes of exercise logged and cumulative number of total pounds lost from baseline to endpoint. Participants compete in teams of two within their college. Each team tracks minutes of meaningful activity using the ChallengeRunner app or website (challengerunner.com), submitting data weekly. Participants also provide self-reported weight

prior to and at the conclusion of the DFFC. Participants receive approximately weekly emails containing general information on physical activity and quality of life, an announcement of weekly winners for minutes exercised, and cumulative minutes throughout the DFFC. At the conclusion of the DFFC, individual winners are determined based on minutes of physical activity and weight loss. The winning college is determined based on a composite score that includes minutes of physical activity and weight loss.

Since 2019, 892 students, faculty, and staff have participated in the DFFC: 101 (11.39%) from AUSCON, 229 (25.82%) from VCOM, and 557 (62.80%) from HCOP (Figure 1.1). The average number of minutes logged per participant for each year was 1,458 in 2019, 2,027 in 2020, and 2,309 in 2021 (Figure 1.2). Since 2019, 278 participants reported losing 1,570 pounds, with 37 participants losing more than 10 pounds. VCOM won the DFFC in 2019, with an average of 30.38 activity hours and an average weight loss of 5.19 pounds per participant, and also won in 2021 with an average of 42.1 activity hours plus an average weight loss of 4.3 pounds per participant. In 2020, AUCON was the winning program with an average of 35.43 activity hours plus 2.6 pounds average weight loss per participant. Since 2019, an average of 93.3% of HCOP participants reported minutes in week 1, compared to 51% in week 9. Similarly, in 2021, 178 HCOP participants provided weigh-in data while 57 provided weigh-out data.



Figure 1.1 Number of DFFC Participants 2019-2021

Figure 1.2 Average Activity Minutes Per Participant



The DFFC plays an important role in promoting physical activity among participating programs. However, participation wanes in minutes and weight reported each year, and the implementation of the DFFC has not been systematically evaluated. Motivational and retention

strategies targeting social cognitive determinants, such as self-efficacy, self-regulation, incentive motivation, and social support should be explored to enhance future design and execution of competition-based interventions among health professional students, faculty, and staff members.

1.2 Problem Statement and Purpose of the Study

Physical inactivity is the fourth leading cause of death globally.⁹ However, participation in physical activity among the U.S. population is declining. Only 23.0% of U.S. adults meet the physical activity guidelines, while 26.0% of U.S. adults do not engage in any leisure-time physical activity.¹⁵ Despite the importance of physical activity, there is a significant drop in physical activity participation and an increase in sedentary behavior among university students.⁴⁵ For university students, 30%-35% have overweight or obesity, and 40%-50% are physically inactive.¹⁸⁻²⁴ As a result, strategies to promote healthy lifestyles in university students are critical.²⁶ A tailored intervention that utilizes evidence-based solutions may be effective in addressing physical inactivity in this population.

The overarching objective of the proposed research is to assess and compare how text message interventions and a competition-based intervention impact university students' physical activity levels, mental well-being, and Social Cognitive Theory constructs. We hypothesize that both the text message intervention and the competition-based intervention will positively influence university students' Physical Activity, Self-Efficacy, Self-Regulation, Social Support, Incentive Motivation, and Mental Well-being. The first research question is, "What is the relevant evidence related to text message interventions targeting physical activity among university students?". The second research question is, "What are the effects of text message interventions and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental well-

being?" The third research question is, "What are participants' perceptions regarding their experience with and acceptability of text message interventions and competition-based interventions?"

1.3 Specific Aims

Specific Aim 1: To conduct a systematic review and meta-analysis of text message interventions targeting physical activity among university students from published literature.

This aim is to identify physical activity outcomes relevant to text message interventions among university students. A systematic review and meta-analysis were used to accomplish this purpose. We hypothesized that text message interventions are effective in improving physical activity among university students. Outcomes of physical activity relevant to university students were identified from the review, including physical activity scores, moderate-intensity physical activity (MPA) minutes, vigorous-intensity physical activity (VPA) minutes, moderate-tovigorous physical activity (MVPA) minutes, walking minutes, weekly steps, and metabolic equivalent of task (MET). Five databases (APA PsycINFO, MEDLINE, SPORTDiscuss, CINAHL, Web of Science) were searched; the search ending with August 2022 publications. Additional articles were identified via a high sensitivity search on Google Scholar and handsearching. Eligibility criteria included interventional studies exploring the effectiveness of university students participating in text messaging physical activity interventions. The Mixed Methods Appraisal Tool (MMAT), Version 2018,^{46,47} was used to assess each article's quality concurrently with data extraction. Publication bias was presented using funnel plot asymmetry tests.⁴⁸ Meta-analysis was performed if at least three studies reported the same physical activity effectiveness outcomes.^{49,50}

Specific Aim 2: To assess and compare the effectiveness of text message interventions and a competition-based intervention in improving physical activity among university students, and evaluate participants' acceptability and experiences of the text message interventions and the competition-based intervention.

Sub Aim 2.1: To conduct a quasi-experimental study to assess and compare the effectiveness of text message interventions and a competition-based intervention among university students.

Sub Aim 2.2: To evaluate participants' acceptability and experiences of text message interventions and the competition-based intervention.

A quasi-experimental design was employed to accomplish this aim. Eligibility criteria of participants included being an adult university student (≥18 years of age), the ability to speak and write English, and possessing a smartphone and a fitness tracker or smartwatch. During the eight weeks competition and four weeks of follow-up, they were expected to report their weekly MVPA via supporting apps to synchronize activity minutes to the ChallengeRunner app or manual data entry. Participants in the 2022 Deans' Fit Family Challenge competition were invited and randomly assigned to the competition intervention group or the competition plus text message intervention group. Students in HCOP, AUCON, and VCOM who did not enroll in the 2022 DFFC were invited as the Control group. Participants in the Control Group did not participate in 2022 DFFC nor receive text messages.

Repeated measures ANOVA was used to examine the effect of the two interventions. We also assessed if these parameters (mental well-being, self-efficacy, incentive motivation, self-regulation, and social support) influenced their PA. A path analysis was used to examine the

relationship between Physical Activity, Mental Well-being, Self-Efficacy, Incentive Motivation, Self-Regulation, and Social Support. The hypothesized relationships were informed by the Social Cognitive Theory. The independent variable was physical activity interventions (e.g., DFFC or DFFC plus text messaging) while the dependent variables were PA levels, Mental Well-being, Self-Efficacy, Incentive Motivation, Self-Regulation, and Social Support. A qualitative thematic analysis approach was used to analyze open-ended questions about participants' acceptability and experiences.

1.4 Study Significance

With health behaviors still under development during young adulthood, increased physical activity and improved health awareness help prevent vital health problems. However, there is limited literature on the health behavior of university students. This study has the potential to contribute to knowledge about this cohort. A systematic review of experiences of university students' physical activity will reveal the remarkable impact of physical activity engagement on individuals of this group. Also, by targeting the intervention in this study to university students, findings are expected to reveal the effects of competition-based physical activity interventions and cognitive variables on the physical activity of young adults. Results may be adapted for interventions for university students adopting other health behaviors.

Evidence has reported positive effects of text message interventions on promoting physical activity among various populations, including university students. However, there is no systematic review and meta-analysis focusing on the effect of text messaging on physical activity among university students. A systematic review and meta-analysis to synthesize the findings will qualitatively and quantitatively describe the effect of text messaging on physical activity among university students. This study has the potential to improve university students' physical activity levels through the use of text messaging and a fitness competition. Even though fitness competition interventions have been found to be effective in promoting physical activity, they have not yet been compared with other interventions. Therefore, this study addresses a critical gap in literature.

There has been inconsistency in the field of health decision-making concerning the most effective types of physical activity interventions. Additionally, the role of fitness competitions in promoting physical activity among university students has not been studied. The outcome of this study yields valuable insight into implementing fitness competitions to promote physical activity, which may be adapted to school-based programs such as fitness campaigns and the general presentation of physical activity information to young adults.

Finally, although text messaging and competition-based physical activity interventions are widely implemented in campus settings, very few text message and competition-based physical activity strategies among university students are based on health behavior theory. In this study, both interventions are informed by the SCT model, which is an innovative approach.

1.5 Innovation

This study is innovative in five respects. First, data among university students *using text messaging* to promote physical activity is scarce. Aim 1 synthesizes the literature on those experiences to identify physical activity outcomes relevant to university students. Second, very few text messaging and competition-based physical activity interventions targeting university students are implemented based on health behavior theory. Third, there is a scarcity of research studies that specifically target university students, encompassing undergraduate, graduate, and professional students, with a majority of campus PA interventions primarily targeting

undergraduate students. This gap highlights the need for further research in understanding the comparative effectiveness of these different PA promotion approaches.

1.6 Organization of Dissertation

This dissertation consists of five chapters. The first chapter introduces the background of the problem, including the study's theoretical background. It also briefly describes the specific aims, methods, significance, and innovation. Chapter Two provides a comprehensive review of the literature and the theoretical framework. Chapter Three introduces the research approach for each aim, including the study design, setting, participant recruitment, data collection, outcomes, measures, and analysis. It also describes the potential challenges, limitations, and timeline of the study. Chapter Four reports the results of data analyses. Chapter Five includes a summary of results, a discussion of the findings, and implications for future research.

Chapter 2. Literature Review

2.1 The Problem: Lack of Physical Activity

2.1.1 Definitions of Physical Activity and Recommended Physical Activity Levels

According to the World Health Organization (WHO), physical activity is defined as any bodily movement produced by skeletal muscles that require energy expenditure,⁵¹ while exercise is a form of physical activity that is planned, structured, repetitive, and performed intending to improve fitness.^{52,53} Physical activity includes all movement that occurs during one's leisure time, travel, and occupation, or education.^{52,53} Exercise that involves large muscles moving rhythmically for a sustained period along with a high heart rate, such as hiking, running, and bicycling is known as aerobic exercise or endurance activity.⁵⁴⁻⁵⁶ Resistance training, also known as muscle strengthening, involves large muscle groups moving against an applied force, such as heavy objects, resistance bands, or an individual's own weight.⁵⁷⁻⁵⁹

The metabolic equivalent of task (MET) is a unit of energy expenditure related to a specific activity.⁶⁰ According to the definition, a MET is the "ratio of the rate of energy expended during an activity to the rate of energy expended at rest".⁵⁹⁻⁶² One MET is equivalent to the rate of energy expenditure during rest, while 4 MET activities expend four times the energy used by the body at rest. Performing a 5 MET activity for 30 minutes is equal to 5 x 30 = 150 MET-minutes (or 2.5 MET-hours) of physical activity. Light-intensity physical activities are defined as non-sedentary behaviors with less than 3.0 METs and reaching 57% - 63% of maximum heart rate, such as cooking, light household chores, and walking at 2.0 miles per hour (2.5 METs); moderate-intensity physical activities (MPA) are 3.0-5.9 METs with 64% - 75% of maximum heart rate, such as walking at 3.0 miles per hour (3.5 METs); vigorous-intensity

physical activities (VPA) are greater than 6.0 METs with 76% - 95% of maximum heart rate, such as running at 6.0 MPH (10 METs). ^{59,60,62-64} Moderate-to-vigorous physical activity (MVPA) is any activity over 3 METs with at least 64% of maximum heart rate.⁶⁴ Table 2.1 contains examples of moderate-intensity activities and vigorous-intensity activities.⁶²

Table 2.1. Different Aerobic Physical Activities⁶²

Moderate-Intensity Activities (3.0-5.9 METs with 64% - 75% of maximum heart rate)	Vigorous-Intensity Activities (> 6.0 METs with 76% - 95% of maximum heart rate)
 With 64% - 75% of maximum heart rate) Walking briskly (2.5 miles per hour or faster) Recreational swimming Bicycling slower than 10 miles per hour on level terrain Tennis (doubles) Active forms of yoga (for example, Vinyasa or power yoga) Ballroom or line dancing General yard work and home repair work Exercise classes like water aerobics 	 Jogging or running Swimming laps Tennis (singles) Vigorous dancing Bicycling faster than 10 miles per hour Jumping rope Heavy yard work (digging or shoveling, with heart rate increases) Hiking uphill or with a heavy backpack High-intensity interval training Exercise classes like vigorous step
 Bicycling slower than 10 miles per hour on level terrain Tennis (doubles) Active forms of yoga (for example, Vinyasa or power yoga) Ballroom or line dancing General yard work and home repair work Exercise classes like water aerobics 	 Vigorous dancing Bicycling faster than 10 miles per hour Jumping rope Heavy yard work (digging or shoveling, with heart rate increases) Hiking uphill or with a heavy backpack High-intensity interval training Exercise classes like vigorous step aerobics or kickboxing

Physical activity recommendations have been developed over the last several decades and have varied among organizations. The CDC and the American College of Sports Medicine (ACSM) published suggestions in 1995 that U.S. adults need to accumulate at least 30 minutes of MPA on most days.⁶⁵ This recommendation was endorsed by the 2005 Dietary Recommendations for Americans and the 2015 U.S. Public Health Service's Surgeon General's Report.^{66,67} In addition, the Institute of Medicine published recommendations addressing both dietary intake and physical activity to maintain a healthy weight and avoid undesirable fat accumulation , suggesting that adults should obtain at least 60 minutes of accumulated moderately intense physical activity each day.⁶⁸

In 2007, the ACSM and American Heart Association recommended a combination of both aerobic and muscle strengthening activities to improve and maintain health.⁶⁹ The ACSM indicated that healthy adults need at least 30 minutes of aerobic moderate-intensity physical activity five times per week or 20 minutes of aerobic vigorous-intensity physical activity three times per week.⁶⁹ The 2008 Physical Activity Guidelines for Americans, published by the U.S. Department of Health and Human Services adopted this recommendation for U.S. adults.⁵⁹ In 2018, the U.S. Department of Health and Human Services published the second edition of the Physical Activity Guidelines for Americans,⁶² suggesting that adults should perform at least 150 minutes moderate-intensity or 75 minutes vigorous-intensity physical activity per week, or "an equivalent combination of moderate and vigorous-intensity aerobic activity," as well as musclestrengthening activities on two or more days in a week

According to the second edition of the Physical Activity Guidelines for Americans, there are four levels of aerobic physical activity: inactive, insufficiently active, active, and highly active (Table 2.2).⁶²

Inactive	Not getting any moderate- or vigorous- intensity physical activity beyond basic movement for
	daily life activities.
Insufficiently	Doing some moderate- or vigorous- intensity physical activity but less than the recommended
active	criteria. This level is below the recommended guidelines for adults.
Active	Doing the equivalent of 150 minutes to 300 minutes of moderate-intensity physical activity a
	week. This level meets the critical guideline target range for adults.
Highly active	Doing the equivalent of more than 300 minutes of moderate-intensity physical activity a week.
	This level exceeds the critical guideline target range for adults.

Table 2.2. Levels of Physical Activity⁶²

2.1.2 Benefits of Physical Activity

There is ample evidence to support that physical activity is beneficial among adults. An inverse correlation has been identified between the volume of physical activity and all-cause

mortality. Literature also indicates that regular physical activity decreases the risk of many health concerns, including cardiovascular diseases, type 2 diabetes, osteoporosis, obesity, cancer, and falls in older adults. Additionally, physical activity improves cognitive function, quality of life, and mental well-being. Each of these findings is addressed below.

2.1.2.1 Lower Risk of All-Cause Mortality

Compelling evidence demonstrates that physical activity delays death from all causes. People who exercise about 150 minutes per week have a 20%-30% lower risk of all-cause mortality than physically inactive individuals (Figure 2.1).⁷⁰⁻⁷² Compared with people who did not meet the physical activity recommendation, those who engaged in recommended muscle strengthening activity (hazard ratio 0.89; 95% CI, 0.85 to 0.94) or aerobic activity (hazard ratio 0.71; 95% CI, 0.69 to 0.72) were at reduced risk of all-cause mortality; and even larger survival benefits were found in those engaged in both activities (hazard ratio 0.60; 95% CI, 0.57 to 0.62) (Figure 2.2).^{73,74} Compared with being inactive, achievement of activity levels that approximate the recommendations for moderate activity or vigorous exercise was associated with a 27% (RR, 0.73; 95% confidence interval [CI], 0.68-0.78) and 32% (RR, 0.68; 95% CI, 0.64-0.73) decreased mortality risk, respectively.⁷⁵ Physical activity reflective of meeting both recommendations was related to substantially decreased mortality risk overall (RR, 0.50; 95% CI, 0.46-0.54).⁷⁵ A 2011 study found that a person's risk of death from any cause decreased by 4% for every additional 15 minutes of physical activity up to 100 minutes per day throughout the study.¹⁰ Throughout the world, physical activity has helped reduce premature mortality.⁷⁶



Figure 2.1. Relationship of Moderate-to-Vigorous Physical Activity to All-Cause Mortality⁷²

Data were derived from 2018 Physical Activity Guidelines Advisory Committee. 2018 Physical Activity Guidelines Advisory Committee Scientific Report. Washington, DC: U.S. Department of Health and Human Services, 2018. The data analysis and conclusions in this study are those of the author and do not necessarily represent the official position of the Centers for Disease Control

2.1.2.2 Extend Expected Lifespan

Physical activity increases longevity.^{77,78} Active adults tend to have longer telomeres (a repetitive DNA sequence to protect the end of a chromosome from being frayed or tangled), resulting in fewer years of cellular aging than their more sedentary counterparts.⁷⁹⁻⁸¹ Evidence from eight cohort studies indicated that regular physical activity is associated with a 0.4-6.9 years longer life expectancy.⁸² A prospective cohort study of 416,175 individuals found that

people who performed 15 minutes a day or 92 minutes of physical activity per week extended their expected lifespan by three years compared to inactive people. An additional four years of life expectancy was gained by exercising for 30 minutes a day.¹⁰

2.1.2.3 Lower Risk of Cardiovascular Disease

There is an association between physical activity and lower rates of cardiovascular disease. An increase from being inactive to achieving recommended physical activity levels was associated with lower risk of cardiovascular disease mortality and incidence by 23% (RR, 0.77; 95% CI, 0.71–0.84) and 17% (RR, 0.83; 95% CI, 0.77–0.89), respectively.⁸³ The greatest physical activity levels were associated with a significantly reduced risk of heart failure (pooled HR, 0.70; 95% CI, 0.67–0.73).⁸⁴ Compared with people reporting no leisure-time physical activity, individuals engaged in recommended physical activity had reductions in heart failure risk (pooled HR, 0.90; 95% CI, 0.87-0.92).⁷¹

2.1.2.4 Lower Risk of Hypertension

Clinical trials have confirmed the favorable effects of exercise on blood pressure reduction. Among people with prehypertension, the impact of aerobic exercise training on blood pressure was significant for systolic and diastolic blood pressure, with net reductions of 4.3 mmHg and 1.7 mmHg reported, respectively.⁸ Cornelissen and colleagues investigated significant pooled reductions in daytime blood pressure with net reductions of 2.2 mmHg for systolic blood pressure and 3.3 mmHg for diastolic blood pressure.⁸⁵ Findings from the ASUKI Step study suggested the goal of walking 10,000 steps/day could be effective in reducing blood pressure.⁸⁶

2.1.2.5 Lower Risk of Type 2 Diabetes

There is an association between higher levels of physical activity and a 20%-30% reduction in diabetes risk.⁷ In men, vigorous physical activity is associated with a 40%-50% reduction in risk of diabetes.⁸⁷ An increase from being inactive to achieving recommended physical activity levels was associated with a lower risk of type 2 diabetes incidence by 26% (RR, 0.74; 95% CI, 0.72-0.77).⁸³ People who reported >10.4 MET-hours per week had a relative risk of diabetes of 0.59 (95% CI, 0.46-0.75) compared with those who were sedentary (\leq 2 MET-hours per week).⁸⁸ The benefit of physical activity in preventing type 2 diabetes are most remarkable in those at highest risk of the disease.⁸⁹

2.1.2.6 Lower Risk of Cancer Recurrence

Research shows that physical activity is safe and helpful for most people before, during, and after cancer treatment.³⁻⁶ Physical activity has positive effects on reduced risk of cancer recurrence, physical functions, psychological outcomes, and patients' quality of life after cancer treatment, including breast cancer,^{5,6} colorectal cancer,⁹⁰ prostate cancer,⁹¹ colon cancer,⁹² endometrial cancer,⁹³ esophagogastric cancer,⁹⁴ and lung cancer.⁹⁵

2.1.2.7 Improved Cognition

Physical activity can prevent or delay the onset of age-related cognitive impairment or dementia.^{96,97} Studies provide evidence for physical activity to be associated with a modest reduction in relative risk of cognitive decline.^{16,98} Physical activity is also a potent gene modulator that induces structural and functional changes in the brain.¹¹

2.1.2.8 Improved Quality of Life

There is an association between physical activity and improved quality of life.¹² Higher levels of PA are associated with improved quality of life (regression coefficient: 0.026-0.072).⁹⁹

Evidence also demonstrates that physical activity improves well-being.^{11,35,100} Higher levels of satisfaction and lower impact of COVID-19 on quality of life were related to higher total, vigorous, and moderate physical activity levels.¹⁰¹

2.1.2.9 Improved Mental Well-being

It has been found that physical activity contributes to higher levels of norepinephrine and endorphin in the brain, boosting people's spirits and creating feelings of happiness and levity.^{13,14} Therefore, physical activity has been associated with better mental health and emotional wellbeing, improved sleep, memory, and mood, and lower rates of mental illness, such as anxiety, depression, and post-traumatic stress disorder (PTSD).¹⁰²⁻¹⁰⁵ Evidence shows that regular exercise can improve self-esteem, moderate social withdrawal, boost the immune system, and reduce the impact of stress.^{100,106,107} For people with mental health diseases, physical activity boosts patients' mood, concentration, and alertness.¹⁰⁸⁻¹¹⁰ Researchers suggest that in addition to antidepressants and cognitive behavioral therapy, physical activity is an effective treatment for mild-moderate depression.^{13,14,109} Exercise regularly can also prevent or avoid depression relapse.^{14,102,107}

2.1.3 Risks and Costs of Physical Inactivity

Physical inactivity is detrimental to health, and research demonstrates the importance of avoiding inactivity. In 2012, Lancet Physical Activity Series Working Group indicated that physical inactivity is the fourth leading cause of death worldwide.⁹ Decreased physical activity is a robust independent predictor of mortality among men¹¹¹ and women.¹¹² Low levels of physical activity are inversely correlated to rates of coronary heart disease and death compared to men with moderate to high levels of physical activity.¹¹³ Not getting sufficient physical activity is a major cause of chronic diseases, as well as chronic diseases risk factors.¹¹⁴ Lee and colleagues

estimated that 6% of the burden of disease from coronary heart disease, 7% (95% CI, 3.9-9.6) of type 2 diabetes, 10% (95% CI, 5.6-14.1) of breast cancer, and 10% (95% CI, 5.7-13.8) of colon cancer can be attributed to physical inactivity.^{115,116} In 2008, physical inactivity caused 9% (95% CI, 5.1-12.5) of premature mortality, and more than 5.3 million of the 57 million deaths that occurred globally.^{115,116}

Globally, over 533,000 deaths could be prevented every year if inactivity was reduced by 10%.¹¹⁶ If inactivity was reduced by 25%, more than 1.3 million deaths could be prevented.¹¹⁶ Estimates indicate that elimination of physical inactivity would increase the life expectancy of the world's population by 0.68 (95% CI, 0.41-0.95) years.^{115,116} Every year, \$117 billion in health care costs are related to low physical activity.¹¹⁷

2.1.4 Physical Inactivity Among the U.S. Population

Despite the many benefits of an active lifestyle, physical activity levels remain low in the U.S. Figure 2.2 and Figure 2.3 indicate the trends of U.S. adults and adults in Alabama meeting aerobic and muscle strengthening guidelines from the 2011-2019 Behavioral Risk Factor Surveillance System (BRFSS), respectively. In 2019, BRFSS indicated that only 23.0% (95% CI, 22.7-23.3) of U.S. adults achieved the recommend physical activity, which includes at least 150 minutes a week of MPA or 75 minutes a week of VPA and engage in muscle-strengthening activities on 2 or more days a week.¹⁵ In Alabama, 17.0% of adults achieved the recommended physical activity levels (95% CI, 16.4-19.0).¹⁵


Figure 2.2. Percent of U.S. Adults Meeting Aerobic and Muscle Strengthening Guidelines 2011-2019¹⁵

Figure 2.3. Percent of Alabama Adults Meeting Aerobic and Muscle Strengthening Guidelines 2011-2019¹⁵



Figure 2.4 and Figure 2.5 indicate the trends of U.S. adults and adults in Alabama who engage in no leisure-time physical activity from the 2011-2019 BRFSS, respectively. It was reported that in 2019, 26.0% (95% CI, 25.7-26.3) of U.S. adults did not engage in leisure-time physical activity, compared to 31.5% (95% CI, 30.0-33.0) in Alabama.¹⁵

Figure 2.4. Percent of U.S. Adults Who Engage in No Leisure-Time Physical Activity 2011-2020¹⁵





Figure 2.5. Percent of Alabama Adults Who Engage in No Leisure-Time Physical Activity 2011-2020¹⁵

Data (Figure 2.2 - 2.5) were derived from https://www.cdc.gov/nccdphp/dnpao/data-trendsmaps. The data analysis and conclusions in this study are those of the author and do not necessarily represent the official position of the Centers for Disease Control

Data from the 2017-2020 BRFSS (Figure 2.6) indicate that the overall prevalence of selfreported physical inactivity is 25.3% among U.S. adults. The lowest prevalence of physical inactivity was in Colorado (17.7%), and the highest prevalence was in Puerto Rico (49.4%).¹⁵ Regionally, states in the South (27.5%) had the highest prevalence of physical inactivity, followed by the Midwest (25.2%), Northeast (24.7%), and West (21.0%). Seven states (West Virginia, Oklahoma, Louisiana, Alabama, Kentucky, Arkansas, a Mississippi) and Puerto Rico had a physical inactivity prevalence of 30% or more.¹⁵ Between 2011 and 2019, the prevalence significantly declined in 12 jurisdictions (CA, CO, HI, KY, ME, MA, MO, OH, OK, OR, VA, PR), and significantly increased in 11 jurisdictions (AL, KS, LA, MD, MN, MT, NC, TN, VT, WA, WV), and did not significantly change in 29 jurisdictions.¹¹⁸ The absolute difference in prevalence ranged from an 8.6% increase in Tennessee to an 11.0% decrease in Kentucky.¹¹⁸ Figure 2.6. Prevalence of Self-Reported Physical Inactivity Among US Adults by State and Territory, BRFSS, 2017-2020¹⁵



Data were derived from https://www.cdc.gov/nccdphp/dnpao/data-trends-maps. The data analysis and conclusions in this study are those of the author and do not necessarily represent the official position of the Centers for Disease Control

2.1.5 Physical Inactivity Among University Students

According to 15 U.S.C. § 1637®(1) and 20 U.S. Code § 1001(a), the term "college student" refers to "an individual who is a full-time or a part-time student attending an institution

of higher education" which "provides an educational program for which the institution awards a bachelor's degree" or equivalent.^{119,120} Typically, college students are always referred as undergraduate students across many studies,^{18,21,22,45,101,121-132} while university students are referred as undergraduate students, graduate students, and professional students.^{20,24,43,44,133-136} However, sometimes these two terms are used interchangeably.^{19,101,137-140} Therefore, we use the term "university students" to represent our target population of individuals who are full-time or part-time students (including undergraduate, graduate, and professional students) attending a college, university, or similar institution offering an associate or higher degree.¹⁴¹

Physical inactivity among US university students is a significant public health concern. On average 30%-35%¹⁸⁻²¹ of university students have overweight or obesity, and 40% to 50%^{20,22-24} of university students are physically inactive. According to a study by Buckworth and Nigg, college students typically engage in sedentary behaviors 30 hours a week.¹²² Only 30.6% of college students engaged in moderate-intensity physical activity on more than five days of the past week.¹²² The Fall 2021 American College Health Association-National College Health Assessment found that only 39.7% of university students (40.2% for undergraduate student and 38.3% for graduate and professional students) met the recommendation for strength training and aerobic activity.¹⁴² It is worth noting that compared with undergraduate students, the physical activity levels and behaviors of graduate and professional students have not been thoroughly studied.¹²⁸ This discrepancy could potentially be attributed to factors such as the larger population size of undergraduate students and the relatively shorter duration of graduate programs, limiting the available research opportunities and focus on this specific student group.

Physical activity during university years impacts habitual physical activity in adulthood, and consequently, has significant implications for short-term and long-term health

outcomes.^{143,144} After transition from high school to university, one third of university students were inactive.^{25,144} There is a growing trend among university students adopting a sedentary lifestyle with limited physical activity.^{25,145} Compared with university students aged ≥ 20 years, university students aged ≤ 19 years were more likely to engage in aerobic exercise (p < 0.01) and strength training (p < 0.02).¹⁴⁶ After graduating from college, about half of college students did not engage in physical activities efficiently and reported reduced physical activity levels, and there may be a greater decline in physical activity among students with less ingrained physical activity habits.¹⁴⁷⁻¹⁴⁹ Research on 233 undergraduate students found decreased physical activity levels during the transition from high school to college.¹²¹ During high school, a majority of the students (65%) reported engaging in regular vigorous physical activity, with an additional 26% participating in moderate activity. However, during their college years, the percentage of students engaging in regular vigorous activity dropped to 38%, while only 20% reported participating in moderate physical activity.¹²¹ Driskell and colleagues found that 47.8% of freshmen and sophomores reported aerobic activity participation at least three times weekly, compared to 42% of juniors and seniors.¹⁵⁰

In summary, university students typically do not meet physical activity recommendations. Among university students, the likelihood of meeting recommendations declines with age, and females tend to be less likely to meet recommendations than males.^{26,125,146}

2.2 A Potential Solution: Physical Activity Interventions

2.2.1 Competitions to Increase Physical Activity

Competition is defined as a rivalry where two or more parties strive for the same goal that cannot be shared, generally resulting in a victor.^{151,152} The competition can be over the attainment of any exclusive goal, including recognition, awards, goods, status, and prestige.

Competition-based behavioral interventions are fitness or exercise programs that integrate rankings, leader boards, status markers, incentives, and other competitive strategies to increase physical activity.^{153,154} Competitive environments motivate individuals to adjust their aspirations upward due to differences in goal attainment.¹⁵³ Competition has been shown to promote the motivation of physical activity among young adults who are competitive.^{33,34,155,156} The dynamic process of comparing oneself to others enhances one's expectations for success, results in extra effort, and ultimately strengthens overall physical activity.^{157,158}

Competition has been recognized to motivate individuals to better perform in various health behaviors, including physical activity.^{32,33} A randomized controlled trial in the UK recruited 281 physically inactive adults and found a significant increase in steps for participants in the competition condition compared to the control group, due to increased goal importance, identified motivation, and intrinsic motivation.¹⁵⁹ In 2019, an 11-week online exercise program compared the causal effects of social support and social comparison.¹⁵³ The study demonstrated that social comparison was more effective for increasing physical activity and did not depend on other individual or team incentives.

Using fitness trackers, a study conducted in 2019 found that participants in the competitive gamification (using "game design elements in nongame contexts") group (adjusted difference, 920; 95% CI, 513-1328; P < 0.001), support group (adjusted difference, 689; 95% CI, 267-977; P < 0.001), and collaboration group (adjusted difference, 637; 95% CI, 285-1017; P < 0.001) had a significant increase in daily steps from baseline during the 24-week intervention.¹⁶⁰ During the 12-week follow-up, physical activity levels continued to be significantly increased in the competition group compared to the control group (adjusted difference, 569; 95% CI, 142-996; P = 0.009), while there was no significant improvement in the

support (adjusted difference, 428; 95% CI, 19-837; P = 0.04) and collaboration (adjusted difference, 126; 95% CI, -248 to 468; P = 0.49) groups. In 2020, a feasibility study in nursing home settings found that motivational climate and physical performance may be improved by physical activity contests.³⁴

Competition may be beneficial to some individuals, but it can be detrimental to those who do not achieve their goals, feel pressured by competitors, or lose the competition.^{159,161,162} In a competitive physical task, goals can also influence physical performance, such as mastering the task or winning.^{162,163} In addition, participants' competitive orientation, which is defined as the desire to succeed in competitions, may impact how much effort they put into their rivals.¹⁶⁴ It is more likely that those with a more competitive behavioral orientation will put more effort into their competitions, whereas non-competitive individuals may find exercise less enjoyable even if they perform well.¹⁶⁵ Competing in rigorous physical activities could be difficult for uncompetitive individuals, particularly if these activities seem to be challenging.¹⁶⁶ Furthermore, compared to a virtual reality opponent, such as someone in an exercise game, individuals are more likely to put more effort into competing versus a "live" competitor.^{167,168}

2.2.2 Mobile Text-Message Interventions

Text messaging is one of the most common methods of interpersonal communication, involving the creation and real-time exchange of alphanumeric messages.³⁶ With text messaging, mobile phone users can be reached via push technology,¹⁶⁹ which is supported by virtually all mobile phones. Increasingly, text messages are being used for interventions designed to improve healthy behavior.^{37,170}

Text messages have been proven to be an effective method of delivering information to people.¹⁷¹ First, the use of text messages is widespread and statistics show that as of 2021, 97%

of U.S. adults own mobile phones¹⁷² Text messaging is the one of the most popular non-voice applications, and 75% of cell phone users send and receive text messages regularly.¹⁷¹ Previous research indicated that 92% of college students texted during their class time.¹⁷³ On average, college students send 71.0 (SD = 86.4, range from 1 to 300) texts per day¹⁷⁴, and spend 14.35 hours each week texting.¹⁷⁵ Second, text messages can be sent automatically and asynchronously. Recipients have the option to choose whether and when to read or reply to their messages.¹⁷⁶ Additionally, text messaging is also relatively cost effective. Text messages do not incur a great deal of cost to wireless carriers and researchers because the technology involved is very simple.¹⁷⁷

Text-based physical activity interventions are growing rapidly.^{37,178} Evidence from systematic reviews^{36,170,179-181} and meta-analyses^{171,182,183} report the effectiveness of text message interventions to address physical activity, highlighting the important roles of interventions with tailored messages and varying frequency. Physical activity messaging is intended to educate or persuade individuals or groups, with the ultimate goal of increasing their physical activity levels. A scoping review of physical activity messaging in 2020 identified concepts which should be included in physical activity messaging: 1) emphasize short-term social and mental health outcomes; 2) tailored and targeted content to intended recipient(s); and 3) integrate formative research, psychological theory, and social marketing principles.¹⁸⁰ In 2021, Williamson and colleagues developed and revised a provisional Physical Activity Messaging Framework (PAMF) to assist physical activity message generation and assessment.¹⁸⁴ It includes three domains: 1) Who, when, what, and how to send texts and why; 2) the message contents; and 3) message format and delivery methods.

2.2.3 Website-Delivered Interventions

The internet is an innovative medium with interactive technologies, creating reach, availability, and opportunities for health behavior change.^{185,186} Since 2000, there has been a more than 300% increase in internet usage globally.¹¹⁸ Today, 50% of the global population is internet-connected, and more than 90% of Americans have internet access.¹⁷² The popularity of the internet provides alternate means of health care services for individuals with physical disability or living in rural areas. Additionally, it produces chances to deliver health behavior interventions via the internet.¹⁸⁶ In recent years, online or website-based health behavior interventions for physical activity have become increasingly common, and have been reported to be effective.¹⁸⁶⁻¹⁹⁰ Interventions varied in content, duration, and frequency among studies, but included education, self-monitoring, feedback, tailored information, self-management, personal exercise plans, and online communication between health care providers and patients.¹⁹⁰ However, it remains unclear whether website-based interventions are effective in promoting long-term physical activity levels.¹⁸⁵ Further evaluations are needed to optimize website-based interventions characteristics.^{188,190}

2.2.4 Community Physical Activity Campaigns

Community-wide campaigns encompass large-scale, intensive, highly visible campaigns delivered to a large audience through various methods.²⁹ These campaigns may focus on overall health care, health behavior, or only focus on physical activity. An effective community campaign consists of building partnerships, delivering messages to a broad audience using multiple forms of media, and implementing programs as well as environmental and policy changes in concert.^{29,31,41} Examples of community events in the U.S. include self-help fitness groups, physical activity counseling, marketing materials, and environmental strategies (Table

2.3). Evidence from systematic reviews also supports the effectiveness of community-based

physical activity interventions.^{31,40-42}

Table 2.3.	Examples	of C	ommunity	Events	in	the	U.S.
1 4010 2.01	Linampies	01 U	ommonie	L , ento			0.0.

Community Physical Activity	Example	Description
Self-help fitness	Black Girls Run!	This organization offers resources for African American women
groups	$(BGR!)^{191}$	to join the health movement. BGR! aims to reduce the number
		of women suffering from chronic diseases caused by unhealthy
		diets and sedentary lifestyles.
Physical activity	Parks Rx Program ¹⁹²	Parks Rx promotes communications with patients and health
counseling		providers to discuss how to get active and use neighborhood
U		facilities to practice physical activities. From July 2016 to May
		2017, 7,360 adults and children received the "exercise as
		medicine" prescription.
Marketing	Birmingham REACH	This campaign is aimed to inform residents of Birmingham,
Materials	for Better Health:	Alabama, of the availability of fresh produce and physical
	Outdoor Physical	activity opportunities in the local area. The advertisement
	Activity ¹⁹³	features a female African American tying her shoe before
		running.
Environmental	The Atlanta Beltline	The Beltline is a 22-mile paved trail which is great for exercise
Strategies	Eastside Trail ¹⁹⁴	like running, biking, and walking. It is combined by trails, parks,
		rail transit, residential buildings, and commercial areas.

2.2.5 University Campus-based Physical Activity programs

University and college settings are ideal for implementing health promotion programs to engage large numbers of participants.^{24,43,44} The numbers of U.S. university students was 19 million in 2021, and there were 303,030 students in Alabama universities and colleges.¹⁹⁵ Campus-based programs are effective in refining students' exercise skills, competence, and interest in physical activity participation, which can promote an increase in physical activity levels (Table 2.4).^{21,26,28,43,44,129} Students at universities and colleges are the ideal target population for lifestyle interventions because they have access to multidisciplinary health professionals, world-class facilities, and specialized researchers in a learning environment.^{26,28}

University Campus-based Physical Activity programs	Example	Description	Outcomes
Competitions to Increase Physical Activity	Fitbit Competitions to Increase Physical Activity in College Students ¹⁹⁶	17 students from Texas A&M University were randomly assigned to either Fitbit or pedometer groups to assess if wearing a Fitbit watch and participating in weekly competitions via the app would increase weekly step counts as compared with a simple pedometer.	Fitbit participants were more motivated to engage in physical activities, while the pedometer group reduced their step count. The results confirmed that social and competitive aspects of a mobile app can be a powerful motivator for college students.
Mobile Text- Message Interventions	Daily Motivational Text Messages to Promote Physical Activity in University Students ¹³⁶	103 college students from the University of California, Berkeley were recruited to receive different types of motivational messages.	Sending text message resulted in an increase in daily steps (729 step increase, $p = 0.012$), but this effect decreased with time.
Website- delivered Interventions	Web-based Physical Activity Course ¹⁹⁷	233 college students were randomly allocated to a 10-week online course, a physical activity group, or a general health group.	Vigorous physical activity, self- regulation, and outcome expectancy significantly increased in the Web-based and physical activity course groups (p < 0.01).

Table 2.4. Examples of Campus-based Physical Activity Programs

2.3 Motivation and Barriers of Physical Activity among University Students

2.3.1 Motivation of Physical Activity among University Students

Why do students start participating in physical activity? What motivates them to continue, or even increase their participation? Understanding why university students engage in physical activity is important for intervention efforts to encourage more active health behaviors.¹⁹⁸ Physical activity fulfills a variety of needs, and people are involved in physical activity for considerably specific and different reasons.¹⁹⁹ Published literature demonstrates that compared to the general population, university students have similar reasons for engaging in physical activity, but motivations differ between male and female.^{125,198} Female students rated

weight management higher than male students, while male students rated competition, social recognition, and strength and endurance more critical.

Several intrinsic motivators drive physical activity among university students. Mullen and colleagues found that enjoyment from physical activities can increase exercise engagement and intentions,²⁰⁰ and some physical activities are enjoyable and do not require extra incentives.^{201,202} Kilpatrick's study demonstrated that enjoyment, appearance, weight management, and mental well-being were commonly reported motivations for engaging in PA and exercise among college students.¹²⁵ Alkhateeb and colleagues surveyed 417 university students in the Kingdom of Saudi Arabia and reported several reasons for practicing sports: appearance, weight management, strength gains, and avoiding illness.¹³⁵ Similarly, a qualitative interview of 20 German university students revealed five motivations: self-satisfaction, improving health, enjoyment, fitness, and study-life balance.²⁰³ Other intrinsic motivations include daily routine, past PA habits, and personal preferences.⁴⁴

Extrinsic motivators also served as influential factors in motivating university students to engage in physical activity. A study assessed the relationships between motivational climate and physical activity among university students in Spain and Romania.²⁰⁴ The results demonstrated that university students are more likely to be motivated to engage in physical activity if the fitness practice is task-involved (e.g., improve leadership skills in sport) rather than ego-involved (e.g., pursuit of competition). In addition to differences in personal interests and goals, social support has been found to be a critical factor to increase physical activity among university students.¹⁹⁹ Common physical activity motives included group involvement and friendship, along with reasons related to competition, gaining skills, adventure, and fitness associated with body image.^{125,135,198,203,205} Gaining friends and peers who share their experiences can allow college

students to engage in physical activities even outside the classroom or off campus.⁴⁴ Deliens and colleagues reported that pressure from parents, models, and peers also affect university students' PA.

2.3.2 Barriers to Physical Activity among University Students

College and university students face many barriers and challenges in balancing their academic studies, social life, and health. A substantial body of literature indicates that lower physical activity among university students has been attributed to individual (cost, health issues, self-regulation, time, and convenience), social (social support and monitoring), and environmental (neighborhoods, resources, weather, and communication) factors (Table 2.5). It has been demonstrated that a lack of concern for health leads to physical inactivity among university students. People who lead a healthy lifestyle are more likely to pay attention to exercise and persist with their habits.^{139,206,207} In addition, some physical education curricula or campus physical inactivity campaigns cannot accommodate students who have inadequate physical skills and knowledge.²⁰⁸

	Lack of motivation
Internal barriers ^{126,134,145,203,204}	Lack of skills and knowledge
	Lack of support
	Lack of access to appropriate facilities
Characteristics of the anvironment44,126,134,209	Unsafe neighborhoods
Characteristics of the environment states	Poor weather
	Lack of transportation
Time associated with physical	Travel time to physical activity facilities
activity ^{44,135,145,203}	Work and life balance
Nagativa avpariance while appaging in physical	Chronic illness and disability
Negative experience while engaging in physical	Pain, discomfort, and harassment/bullying
	by peers/others

Table 2.5. Barriers to Physical Activity Among University Students

2.4 Social Cognitive Theory Framework

It is acknowledged that the process of changing health behaviors is complicated and multifaceted, involving the interaction of various factors and feedback mechanisms. The process could result in delays in behavior change, adaptations to different interventions or strategies, competitive actions, and unexpected consequences that hinder or facilitate the desired changes. Feedback, both internal and external, plays a crucial role in shaping individuals' responses to their behaviors. ^{9,210} Therefore, embracing the complexity of the "entire system" is crucial for developing health behavior interventions, instead of focusing only on individual or environmental aspects.^{206,211,212} In other words, physical activity interventions should address not only how people practice physical activity in specific environments, but also how changes in environments influence people's engagement in physical activity.⁹ In this study, Social Cognitive Theory (SCT) will be applied to develop, deliver, and implement physical activity interventions.

Social Cognitive Theory (Figure 2.7) has been used to guide intervention design to increase the uptake of physical activity among university students. Proposed by Bandura in 1989, the SCT is a behavioral theory explaining people's behavior and the regulation thereof in a social context.²¹³ This theoretical framework explains "how people regulate their behavior through control and reinforcement to achieve goal-directed behavior that can be maintained over time."²¹⁴ The SCT began as Social Learning Theory (SLT) in 1978²¹⁵ and developed into the SCT in 1986.²¹³

The SCT posits that human behavior results from the dynamic interplay of personal, behavioral, and environmental influences. ^{213,216,217} The core idea is that humans can alter and construct environments according to their individual needs and, meanwhile, could be influenced by their surroundings. ^{213,216,218} Table 2.6 shows the primary constructs of the social cognitive

theory, including self-efficacy, self-regulation, incentive motivation, social support, outcome expectations, collective efficacy, observational learning, facilitation, and moral disengagement.

Figure 2.7. Social Cognitive Theory²¹⁹



Table 2.6. Social Cognitive Theory Constructs

Social Cognitive Theory Constructs	Definition	Description in Physical Activity Context
Self-efficacy	A person's confidence that he or she can perform a behavior	A person's confidence that they can do sufficient physical activity to meet the recommended guidelines
Self-regulation	Controlling oneself through self-monitoring, goal setting, feedback, self-reward, self- instruction, and enlistment of social support	Controlling oneself to perform sufficient physical activity through self-monitoring, goal setting, feedback, self-reward, self- instruction, and enlistment of social support
Incentive motivation	Using and misusing rewards and punishments to modify behavior	The use and misuse of rewards and punishments to modify physical activity behavior
Social support	Having friends, family, and other people, to turn to in times of need	Discussing physical activity, invitations to do physical activity, and celebrating the enjoyment of doing physical activity with peers, friends, family, and other people
Outcome expectations	Beliefs about the likelihood of the consequences from the behaviors that a person might choose to perform	Beliefs about the likelihood of beneficial consequences from physical activity
Collective efficacy	Beliefs about the ability of a group to perform certain actions	Beliefs about the ability of group physical activity
Observational learning	Learning to perform new behaviors by exposure to interpersonal or media displays of them, particularly through peer modeling	Learning to do physical activity by exposure to interpersonal or media displays of them, particularly through peer modeling
Facilitation	Providing tools, resources, or environmental changes that make new behaviors easier to perform	Providing tools, resources, or environmental changes that make PA behaviors easier to perform

	Ways of thinking about harmful behaviors	
Moral	and the people who are harmed that make	Considering the adverse effects of physical
disengagement	infliction of suffering acceptable by	inactivity
	disengaging self-regulatory moral standards	

According to the SCT, human behaviors are controlled by forethought through two main cognitive processes: self-efficacy and outcome expectations.^{214,217} Self-efficacy refers to the confidence or belief in one's ability to perform a specific behavior, while outcome expectation refers to beliefs about the likelihood and value of the results and consequences of a behavioral choice.^{128,213,220} As an interpersonal level theory, the SCT emphasizes reciprocal determinism which refers to the dynamic interaction between personal factors, behavior, and external environments.^{213,217} The environment influences both individuals and groups, but people and groups can regulate their own behavior as well as influence the environment.^{213,220}

It is essential that physical activity interventions are based on a strong theoretical framework.¹³⁰ Social Cognitive Theory has been widely used by researchers in designing and evaluating interventions targeted to improve physical activity, especially in Western and Asian cultures.^{123,221,222} A random-effects meta-analysis of 44 eligible studies of different populations revealed that these SCT models accounted for 31% of the variance in physical activity behavior.²²¹ The results of a prospective study indicated good fit (accounting for 55% of the variance) of a SCT-based model to the data of the physical activity in a sample of 277 university students in Virginia.²²³ It was found that self-efficacy was the strongest determinant of physical activity, followed by self-regulation and social support. Research involving 396 university students indicated that psychosocial factors were significant predictors of physical activity, including self-efficacy, outcome expectations, and social support.¹²⁴ Results from a study of 350 college students revealed that SCT constructs (role identity, self-regulation, outcome expectance,

social support, self-efficacy, and positive experience) accounted for 27% of the variance of VPA days.²²⁴ In a study involving 787 undergraduate students in Thailand, SCT constructs (self-efficacy, self-regulation and outcome expectations) were found to be significant predictors of physical activity, accounting for 29% of the variance.²²⁵ Choi and colleagues surveyed 688 Korean college students and found that SCT constructs were significant predictors of physical activity for both males (goal setting and self-efficacy) and females (goal setting and outcome expectations).²²⁶ Similarly, in a study of 937 undergraduate students, self-efficacy was found to be significant predictor for both genders²²⁷ In addition, the findings from 449 college students reported that self-efficacy and self-regulation were significant predictors of physical activity for both Black and White groups.¹³⁰

SCT and its constructs have also been used as a foundation for telehealth interventions. A qualitative study of 56 college students suggested the importance of self-efficacy for developing effective text message interventions to promote physical activity. In 2019, Wang and colleagues found that the use of mobile fitness apps had a significantly effect on physical activity levels through social support (β =0.126; P<0.001) and self-efficacy (β =0.294; P<0.001) among 384 Chinese college students.²²⁸ A focus groups study of 56 college students reported that SCT and its constructs provide a valuable framework for understanding how text messages can influence physical activity.¹²⁸

2.5 Summary of the Literature Review

A substantial body of literature highlights the association between physical activity and a wide range of physiological and psychological benefits. Being physically active can lower mortality risk, help manage weight, reduce the risk of disease and cancer, strengthen muscles, and improve mental well-being. ^{3-6,13,14,70,71} It is recommended that adults should perform at least

150 minutes of moderate-intensity or 75 minutes of vigorous-intensity physical activity per week, as well as muscle-strengthening activities on two or more days per week.⁶² However, physical activity levels remain low in the U.S. In 2019, only 23.0% of U.S. adults achieved the recommend physical activity, compared to 17.0% in Alabama.¹⁵ Furthermore, physical inactivity among university students is a significant public health concern: 30%-35% of university students have overweight or obesity, and 40%-50% of university students are physically inactive. ¹⁸⁻²⁴

Given the low prevalence rate, research on the effectiveness of different physical activity interventions is of significant importance for public health, particularly for university students.²⁵ Competition-based interventions have been recognized to motivate individuals to better perform in various health behaviors, including physical activity.^{32,33} Evidence supports the effectiveness of text message interventions in addressing physical activity.^{36,170,179-181} Campus-based programs also play critical roles in promoting physical activity among university students.^{21,26,28,43,44,129} Factors that motivate exercise among university students include perceived enjoyment, physical and mental health benefits (e.g., strength gains, improved athletic performance, improved body composition, improved cognition, improved affect/mood), and social factors (e.g., peers, family, friends, and environment), while internal barriers, ^{44,125,198,205} unsatisfactory environment, time management, and negative experience are common reasons why university students do not start in the first place or stop exercising.^{126,134,145,203,204}

Proposed by Bandura, SCT is one of the most widely used and robust behavioral theories for promoting physical activity interventions.^{128,138,213,222} SCT describes the dynamic interplay of personal, behavioral, and environmental influences.²¹⁹ The main constructs of SCT are selfefficacy, self-regulation, incentive motivation, social support, outcome expectations, collective

efficacy, observational learning, facilitation, and moral disengagement.^{213,220} Research has shown self-efficacy, self-regulation, and social support serve as strong influencers of PA.

Physical activity promotion studies that specifically target university students are limited. To provide evidence-based solutions for this group, well-designed interventions are highly recommended. Hence, we conducted a systematic review and meta-analysis to synthesize evidence of the effectiveness of text message interventions targeting physical activity from published literature. We conducted a three-group quasi-experiment study to assess and compare the effectiveness of text message interventions and a competition-based intervention in improving physical activity among university students, as well as evaluate participants' acceptability and experiences.

Chapter 3. Methods

The overall goal of this project was to assess and compare how text message interventions and a competition-based intervention impact university students' physical activity levels, mental well-being, and Social Cognitive Theory constructs. It was hypothesized that both the text message intervention and the competition-based intervention would positively influence university students' Physical Activity, Self-Efficacy, Self-Regulation, Social Support, Incentive Motivation, And Mental Well-being. The study design and methods used to examine these anticipated relationships are described in this chapter.

The first research question is, "What is the relevant evidence related to text message interventions targeting physical activity among university students?" This question was addressed by the systematic review and meta-analysis in Aim 1. Aim 1 inspired the development of text messages for Aim 2.1. The second research question is, "What are the effects of text message interventions and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental well-being?" This question was addressed by a quasi-experimental study in Aim 2.1. The third research question is, "What are participants' perceptions regarding their experience with and acceptability of text message interventions and competition-based interventions?" This question was addressed in Aim 2.2.

3.1 Specific Aim 1

To conduct a systematic review and meta-analysis of text message interventions targeting physical activity among university students from published literature.

Previous systematic reviews and meta-analyses have reported positive effects of text message interventions on improving physical activity levels in adults,^{179,229,230} children,^{231,232} adolescents,^{27,169,170,231,232} older adults,^{30,233-235} type 2 diabetes patients,^{235,236} and the general population^{38,237-239}. Therefore, we hypothesized that text message interventions are effective in improving physical activity among university students. However, there is no systematic review specifically focused on the effect of text messaging on physical activity among university students. A systematic review and meta-analysis to synthesize the findings qualitatively and quantitatively is foundational for the proposed project.

3.1.1 Eligibility Criteria

Inclusion criteria for studies in the systematic review were: 1) peer-reviewed articles; 2) written in English; 3) including university or college students; 4) using text messaging as an intervention to promote physical activity; 5) outcome variables are physical activity, fitness, or exercise; and 6) physical activity outcomes are measured at two or more time points (Table 3.1). Studies were excluded if they: 1) include people other than university students; 2) only include collegiate athletes; 3) do not report physical activity outcomes or are not aimed to improve physical activity; 4) do not include any intervention components about text messaging; or 5) are not published in English.

Table 3.1. Population, Intervention, Comparator, Outcome, and Study Design (PICOS) Criteria for the Systematic Reviews and Meta-Analysis

	Criteria	Justification
P opulation	University or college students	Aim 1 and 2 focus on university or college students
Intervention	Using text messaging as an intervention to increase physical activity or exercise	Sub Aim 1 is intended to determine the effectiveness of text messaging specifically
Comparator	Interventions with and without comparison or control group	Although lack of comparison or control group will increase bias, only including comparative studies could lead to fewer studies and more narrow review

Quitaoma	Physical activity levels, including physical activity scores, MPA minutes, VPA minutes, MVPA minutes, walking minutes, weekly steps, and metabolic equivalents (METs)	Sub Aim 1 is intended to determine the effectiveness of text messaging on physical activity levels
Outcome	Physical activity outcomes are measured at two or more time points	Sub Aim 1 is intended to detect the difference before and after text message interventions, so both baseline and follow-up measures are needed
Study design	Observational and interventional study	Sub Aim 1 is a systematic review and meta-analysis of text message interventions.

3.1.2 Data Source

A structured electronic search employing Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA)²⁴⁰ reporting guidelines was performed on text message intervention studies carried out in tertiary-level educational institutions. We searched the APA PsycINFO, National Library of Medicine's bibliographic database/article index (MEDLINE), SPORTDiscuss, Cumulative Index to Nursing and Allied Health Literature (CINAHL) through the EBSCO interface, and keywords through the Web of Science interface and a high sensitivity search for published, peer-reviewed studies via Google Scholar. We used the following keywords: (exercise or physical activity or fitness) AND (college students or university students or undergraduates or graduate students or professional students) AND (texting or text messaging or text message or SMS or short message service). Hand-searched published studies were also included using a snowballing method from the citations of retained articles. All retrieved studies were downloaded to Endnote X9 citation management software (Thomson Reuters, Philadelphia, PA, USA). Articles written in English and published in peer-reviewed journals were considered for review.

3.1.3 Study Screening and Data Extraction

Articles underwent initial title and abstract screening, followed by full-text screening. Each screening was first performed by the primary investigator and then checked by a second independent investigator. A standardized form based on article inclusion criteria was used to guide each screening (Table 3.2). Two investigators met before using this form to pre-test it with five purposefully selected articles from the initial search to discuss and fine-tune the inclusion and exclusion criteria and ensure consistency of application. If articles were included by both investigators, then they were included in the review. A consensus was reached over discrepancies in retained articles via discussion between the two reviewers, and mentors if needed.

To capture preferences from different types of university students, exclusions were not made based on university students' gender, age, majors, race, ethnicity, or other socioeconomic factors, and both undergraduate students and graduate students were included. During title and abstract screening, the following criteria were required to be met for inclusion: English, university students, text message interventions, and physical activity. All inclusion criteria were required to be met during the full-text screening for article inclusion in the review. Table 3.2. Standardized Article Screening Form for Systematic Review

Study	English	University students	Text message interventions	Physical activity	Outcomes	Extraction	Include	Exclude

Data extraction was performed by the primary investigator using a standardized form developed by Cochrane Reviews²⁴¹ and consisted of the following dimensions: target sample and size, study design, intervention description and duration, PA measurement, and key findings. For meta-analysis, the value of physical activity scores, MPA minutes, VPA minutes, MVPA minutes, walking minutes, and weekly steps (mean and standard deviation) were extracted from each intervention group.

3.1.4 Quality Assessment

The risk of bias within each study was assessed by the primary investigator using the Mixed Methods Appraisal Tool (MMAT), Version 2018, ^{46,47} and was performed simultaneously with data extraction. Results were reviewed by another investigator for accuracy and completeness. The MMAT was first published in 2009 and has five categories of study design: (1) qualitative, such as case studies and grounded theory; (2) randomized controlled trials; (3) nonrandomized trials, such as cohort studies and case-control studies; (4) quantitative descriptive studies, such as surveys and case series; and (5) mixed methods studies. Each category contains five methodological criteria with assessment questions. Possible responses to the questions include 'Yes,' 'No,' and 'Can't tell.' The 'Can't tell' response category means that the paper does not report appropriate information to answer, 'Yes' or 'No', or that it reports unclear information related to the criterion.

3.1.5 Meta-analysis

Due to the possible variations in physical activity measurement, the data were combined across studies using standardized mean difference (SMD)²⁴² if studies reported the same outcomes. Meta-analysis was performed using Review Manager Software 5.3 Copenhagen (The Nordic Cochrane Center, the Cochrane Collaboration, 2014) if at least three studies^{49,50} reported the same physical activity effectiveness outcomes (MPA minutes, VPA minutes, MVPA minutes, weekly walking minutes, weekly steps, physical activity scores, or METs). The physical activity levels, including mean difference (MD), standard deviation (SD), or 95% CI were extracted from each intervention group if 1) both pre- and post-intervention values or both intervention groups' values and comparative groups' values were available; 2) the sample size was known; and 3) the measurements of outcomes were sufficiently similar and comparable.

Heterogeneity among studies was assessed using Higgins I² statistic.^{26,39,243} If I² value was greater than 75%, a random-effects model was conducted, otherwise we chose a fixed-effects model.^{26,39,243} Thus, we conducted subgroup meta-analysis of each outcome if there were at least three studies having reported the same outcome (MPA minutes, VPA minutes, MVPA minutes, weekly walking minutes, weekly steps, physical activity scores, and MET). For all analyses, the significance level was p < 0.05.

3.1.6 Publication Bias

A major challenge for conducting systematic reviews was that not all studies are published, and those that are published may differ from unpublished studies in the statistical significance of their results.^{244,245} The probability of publishing research with statistically significant results is higher than that of work with null results or without significant results.^{244,246,247} Publication bias was assessed using a funnel plot asymmetry test (Figure 3.1).^{248,249} In funnel plots, the effect estimates (on the X-axis) from individual studies are plotted against the precision of each study on the Y-axis (usually measured by the standard error).²⁵⁰ Studies with a smaller sample size are more likely to appear wide at the bottom, while studies with a larger sample size are more likely to appear narrow at the top.⁴⁸ If there is no bias and between-study heterogeneity, the scatter of included studies will result only from sampling variation, and the plot will seem like a symmetrical inverted funnel, as shown in Figure 3.1.²⁵¹ Note that the funnel plot asymmetry test is only used for meta-analyses containing more than ten studies.²⁵²

In the fixed-effect meta-analysis model, it is assumed that all studies share a common effect size (Figure 3.1).^{251,253} Figure 3.1 shows that the triangle area is centered on the center line, which shows the estimated common effect of all studies in the meta-analysis. In the absence

of heterogeneity and biases (such as publication bias and data irregularities), the pseudo 95% CI (triangle area) shows the predicted distribution of studies. Conversely, in the random-effect meta-analysis model (Figure 3.2), it is assumed that the estimated effects in the different studies are not identical because of the real differences between studies and sampling variability.^{253,254} Therefore, it is not necessary to add a pseudo 95% CI.

Figure 3.1. Example of Symmetrical Funnel Plot (Fixed-Effect Meta-Analysis Model)⁴⁸



Figure 3.2. Example of Funnel Plot for the Random-Effects Meta-Analysis²⁵⁵



3.1.7 Potential Limitations and Alternative Strategies

First, to avoid insufficient search of electronic databases and failure to include all relevant studies, we worked with the Auburn University Health Sciences Librarian to create

appropriate research queries for each electronic database. Second, bias in selecting studies cannot be totally avoided in a systematic review or meta-analysis. Although assessments of the magnitude of the bias or how this source of bias can be reduced are not available,²⁵⁶⁻²⁵⁹ we evaluated risk of bias for all included studies. In order to identify and possibly reduce such bias, we used a dual review process with transparent inclusion criteria, and referred to gray literature (e.g., U.S. FDA documents, trial registry reports) if needed.²⁵⁶⁻²⁵⁸ Last, funnel plot asymmetry tests should not be used if there are less than 10 studies included in the meta-analysis.^{241,260} Thus, the alternate strategy is to conduct a narrative synthesis of evidence instead of a quantitative analysis.

3.2 Specific Aim 2

Specific Aim 2: To assess and compare the effectiveness of text message interventions and a competition-based intervention in improving physical activity among university students, and evaluate participants' acceptability and experiences of the text message interventions and the competition-based intervention

Specific Aim 2.1: To conduct a quasi-experimental study to assess and compare the effectiveness of text message interventions and a competition-based intervention among university students

Specific Aim 2.2: To evaluate participants' acceptability and experiences of the text message interventions and the competition-based intervention

To encourage health professional students to be physically active, Auburn University (AU) holds the Deans' Fit Family Challenge each fall semester since 2019 among the Harrison College of Pharmacy (AUHCOP), College of Nursing (AUCON), and the Edward Via College

of Osteopathic Medicine (VCOM). However, the implementation and impact of the DFFC have not been systematically evaluated. Previous systematic reviews and meta-analysis indicate positive effectiveness of text message interventions in promoting physical activity and other behavioral outcomes in adults, youth, and the general population.^{36,38,39,170,182,183,261} Evidence also shows that theory-based interventions on physical activity behavior are more effective than those that are not.²⁶²⁻²⁶⁵ Hence, we hypothesized that:

1) Participants in both the text message and DFFC competition groups would report significant increases in physical activity.

2) There is a significant difference in post-test SCT variables and mental-welling scores between intervention groups and control groups.

3) The effects of physical activity interventions are mediated by SCT variables and mental well-being.

3.2.1 Research Design / Experimental Approach

A quasi-experimental study was utilized to accomplish Aim 2. Participants in the 2022 Deans' Fit Family Challenge competition were invited and randomly assigned to the competition intervention group (Group 1) or the competition plus text message intervention group (Group 2). Students in HCOP, AUCON, and VCOM who did not enroll in the 2022 DFFC were invited as the Control Group (Group 3). Participants in Group 3 did not participate in 2022 DFFC or receive text messages from study personnel. Recruitment plans are explained below.

3.2.1.1 Sample Size Calculation

The effect size of text message interventions on improving physical activity varies in the literature. Fanning and colleagues reported that the overall weighted mean effect size of mobile device on increasing physical activity is 0.54.¹⁸³ Smith and colleagues also indicated that the

effect on increasing physical activity of one-arm text message studies is 0.54.²²⁹ Head and colleagues found that SMS interventions were most successful on smoking cessation (d = 0.447; 95% CI = 0.367, 0.526; p < 0.001) and physical activity (d = 0.509; 95% CI = 0.236, 0.781; p < 0.001).¹⁷¹ Armanasco and colleagues suggested that text message interventions targeting physical activity showed a small but significant pooled effect (d=0.35, 95% CI=0.17, 0.53, p<0.001).²⁶⁶ However, there are few meta-analysis that have reported effect sizes for fitness competitions.

To achieve a power of 80% and a level of significance of 5% (two sided) for detecting an effect size of 0.3 using Repeated measures ANOVA, this study required a sample size of 22 per group (66 in total).²⁶⁷ We targeted a larger sample size goal of 66 inflated to 100 to allow for attrition.

3.2.1.2 Incentives

Participants were incentivized with a total of \$15 Amazon gift cards to encourage enrollment and prevent attrition. Specifically, \$5 was provided at baseline after completing the week-0 survey, and a \$10 Amazon gift card was provided upon completion of the study. At the conclusion of DFFC, there was a drawing for participants from all three groups who completed both surveys, one for \$20, one for \$30, and one for \$50.

3.2.2 Participants and Settings

3.2.2.1 Inclusion and Exclusion Criteria

Our recruitment pool for the DFFC Group and DFFC + Text Group included all students who were enrolled in the 2022 DFFC in the HCOP, AUCON, and VCOM, while the recruitment pool for the Control Group included all HCOP, AUCON, and VCOM students who did not enroll in the 2022 DFFC. We recruited university students (\geq 18 years of age) who could read and write English, possessed a smartphone and a fitness tracker or smartwatch, and a expressed a willingness to receive a weekly email including motivational information, instructions, and reminders for physical activity. Participants were expected to automatically submit their weekly MVPA minutes using their fitness trackers or smartwatches (if using one of the following apps, which could synchronize activity minutes to the ChallengeRunner app: Apple Health, Fitbit, Garmin, Google Fit, Omron, Oura, Polar, Samsung Health, Strava, Suunto, Under Armour MapMyFitness, Withings). Participants could also choose to manually enter data on the ChallengeRunner website (https://www.challengerunner.com). Participants were also required to be able to perform physical activities. Pregnant women, and individuals with disabilities or chronic diseases that prevent them from being physically active were excluded.

ChallengeRunner is an online platform for establishing, tracking, and managing fitness and wellness challenges among employees.²⁶⁸ With ChallengeRunner, administrators are able to easily set up and run many types of fitness challenges. Also, the leaderboard in the ChallengeRunner app and website, presented as a bar chart displaying each team's updated minutes, enables participants to stay informed about their rankings. It is also simple for users to enter, monitor, and edit activity data from the ChallengeRunner website, mobile apps, fitness activity trackers, or via text message.²⁶⁸ Everyone who participates in the 2022 Challenge submitted their data via ChallengeRunner, independent of their participation in the study described here.

3.2.2.2 Recruitment

All research participants were recruited from HCOP, AUCON, or VCOM students. Information about the study was included in the first email sent about the DFFC (Appendix 1). The "2nd DFFC email" (Appendix 2) contained a link for non-DFFC students to sign up for the study (Appendix 3). The "3rd DFFC email" (Appendix 4) was only sent to those who signed up

for the DFFC, containing a link to sign up for the study (Appendix 5). Students desiring to participate in this project provided the informed consent approved by the Institutional Review Board (IRB) of Auburn University or VCOM by selecting "Yes" to the question, "Do you agree to participate in this project?" Participants answered three screening questions to indicate they were at least 18 years of age, not pregnant or with disabilities or chronic diseases, and would submit their minutes to ChallengeRunner automatically through their fitness tracker or smartphone apps. Participants' Auburn or VCOM email addresses were collected in the recruitment survey. This study also received IRB approval from VCOM prior to recruiting VCOM students.

3.2.3 Interventions

There were two intervention groups and one control group (Figure 3.3). Participants in the DFFC Group and DFFC + Text Group participated in the 8-week DFFC competition among HCOP, AUCON, and VCOM from September 19 to November 11. Participants in DFFC + Text Group received three text messages about physical activity per week, on Mondays, Wednesdays, and Friday. The literature suggests that interventions involving daily^{136,137,269} or 2-4 text messages per week^{131,140,270-272} are feasible and acceptable among university students. Participants in the Control Group were students in HCOP, AUCON, and VCOM who did not enroll in the 2022 DFFC. They did not participate in 2022 DFFC, nor receive any text messages. After the DFFC, physical activity data were collected for an additional four weeks follow-up for all three groups.





3.2.3.1 The Deans' Fit Family Challenge

The Deans' Fit Family Challenge began on September 19, 2022, and concluded on November 11, 2022, lasting eight weeks. Table 3.3 depicts the relationships between components of DFFC and SCT constructs, which are discussed in the following section. Considering the nature of both interventions, these domains were selected because of their ability to influence physical activity in this population. Self-Regulation: Self-regulation refers to controlling oneself to perform sufficient physical activity through self-monitoring, goal setting, feedback, selfreward, self-instruction, and enlistment of social support. The ChallengeRunner app allowed participants to automatically submit activity minutes, after initially connecting the app to their fitness trackers. Participants received a weekly email reminder to track their physical activity minutes. They could also edit their minutes manually on the ChallengeRunner app or website, if desired. Self-Efficacy: Self-efficacy refers to a person's confidence that they can perform sufficient physical activity to meet the recommended guidelines. DFFC also focused on developing self-efficacy through weekly motivational emails that include general information, instructions, and reminders to promote physical activity and mental well-being behavior. During the competition period, fact sheets, flyers, and infographics from the CDC website containing the

knowledge, information, guidance, recommendation, and initiative were shared via email to promote exercise.²⁷³ Evidence indicates that advertisements can improve self-efficacy and motivation of physical activity.^{274,275} <u>Social support</u>: Social support refers to discussing physical activity, invitations to perform physical activity, and celebrating the enjoyment of doing physical activity with peers, friends, family, and other people. Participants competed in teams of two throughout the competition. Additionally, group fitness activities provided opportunities to participate in physical activity classes with peers. The group fitness activities in 2022 included a yoga class, F45 class, True 40 class, HIIT class, and an Orange Theory Fitness class. <u>Incentive motivation</u>: Incentive motivation refers to the use and misuse of rewards and punishments to modify physical activity behavior. Weekly winners were announced based on activity minutes for the previous week. Additionally, the winning college was awarded the Deans' Fit Family Challenge Trophy at the conclusion of the competition.

Theoretical construct	DFFC components	Examples
Self-Efficacy	Weekly motivational email	• Receive weekly emails about information, instructions, and reminders to promote physical activity and mental health behaviors
Self- Regulation	Weekly activity minute logging	 Participants receive a weekly email reminder to track their minutes The ChallengeRunner app allows participants to enter their minutes manually, or automatically submit activity minutes after initially connecting the app to their fitness tracker
Regulation	Weight reporting	• Participants provide self-reported weight prior to and at the conclusion of the DFFC
	Working with teammate	Participants compete in teams of two throughout the DFFC
Social Support	Group fitness classes	• Orange Theory Fitness, Yoga, F45, True 40 HIIT at student recreation center
Incentive Motivation	Prizes and awards throughout the Challenge	 The winning college is awarded the DFFC Trophy by the university's Provost Weekly winners are announced based on physical activity minutes for the previous week Individual winners are determined based on minutes of physical activity and weight loss

Table 3.3. The SCT Components of the Deans' Fit Family Challenge

3.2.3.2 Text message interventions

Text message frequency was informed by the literature which suggests that college students prefer not to see repeat text messages, or receive text messages too early in the morning,

late at night or during weekends, while they prefer to receive messages between 10am and 7pm.^{132,270,276} Participants recruited from 2022 DFFC were randomly assigned to the DFFC Group or the DFFC + text group. Participants in DFFC + text group received three text messages per week, on Mondays, Wednesdays, and Friday.

Text messages were sent through a third-party text platform, EZ Texting.^{277,278} Table 3.4 contains the detailed texts schedule and components for the DFFC + Text Group. Appendix 6 shows 20 texts that were sent to the DFFC + Text Group. Reviews of physical activity messaging indicate that text messages to adults should be framed positively instead of negatively.^{180,181} Accordingly, we designed gain-framed messages rather than loss-framed messages to address participants' Self-Efficacy, Social Support, Self-Regulation, and Incentive motivation. Self-Efficacy: We developed messages focusing on self-efficacy to make participants feel encouraged and increase awareness of their own activity.^{180,184,279} Social Support: In text messages focusing on social support, participants received guidance or suggestions to practice physical activity with friends.^{280,281} Incentive motivation: Research indicates that gentle and suggestive motivational messages are acceptable by providing practical tips and suggestions to promote exercise.¹⁷⁷ Self-Regulation: Messages targeting self-regulation were sent every Friday to remind participants that their minutes were collected and to check if they achieved the recommended 150 minutes for the week.^{177,282} We drafted the textual content, drawing inspiration from published literature. Two committee members reviewed and refined the drafted texts. We also asked 3-5 Doctor of Pharmacy students and graduate students to test all text messages, and we revised the messages based on students' feedback.

	Theoretical construct	Texting components	Examples
Monday	Self-efficacy	To encourage participants and increase awareness of their own activity	• The first step is the hardest. Focus on getting out there and getting started. You've got this!
Wednesday	Social support	Guidance or suggestions to practice physical activity with friends/peers/teammates	 If you're having trouble finding a way to stop sitting, grab a friend and exercise!
Fuidan	Self- regulation	To remind participants that their PA minutes will be collected and to check if they achieved the recommended 150 minutes for the week.	Happy Friday! Your exercise minutes will be collected today. Set a regular time to exercise daily, so you'll be healthy!
гнау	Incentive motivation	Motivational messages providing practical tips and suggestions to promote exercise	• Even if you're glued to your phone, you don't have to be glued to your seat! Make it a habit this week to talk and walk whenever possible!

Table 3.4. Text Message Schedule and Examples Based on SCT Components

3.2.4 Measures and Data Collection

The primary outcome of this project is physical activity levels, measured by weekly moderate-to-vigorous activity (MVPA) minutes. Secondary outcomes were collected via selfreport survey including Self-Efficacy, Self-regulation, Social Support, Incentive motivation, Mental Well-being, and participants' experiences and acceptability with the DFFC and text messages. Self-Efficacy, Self-regulation, Social Support, Incentive motivation, and Mental Wellbeing were assessed by validated scales during both the week-0 and week-8 surveys. Participants' experiences and acceptability were assessed by multiple choice, 5-point Likert scale, and open-ended questions during the week-8 survey. Week-0 survey and Week-8 survey for the competition group can be found in Appendices 7-10. After completing week-0 and week-8 surveys, participants were redirected to a separate survey (Appendix 11-12) to enter their email address for a \$5 or \$10 Amazon gift card. The information collected in the contact information survey was maintained in a separate database and not linked in any way to participants' responses to their main surveys. The outcomes, measurement methods, and data collection methods are presented in Table 3.5 below.
Outcomes		Measurement	Data collection methods	Time Point
Demograp	ohic Variables	Race, ethnicity, gender, and age	Self-reported survey	Week-0
		Weekly moderate and vigorous	Self-reported Survey (One multiple choice question)	Week-0
Outcome	levels	activity (MVPA) minutes	Collected via the ChallengeRunner app	Weeks 1- 12
	Self-Efficacy	9-item Exercise Self Efficacy Scale ²⁸³		
	Self-Regulation	10-item Self-Regulation Scale ²²³		
	Social Support	5-item Friend Support for Exercise Habits Scale ²²³		Week-0 and week-
Secondary Outcomes	Incentive motivation	23-item Physical Activity Motivation Scale ^{284,285}	Self-reported survey	8
	Mental Well- being	The World Health Organization Five Well-being Index (WHO- 5) ^{286,287}		
	Participants' experiences	Two multiple choice and two open-ended questions		West 9
	Acceptability	Two 5-point Likert scale questions		week-8

Table 3.5. Outcomes, Measurement, and Data Collection

3.2.4.1 Demographic Information

During the week-0 survey, participants' demographic information was collected via Qualtrics, including gender, age, ethnicity, and race. Students were asked if they participated in other organized fitness programs during the study period. Additionally, participants' health literacy was assessed by Brief Health Literacy Screening Tool – BRIEF.²⁸⁸ The BRIEF scores range from 2 to 20, where scores of 2 to 12 indicate *limited* health literacy ("Not able to read most low literacy health materials; will need repeated oral instructions; materials should be composed of illustrations or video tapes. Will need low literacy materials; may not be able to read a prescription label."), scores of 13 to 16 indicate *marginal* health literacy ("May need assistance; may struggle with patient education materials."), and scores of 17 to 20 indicate *adequate* health

literacy ("Will be able to read and comprehend most patient education materials"). Research indicates that BRIEF had a strong internal consistency in adults (Cronbach's $\alpha = 0.77$).²⁸⁸

3.2.4.2 Physical Activity

The primary outcome of this aim was physical activity, measured by weekly MVPA minutes. Weekly MVPA minutes were collected via fitness trackers or smartwatches and synchronized to the ChallengeRunner app or manual data entry (Figure 3.4 & Table 3.6). During the week-0 survey, participants were encouraged to look at data collected the previous week by their existing activity tracker and report their weekly MVPA minutes. Additionally, participants were asked if they participate in any other organizational program to increase their physical activity during the study period. Weekly MVPA minutes were analyzed to determine the effectiveness of interventions. Participants were asked to track and report their weekly MVPA minutes for 12 weeks, from September 19, 2022, to December 11, 2022.

Figure 3.4. Process of Collecting MVPA Minutes Via the ChallengeRunner App



	Steps	Distance	Active	Calories	Sleep	Walking, Running,	Walking, Running,
	Walked	Traveled	Minutes	Burned	Hours	Swimming, or Cycling	Swimming, or Cycling
						Distance	Minutes
Apple Health	Х	Х	Х	Х	Х	Х	Х
Fitbit	X	Х	Х	Х	Х	Х	Х
Garmin	X	Х	Х	Х	Х	Х	Х
Google Fit Online	Х	Х	Х				Х
Google Fit Mobile	Х	X	Х	Х	Х	Х	Х

Table 3.6. ChallengeRunner Trackers and Activities

Misfit	Х	Х		Х	Х		
Omron	Х	Х	Х	Х			
Oura	Х	Х	Х	Х	Х		
Polar	Х	Х	Х	Х	Х	Х	X
Samsung Health	Х	Х	Х	Х	Х	Х	X
Strava	Х	Х	Х			Х	X
Suunto	Х	Х	Х	Х		Х	X
Under Armour MapMyFitness	Х	X	Х	Х	Х	Х	Х
Withings	Х	Х	Х	Х	Х		

*All apps support tracking steps, 13/14 apps support collecting active minutes, 12/14 apps support collecting burned calories.

In the week-0 and week-8 survey, the Physical Activity Vital Sign (PAVS)²⁸⁹ was used to screen PA and muscle strengthening exercises for participants. PAVS is a widely used PA assessment tool incorporating three self-reported questions: 1) On average, how many days per week do you engage in moderate to vigorous physical activity (like a brisk walk)?, 2) On average, how many minutes do you engage in physical activity at this level?, and 3) How many days a week do you perform muscle strengthening exercises, such as bodyweight exercises or resistance training?

3.2.4.3 Mental Well-being

The World Health Organization Five Well-being Index (WHO-5) was used to assess subjective psychological well-being.²⁸⁷ The WHO-5 score is the sum of responses to five items: (1) 'I have felt cheerful and in good spirits'; (2) 'I have felt calm and relaxed'; (3) 'I have felt active and vigorous'; (4) 'I woke up feeling fresh and rested'; and (5) 'My daily life has been filled with things that interest me'. These items are scored on a five-point Likert type scale from 0 (at no time) to 5 (all the time), summed, and the summed score is multiplied by a factor of four. Clinical validity of the WHO-5 has been evaluated as high since it can be used across many different settings, regardless of the underlying illness.^{286,287} Usually, a score of \leq 50 is recommended for clinical depression screening.^{286,287} As shown in previous research, WHO-5 has demonstrated high internal consistency in university students (Cronbach's α = 0.85, 0.81, and 0.89).^{290,291} A referral list (Appendix 13) of mental health providers, including on campus services, off campus services, and emergency services, was provided for participants at the end of survey.

3.2.4.4 Self-Efficacy

A 9-item Self-Efficacy for Exercise Scale was used to measure self-efficacy by assessing participants' belief that they are able to meet the physical activity guidance for adults when encountering different barriers. ²⁸³ In each situation, participants provide their current confidence level ratings on a scale from 1 (not confident) to 5 (very confident). The overall score is calculated as the average of the 9-item ratings, with higher scores indicating higher self-efficacy. This scale has been used widely in studies applying SCT to understanding physical activity^{283,292-294} and has demonstrated high internal consistency among older adults (Cronbach's $\alpha = 0.92^{283}$ and 0.75^{293}) nursing and medical students (Cronbach's $\alpha = 0.88$),²⁹² and adults (Cronbach's $\alpha = 0.90-0.94$).²⁹⁴

3.2.4.5 Incentive motivation

Incentive motivation was measured using the 23-item Motivation for Physical Activities Measure.²⁸⁴ Participants rate 23 statements on a 5-point scale ranging from 1 (strongly disagree) to 5 (strongly agree) with higher scores indicating higher Incentive Motivation. This scale has been used in studies applying SCT to understanding physical activity²⁸⁵ and has demonstrated high internal consistency among adults (Cronbach's $\alpha = 0.92^{285}$ and Cronbach's $\alpha = 0.82^{295}$).

3.2.4.6 Social Support

Social support from friends, family, and teammates was measured by five items from the Social Support for Exercise Habits Scale.²⁹⁶ This scale consists of 15 items from three subscales: Family Participation, Family Punishment, and Friends Participation. The score of each item is from 1 (never) to 5 (very often) with higher scores indicating higher Social Support. The Friends scale has been used widely in studies applying SCT to understanding physical activity,^{124,138,223,297-300} and has demonstrated high internal consistency among Hispanic populations (RMSEA=0.106, CFI=0.919, TLI=0.839, and SRMR=0.045),³⁰¹ urban African American adults (Cronbach's $\alpha = 0.967$),²⁹⁷ young university students (Cronbach's $\alpha = 0.91$),²²³ and African American female college students (Cronbach's $\alpha = 0.95$).¹³⁸

3.2.4.7 Self-Regulation

Self-regulation was measured using the 10-item Exercise Planning and Scheduling Scale (EPS) instruments.²²³ Participants rate EPS items on a 5-point scale ranging from 1 (does not describe) to 5 (describes completely) with higher scores indicating higher Self-regulation. This EPS scale has been used widely in studies applying SCT to understanding physical activity^{223,298-300,302,303} and has demonstrated high internal consistency among Chinese adolescents (Cronbach's $\alpha = 0.91$),²⁹⁹ young university students (Cronbach's $\alpha = 0.87$),²²³ and female undergraduate students ($\alpha = 0.72 - 0.89$).³⁰⁴

3.2.4.8 Participants' Experiences

Participants were asked to identify an enabler and barrier to physical activity during the study through two multiple choice questions (one for enablers, and one for barriers) on the week-8 survey. The survey also included two open-ended questions asking participants to describe an

aspect of the intervention (i.e., the DFFC or text messages) they enjoyed the most and what they dislike about the intervention.

3.2.4.9 Acceptability

Acceptability was assessed via two 5-point Likert scale questions: (1) 'Overall, how satisfied were you with the DFFC/text message intervention', the score of each item is from 1 (very dissatisfied) to 5 (very satisfied); and (2) 'How likely will you choose to participate in next year's Challenge if you are still eligible/How likely are you to choose to receive text messages about physical activity in the future, the score of each item was from 1 (extremely unlikely) to 5 (extremely likely).

3.2.5 Statistical Analysis

3.2.5.1 Quantitative Analysis

Descriptive analyses were conducted to analyze demographic variables and SCT variables prior to the study. Two sample t-test (for continuous variables) and chi-square test or fisher exact test (for categorical variables) were used to compare the baseline characteristics between the Control Group and other participants who enrolled in the DFFC (including DFFC Group and DFFC + Text Group).

Path analysis was performed to assess hypothesized relationships between mental wellbeing, self-efficacy, self-regulation, incentive motivation, social support, and physical activity using the pre-test survey data (Figure 3.5). Arrows present a direct effect of each variable on an endogenous variable.

61

Figure 3.5 Path Analysis of the Effect of Interventions on Weekly MVPA Minutes, Mental Wellbeing, Self-Efficacy, Self-Regulation, Incentive Motivation, and Social Support



Before interpreting the causal relationship of the structural model, measurements of model fit were assessed.³⁰⁵ Fit refers to the capability of a model to generalize data; a good model fit refers to the capability of a model to accurately predict unknown data.^{305,306} There could be an error in the proposed structure model and/or data if the model does not fit the data. As a result, the predicted parameter values could be questionable.³⁰⁷ The Chi square and Root Mean Square Error of Approximation (RMSEA) were calculated to measure the accuracy of the model. Non-significant Chi-square value indicates a good model fit.^{276,308} A RMSEA value between 0.08 and 0.1 is considered borderline fit, a value between 0.05 and 0.08 is considered acceptable fit, and a value less than 0.05 indicates an excellent model fit.^{276,309}

Since we had repeated measurements of weekly MVPA minutes for all three groups for 8 weeks, the Repeated Measures ANOVA was used to test the effect of the DFFC and text

messaging during the DFFC, as well as the four weeks after the DFFC concluded. One-way ANOVA was then used if weekly MVPA minutes were statistically significant different between the groups, and the paired t-test was used for within group comparations. A p-value of less than 0.05 was considered statistically significant. Data were analyzed using SPSS version 22.0 (IBM Armonk, New York, USA).

One-way ANOVA was performed to determine if Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation were significant different between the three groups after the interventions. Five mediation models were developed to determine if the effect of the interventions on Physical Activity was mediated by SCT variables. For each model, the independent variable is the interventions, the dependent variable is physical activity, and the mediator variables are Mental Well-being, Incentive Motivation, Self-Efficacy, Social Support, and Self-Regulation, respectively.

3.2.5.2 Qualitative Analyses

Responses to the open-ended questions were analyzed using open coding and thematic analysis with NVivo 12 (QSR International, Melbourne, Australia) software. The coding team comprised two researchers. Prior to analyzing data, an initial guide of deductive coding based on the constructs of enablers, barriers, experience, and acceptability was developed by the primary investigator.^{310,311} Two coders were both responsible for coding the entire dataset. Similar codes or patterns were grouped into larger themes, and emerged themes were assessed for internal homogeneity (e.g., internal cohesiveness) and exclusivity between categories.³¹²⁻³¹⁴ Representative quotes were selected to further illustrate each theme. To maintain methodological rigor and trustworthiness of data analysis, kept a detailed audit trail and reflection journal of our initial definitions of categories, sub-categories and schema, and our discussions during these

63

meetings.³¹⁵⁻³¹⁷ Disagreements were solved through discourse and consensus. To support the trustworthiness of interpretations and analyses, we kept an audit trail to track who, when, and how to collect and analyze data.³¹⁸

Table 3.7 below provides a concise summary of the data analysis plan for the studies.

Table 3.7. Summary of Data Analyses Plan

Aim	Outcomes	Variables	Instrument	Time Point	Analysis	
1	Synthesis and effectiveness of text message interventions in university students	-		-	Systematic review and meta-analysis	
	Change in MVPA minutes in three groups	IV: Interventions DV: MVPA	MVPA minutes, PAVS	Weeks 1 - 12	Repeated Measures ANOVA, Paired t- test, and One-way ANOVA	
	Change in Self-Efficacy in three groups	IV: Interventions DV: Self-Efficacy	Exercise Self Efficacy Scale			
	Change in Self-Regulation in three groups	IV: Interventions DV: Self-Regulation	Self-Regulation Scale			
	Change in Social Support in three groups	IV: Interventions DV: Social Support	Support for Exercise Habits Scale	Week-0 and	One-way ANOVA	
	Change in Incentive Motivation in three groups	IV: Interventions DV: Incentive Motivation	Physical Activity Motivation Scale	week-8		
	Change in Mental Well-being in three groups	IV: Interventions DV: Mental Well-being	WHO-5			
2.1	Relationship between Physical Activity, Self-Efficacy, Self- regulation, Social Support, and Incentive Motivation (pre-test)	Exogenous: Self- Regulation Endogenous: Self- Efficacy, Social Support, and Incentive Motivation	PAVS, Exercise Self Efficacy Scale, Self-Regulation Scale, Support for Exercise Habits Scale, Physical Activity Motivation Scale,	Week-0	Path-analysis	
	Relationship between interventions, Physical Activity, Mental Well-being, Self-Efficacy, Self-regulation, Social Support, and Incentive Motivation (post- test)	IV: Interventions Mediators: Mental -Well- being, Self-Efficacy, Self- Regulation, Social Support, and Incentive Motivation DV: Physical Activity,	PAVS, Exercise Self Efficacy Scale, Self-Regulation Scale, Support for Exercise Habits Scale,	Week-8	Mediation models	

		Physica	l	
		Activity	r	
		Motivatio	on	
		Scale, WH	O-5	
		Multiple	e	
		choices an	nd	Decominitivo
2.2	Fatticipants experiences	open-end	ed Waak 8	A palvaga and
2.2		question	s week-o	Analyses and
	Participanta (accontability	Likert sca	lle	mematic analysis
	Farticipants acceptability	question	s	

3.2.6 Expected Findings

We expected to find the effects of text messaging and fitness competition on university students' Physical Activity, Mental Well-being, Self-Efficacy, Incentive Motivation, Self-Regulation, and Social Support. Participants expected to identify their enablers and barriers to practice physical activity, and general satisfaction with both interventions.

3.2.7 Potential Limitations and Alternative Strategies

First, due to the non-randomized and quasi-experimental nature of the study, potential for bias arising from differences in baseline risks and sampling bias between three groups could not be completely ruled out. The DFFC was open to everyone, but one member of each two-person team was required to be an employee or student of one of the three colleges. Hence, participants could not be randomly assigned to the competition-based intervention groups. Second, although we aimed to reduce the error inherent in PA measurement by using fitness trackers, there are potential self-report bias, especially for SCT constructs. Therefore, we used validated and clear instruments when framing questionnaires. Third, losses to follow-up could lead to incomplete study results and introduce bias if there were differences in likelihood of loss to follow-up related to exposure status and outcome. Therefore, participants received weekly reminders and incentives after completing all follow-up surveys. Fourth, we did not know if the participants actually read all, some, or none of text messages. However, EZ Texting provides delivery reports indicating whether the text message is delivered successfully.

65

Ethical Approval: Prior to conducting the study, an approval was received from Auburn University' and VCOM Institutional Review Board. All participants provided informed consent to participate in the study.

Chapter 4. Results

This chapter first presents the results of the systematic review and meta-analysis from Aim 1. Results for Aim 2 follow and include the effect of competition-based and text message interventions, as well as relationships between interventions, Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, Self-Regulation and Physical Activity. This is followed by results describing participants' experience and acceptability of the DFFC.

4.1 Aim 1: Systematic Review and Meta-Analysis

4.1.1 Systematic Review

The PRISMA diagram in Figure 4.1 depicts the study selection process. The initial literature searches retrieved a total of 344 records after removing duplicates and 47 full-text articles were screened for eligibility. Out of these, 20 articles were selected for the systematic review and six were selected for meta-analysis.





The details of the 20 included studies are described in Table 4.1. More than half (11, 50%) of included were published in the last five years (2018-2022). There were 15 (75%) RCTs, one focus group study, and four quasi-experimental studies (one study with a nonequivalent control group and three studies with one-group pretest–posttest design). The total number of participants in the RCTs and non-randomized controlled trials was 2,798, accounting for 93.57% of the total number of participants in the 20 included studies. Over half (12, 60%) of the studies lasted for 4-8 weeks, and two lasted over twelve months, while the longest duration was 24 months³¹⁹ and the shortest duration was 6 days.¹³³

Table 4.1. Characteristics of Included Studies

Author Year	Participants	N	Comparisons	Study design	Intervention	Results
Castro 2021	University students	9	N/A	pre-post one group	One-on-one session with automated text-messages based on COM-B components (4 texts per day for 6 days)	Significant reduction in sedentary time (-1.65 h/d, p = .005) and significant increase in standing ($+1.1$ h/d, p = .019) and stepping time ($+0.55$ h/d, p = .003) during weekend days.
Cotten 2016	University students	82	Daily text messages unrelated to sedentary behavior	RCT	Sedentary behavior related text messages twice daily for 6 weeks	Significant increase in LPA $(+74.34 \text{ m/d}, \text{p} = .07)$ for intervention group.
Dillon 2021	University students	114	N/A	RCT	One-on-one behavioral counseling session and tailored text messages twice daily	Significant group by time interaction effects favoring the intervention group for sitting time (p=0.004, $\eta_p^2 =$ 0.10), walking time (p=0.021, $\eta_p^2 = 0.06$) and stretching time (p=0.023, $\eta_p^2 = 0.08$).
Figueroa 2022	University students	93	Daily feedback messages	RCT	Daily motivational messages for 6 weeks	Sending text message initially resulted in an increase in daily steps (729 steps, $p = .012$).
Godino 2016	College students (18-35 yrs) have overweight or obesity	404	General information about health and wellness	RCT	Facebook, mobile apps, a website with blogs, e- mail, SMS (at least 1 text per week, customize frequency), and	Adjusted weight was significantly less in intervention group compared to control group at 6 months (-1.33 kg, p

					occasional 'lifeline' contact with a health	= .011) and 12 months (-1.33 kg, p = .008).
Hardan- Khalil 2021	University students	201	No intervention	Nonequival ent control group design	Daily motivational messages for 8 weeks	No significant interactions on PA outcomes.
Hayes 2021	University students (18-25 yrs)	95	Assignment of dietary weighing goal	RCT	Asked to form implementation intentions and 4 texts per week for 4 weeks	Decrease in weight (d = 0.35, p = .002), BMI (d = 0.33, p = .003), and caloric intake (d = 1.14, p < .001) across groups.
Keahey 2021	Undergraduate students (17– 25 yrs)	158	N/A	pre-post one group	4 texts per week for 4 weeks and 2-week follow- up	Significant decrease in weekly PA after intervention ($p < .001$) and follow-up ($p < .001$).
Lua 2013	University students (18-24 yrs)	417	No intervention	RCT	Conventional lecture, brochures, and texts (1 text per five days) for 10 weeks	Significant increase in MET spending on walking (d = .48), MPA (d = .38), VPA (d = .34), and total PA (d = .75), and decrease in MET for sitting (d = .35).
Mbada 2018	Female undergraduates (BMI>25)	64	Pedometer and text messages were sent as reminders	RCT	Pedometer and structured SMS thrice weekly for 8 weeks	Significant difference in weight (-0.92 ± 2.15 kg vs. 1.52 ± 2.10 kg P = 0.001) and BMI (-0.35 ± 0.84 kg/m ² vs. 0.58 ± 0.81 kg/m ²) at week four between the experimental and control group.
Muñoz 2014	University students enrolled	114	Pedometer	RCT	Text 2-3 times per week for 16 weeks with pedometer	No significant difference in daily steps ($p = .467$).

	in a wellness course					
Nam 2020	Female university students	64	No intervention	RCT	Social-media-based support through Fitbit Flex, Fitbit's smartphone application, text messages (2 texts per week), and e- mail for 28 days	Significant increase in total weekly MET ($p = .006$) for intervention group with significant difference (p = .010) compared with control group.
Napolitano 2013	University students (18-29 yrs and BMI between 25 and 50 kg/m ²)	52	1) Facebook group; and 2) control group (no intervention)	RCT	Facebook group, daily text, and weekly tailored feedback reports for 8 weeks	Facebook Plus group had significantly greater weight loss (-2.4 ± 2.5 kg) than the Facebook (-0.63 ± 2.4 kg) and control (-0.24 ± 2.6 kg) (both Ps < 0.05)
Napolitano 2021	University students (18-35 yrs and BMI between 25 and 45 kg/m ²)	460	General healthy body content	RCT	Facebook group, text message (personalized or generic, 2 texts per week), and weekly report for 18 months	Highly engaged member in TAILORED (personalized) group lost more weight compared to control group (-2.32 kg, p = .004) at 6 months.
Olofintuyi 2018	African American university students (18-25 yrs)	11	Participants did not respond	pre-post one group (with post-hoc grouping)	Text message (week 1, daily message; weeks 2–3, 5 text per day)	No statistically significant differences in PA variable between groups.
Prestwich 2009	University students	155	Motivational message or no intervention	RCT	Implementation intentions and SMS (customized frequency) for 4 weeks	Combined intervention increased exercise frequency significantly more than the other groups.
Reese 2017	African American women university students (19-30 yrs, BMI>25)	14	N/A	Focus group	1-2 texts per day for 2 weeks and 5 texts per week for 1 week	Participants preferred brief, specific, and time sensitive text messages.

Sandrick	Full-time students	60	No	RCT	Face-to-face meeting with	Significant difference in
2017	aged 18-30		intervention		a health coach and SMS	MET (2047 m/w for control
					(< 3 text/week) for 8	group, 3114 m/w for
					weeks	intervention group, $p = .04$)
St Quinton	First-year	289	Generic PA	RCT	Attitude messages, goal	Participants that received
2021	undergraduate		information 4		priority messages, or a	attitude messages had
	student (18-25y)		weeks		combination of these (6	significantly more positive
					texts in 2 weeks and 4-	attitudes ($p = .04$),
					week follow-up)	intentions $(p = .001)$, and
						rates of PA ($p = .001$).
Stark 2016	Health	134	Pedometer	RCT	Daily affective text	No significant difference in
	professional				messages for 2 weeks with	daily steps
	students				a pedometer	

Note: LPA = light-intensity PA; RCT = randomized controlled trial; SMS = short messages service;

There was a restriction on the age of the participants in ten of the studies. Among them, five studies recruited university students younger than 24 or 25 years old^{131,270,320-322}, and five studies recruited students younger than 30 or 35 years old^{140,319,323-325}. There were five studies targeting students who had overweight or obese^{319,323-326}, two studies targeting African American students^{131,325}, and three studies focusing on female students.^{271,325,326} Although the studies were open to all enrolled university students, three of the studies focused on undergraduates^{270,322,326}, while one was designed for students enrolled in a wellness course³²⁷, and one was for health professional students.³²⁸ Only one of 20 included studies was based on a health behavior theory (COM-B Model).¹³³

Text message interventions varied across studies in higher education settings. Text messages were used alone in seven studies^{131,137,269,270,322,325,329}, while some studies used text messages as part of other interventions, such as monitoring steps with a pedometer³²⁶⁻³²⁸, consulting sessions or lectures^{133,140,321,330}, forming implementation intentions^{320,331} (e.g., If there are stairs, then I will take them instead of the elevator), and social-media-based programs.^{271,319,323,324} In three studies, texts were sent more than once per day^{133,269,330}, but no more than four messages per day.¹³³ Texts were sent on a daily basis in six studies^{131,137,323,325,328,329} and were sent 1-4 times per week in nine studies.^{140,270,271,320-322,324,326,327} Participants in two studies were allowed to customize the frequency of receiving messages.^{319,331}

Figure 4.2 summarizes the overall characteristics of 20 articles in systematic review. Most of them are RCTs, and the population of ten studies are university students younger than 35 years old. Interventions and outcome measurement vary across studies.



Figure 4.2. Overall Characteristics of 20 Articles in Systematic Review

The included studies used various measures to assess the effectiveness of text message, including MET, activity minutes,

sitting time, walking time, standing time, steps, weight, BMI, and other clinical outcomes (Table 4.2).

	Randomized Controlled Trials									
	MET	Sitting Time	Walking Time	BMI	Standing Time	Steps	PA time	Weight		
Cotten 2016					Minutes (self- reported using SLIPA questionnaire)		LPA and MPA minutes (self-reported using 7-day PAR questionnaire)			
Dillon 2021		Minutes (self- reported using OSPAQ questionnaire)	Minutes (self- reported using OSPAQ questionnaire)		Minutes (self- reported using OSPAQ questionnaire)					
Figueroa 2022						Steps (mHealth app): Final GEE models presented but no numerical data provided				
Hayes 2021				BMI (objective measurement): Final ACOVA and paired t- test statistics presented but no numerical data provided						
Lua 2013	MET (self- reported using IPAQ)			BMI (measured objectively)				Weight (measured objectively)		
Mbada 2018				BMI (measured objectively)		Steps (measured objectively using pedometer)				
Muñoz 2014						Steps but SD unknown (measured objectively using pedometer)				
Nam 2020	MET (self- reported using IPAQ)									
Napolitano 2013							PA (self-reported using Godin-Shephard leisure- time physical activity questionnaire): Bonferroni correction	Height and weight differences (objective measurement)		

Table 4.2. Categories of Outcomes for 20 Included Studies

							statistics presented but no	
							numerical data provided	
Napolitano								Adjusted weight
2021								(measured
2021								objectively)
Godino				BMI (measured				
2016				objectively)				
Prestwich							Differences in exercise	
2009							behavior (self-reported	
	MET						items)	
	(self-							
Sandrick	reported							
2017	using							
	IPAQ)							
St Quinton							PA scores (self-reported	
2021						Ctore (management	items)	
						objectively using		
						pedometer). Final		
Stark 2016						t-test statistics		
						presented but no		
						numerical data		
-						provided		
	[Quasi-experime	ntal studies	1		
Hardan-		Minutes (self-					LPA and MPA minutes	
Khalil 2021		reported using		BMI (self-reported)			(self-reported IPAQ)	
2021		II AQ)		One group pre	nost study			
		Minutes (self-	Minutes (self-		Minutes (self-			
		reported using NWU	reported using NWU		reported using NWU			
Castro		questionnaire and	questionnaire and		questionnaire and			
2021		measured	measured		measured			
		objectively using	objectively using		objectively using			
		accelerometer)	accelerometer)		accelerometer)		MVDA minutes (solf	
Keahey		using PASB-O					reported using PASB-O	
2021		questionnaire)					questionnaire)	
	MET	questionnune)					questionnuire)	
Olofintuvi	(self-			PMI (mangurad				
2018	reported			objectively)				
2010	using			objectively)				
	IPAQ)			Omel's st				
Page 2017			Participante expras	Quantative	study	a targeted and time of	neitiva	
Reese 2017			Participants express	sed a preference for text n	lessages that were concis	e, targeteu, and time-se	clisiuve	

Note: IPAQ = International Physical Activity Questionnaire; NWU = Nightly-Week-U; OSPAQ = Occupational Sitting and Physical Activity Questionnaire; PAR = Physical Activity Recall; PASB-Q = Physical Activity and Sedentary Behavior Questionnaire; SLIPA = sedentary and light intensity physical activity

The MET was measured by the self-reported International Physical Activity Questionnaire (IPAQ) and was calculated using the following formulas:

Total MET (min/week) = Walking MET + Moderate MET + Vigorous MET

Walking MET = 3.3 * Walking minutes * Walking days

Moderate MET = 4.0 * MPA minutes * MPA days

Vigorous MET = 8.0 * VPA minutes * VPA days

Activity minutes, sitting time, walking time, standing time, and PA times were measured subjectively using self-reported surveys, except for in the study conducted by Castro et al., which reported both accelerometer-based and self-reported measurements for sitting time, walking time, and standing time.¹³³ Steps were measured using pedometers or mHealth apps in all studies where steps were included as an outcome. Objective measurements of BMI or weight were used in most studies, with the exception of two studies: one conducted by Stark et al. and another conducted by Hardan-Khalil et al., where self-reported BMI was used.^{137,328}

Of the 15 RCTs, 13 reported significant improvements in PA in different aspects^{140,269,271,319-324,326,329-331}, and two studies^{133,328} did not report significant differences between intervention and control groups. Muñoz et al. assessed the effectiveness of text message interventions for participants who were enrolled in a wellness course.³²⁷ The intervention group received texts about healthy behaviors two to three times per week over 16 weeks of the semester³²⁷; while Stark's study was among health profession students, and participants received two texts per week for 2 weeks.³²⁸ Both studies used a pedometer to measure daily steps and did not find a significant difference between the intervention group and the control group.^{327,328} Half of the four quasi-experimental studies found that the intervention led to a significant increase in PA. Castro et al. developed one educational session and 24 text messages based on COM-B components, and found significant post-test reduction in sedentary time (-1.65 h/d, p = .005) and significant increases in standing (+1.1 h/d, p = .019) and stepping time (+0.55 h/d, p = .003) during weekend days.¹³³ In Keahey's study, texts were sent four times per week for 4 weeks. A significant decrease was found in weekly PA after intervention (p < .001) and follow-up (p < .001).²⁷⁰ Regarding the one qualitative study by Reese et al., a focus group was conducted among African American female university students under 30 years old with BMI>25, and found that brief, specific, and time sensitive text messages were preferred.³²⁵ Overall, commonalities across the included studies that contributed to positive findings included: (1) personalized or tailored messages, (2) customized frequency, (3) relevant and engaging content that was comprehensible, and (5) integration into other programs.

The results of quality assessment using MMAT are shown in Table 4.3. Ratings from the MMAT indicate that all included 15 RCTs performed randomization appropriately and provided complete outcome data, and all but one of the RCT studies assessed baseline characteristics across groups. Among five of the RCTs, research personnel were unaware of intervention allocation, and four RCTs reported more than a 30% drop-out rate. For three quantitative, nonrandomized trials, only one met all five quality criteria, while two did not account for the confounders in the design and analysis. All quality criteria were met in one qualitative study and one mixed methods study.

Table 4.3. Article Quality Assessment

Quantitative randomized controlled trials										
Author Year	Is randomization	Are the groups	Are there	Are outcome	Did the participants adhere					
	appropriately	comparable at	complete	assessors blinded	to the assigned					
	performed?	baseline?	outcome data?	to the intervention	intervention?					
				provided?						
Cotten 2016	Yes	Yes	Yes	No	No 31% drop out					
Dillon 2021	Yes	Yes	Yes	Yes	Yes					
Figueroa 2022	Yes	No	Yes	No	Yes					
Hayes 2021	Yes	Yes	Yes	Yes	Yes					
Lua 2013	Yes	Yes	Yes	Can't tell	Yes					
Mbada 2018	Yes	Yes	Yes	Can't tell	Yes					
Muñoz 2014	Yes	Yes	Yes	Yes	No about 45% drop out					
Nam 2020	Yes	Yes	Yes	Can't tell	Yes					
Napolitano 2013	Yes	Yes	Yes	Can't tell	Yes					
Napolitan 2021	Yes	Yes	Yes	Yes	No 38% drop out					
Godino 2016	Yes	Yes	Yes	Yes	Yes					
Prestwich 2009	Yes	Yes	Yes	Can't tell	Yes					
Sandrick 2017	Yes	Yes	Yes	No	Yes					
St Quinton 2021	Yes	Yes	Yes	Can't tell	No 40% drop out					
Stark 2016	Yes	Yes	Yes	Can't tell	Yes					
		Quantitative no	onrandomized stu	dies						
	Are the	Are	Are there	Are the	During the study period, is					
	participants	measurements	complete	confounders	the intervention					
	representative of	appropriate	outcome data?	accounted for in	administered (or exposure					
	the target	regarding both the		the design and	occurred) as intended?					
	population?	outcome and		analysis?						
		intervention (or								
		exposure)?								
Hardan-Khalil 2021	Yes	Yes	Yes	No	Yes					
Keahey 2021	Yes	Yes	Yes	Yes	Yes					

Olofintuyi 2018	Yes	Yes	Yes	No	Yes				
Qualitative study									
	Is the qualitative	Are the	Are the findings	Is the interpretation	Is there coherence between				
	approach	qualitative data	adequately	of results	qualitative data sources,				
	appropriate to	collection	derived from	sufficiently	collection, analysis and				
	answer the	methods adequate	the data?	substantiated by	interpretation?				
	research question?	to address the		data?					
		research							
	question?								
Reese 2017	Yes	Yes	Yes	Yes	Yes				
		Mix	ed methods						
	Is there an	Are the different	Are the outputs	Are divergences	Do the different				
	adequate rationale	components of the	of the	and inconsistencies	components of the study				
	for using a mixed	study effectively	integration of	between	adhere to the quality				
	methods design to	integrated to	qualitative and	quantitative and	criteria of each tradition of				
	address the	answer the	quantitative	qualitative results	the methods involved?				
	research question?	research	components	adequately					
		question?	adequately	addressed?					
			interpreted?						
Castro 2021	Yes	Yes	Yes	Yes	Yes				

4.1.2 Meta-analysis

Meta-analysis was conducted of each outcome if there were at least three studies reporting the same outcome using the same study design. Therefore, two PA outcomes in six RCTs were included for the meta-analysis: total MET (n=3) and BMI (n=3). The Higgins I^2 statistics for both meta-analyses were lower than 75%, indicating no statistical heterogeneity between the studies and suggesting the use of fixed effects models.

There were three RCTs included in the analysis of intervention effect on total MET. When compared with control groups that did not receive PA text messages, text message interventions led to a greater total MET (standardized mean difference [SMD] = 0.67, 95% CI = 0.49, 0.85, p < 0.01; Figure 4.3) in the intervention groups. MET was measured subjectively by IPAQ in these three RCTs. However, for the analysis of intervention effects on BMI, statistically significant differences were not observed in three RCTs (SMD = -0.15, 95% CI = -0.32, 0.02, p = 0.08; Figure 4.4). BMI was measured objectively in these three RCTs.

Figure 4.3. Im	pact of Text Message	Interventions on weekly	/ total MET (RCTs, n=3)
<i>(</i>)				

	Exp	perimental		(Control			Std. Mean Difference		Std. Mean	Difference	
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Fixed, 95% CI		IV, Fixe	d, 95% Cl	
Lua 2013	4,419.3	2,413.507	178	2,921.6	1,937.187	202	75.2%	0.69 [0.48, 0.90]				
Nam 2020	3,762.84	2,988.6	32	2,146.84	1,586.23	32	12.7%	0.67 [0.16, 1.17]				
Sandrick 2017	3,144	2,324	30	2,074	1,410	30	12.1%	0.55 [0.03, 1.07]				
Total (95% CI)			240			264	100.0%	0.67 [0.49, 0.85]			•	
Heterogeneity: Chi ² = 0 Test for overall effect:	0.24, df = 2 Z = 7.29 (P	(P = 0.89); < 0.00001)	l² = 0%						-2	-1 Control	0 1 Experimental	2

Figure 4.4. Impact of Text Message Interventions on BMI (RCTs, n=3)

	Exp	periment	al		Control			Std. Mean Difference		Std. Mean Difference	
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Fixed, 95% C		IV, Fixed, 95% CI	
Godino 2016	28.6	2.1112	50	28.9	0.3519	50	18.9%	-0.20 [-0.59, 0.20]			
Lua 2013	21.2	3.8691	178	21.57	4.2638	202	71.9%	-0.09 [-0.29, 0.11]			
Mbada 2021	27.4	3.22	25	28.9	2.11	25	9.1%	-0.54 [-1.11, 0.02]	←		
Total (95% CI)			253			277	100.0%	-0.15 [-0.32, 0.02]		•	
Heterogeneity: Chi ² = 2	2.24, df	= 2 (P = 0	0.33); l [;]	² = 11%					1		
Test for overall effect:	Z = 1.74	(P = 0.0	8)						-1	Experimental Control	I

4.2 Aim 2.1

4.2.1 Study Enrollment

As described in Chapter 3, the study invitation, which included the 2022 DFFC kickoff emails, was sent to all HCOP, AUCON, and VCOM-Auburn students, approximately 1,349 students total. A total of 260 participants in the DFFC were then recruited for this study, and 58 enrolled (Table 4.4). Interested students registered for the study via a sign-up survey in Qualtrics, distributed by the investigator's major professor. Among 58 participants, 48 (82.76%) students contributed to the initial baseline survey, and 52 (89.66%) students completed the second survey. More than 70% in the DFFC Group and DFFC + Text Group submitted weekly MVPA data during the follow-up weeks, in contrast to a loss of over 70% in the Control Group. In DFFC + Text Group, one person dropped out after enrolling, so physical activity data are not reported for this individual. In the Control Group, four participants did not sign up for their ChallengeRunner account, so their physical activity data were not included in the final analysis.

Table 4.4. Response	Rate	for	Two	Surveys
---------------------	------	-----	-----	---------

	Enrolled in study	Completed 1 st survey	Registered for a ChallengeRunner account	Completed 2 nd survey	Follow-up
Group 1: Control	19	15 (78.95)	16 (84.21%)	16 (84.21%)	5 (26.32%)

Group 2: DFFC	19	16 (84.21%)	19 (100%)	16 (84.21%)	14 (73.68%)
Group 3: DFFC+ Text	20	17 (85%)	19 (95%)	20 (100%)	14 (70.00%)
Total	58	48 (82.76%)	53 (91.38%)	52 (89.66%)	33 (56.90%)

4.2.2 Descriptive Results

Baseline characteristics of participants are presented in Table 4.5. Most participants were female (77.08%), non-Hispanic (95.83%), enrolled in HCOP (79.17%), and had adequate health literacy (89.58%). The Control group consisted of students who did not enroll in DFFC, while DFFC participants who enrolled in the study were randomized to the DFFC Group or the DFFC + Text Group. Therefore, we conducted two sample t-test (Mean Age, Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation) and Fisher's Exact test (categorical variables) to compare the baseline characteristics between the non-enrolled and nonrandomized participants (Control Group) and those who enrolled and were randomized (DFFC Group and DFFC + Text Group). The mean age of all students was 24.87 (range: 20-40, SD 4.21), while the average age of participants in the Control Group was significantly higher than groups two and three combined (p = 0.0013). Among all participants, 70.83% of them identified themselves as white, but the proportion of Asian students was significantly higher in Control Group than groups two and three combined (p = 0.0003). At baseline, there was no statistically significant difference between the three groups in Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, or Self-Regulation.

Characteristics	Full cohort	Control	DFFC	DFFC +	P-value (Control
	(n=48)	Group	Group	Text Group	Group vs. two
		(n=15)	(n=16)	(n=17)	intervention groups)
Mean Age (years), range,	24.87, 20-	27.79, 22-	24.62, 20-	23.24, 21-	0.0013
SD	40, 4.21	38, 4.26	40, 4.56	30, 2.41	
Sex (n, %)					
Female	37	10 (66.67)	13 (81.25)	14 (82.35)	0.2831
	(77.08%)				
Race (n, %)					0.0003
Asian	11 (22.92)	8 (55.33)	0	3 (17.65)	
Black, African	3 (6.25)	2 (13.33)	0	1 (5.8)	
American					
White, Caucasian	34 (70.83)	5 (33.33)	16 (100)	13 (76.47)	
Ethnicity (n, %)					1.0000
Other Hispanic,	2 (4.17)	0	2 (12.5)	0	
Latino(a), or			. ,		
Spanish origin					
Not of Hispanic	46 (95.83)	15 (100)	14 (87.5)	17 (100)	
origin					
Health Literacy (n, %)					0.3070
Adequate	43 (89.58)	12 (80)	15 (93.75)	16 (94.12)	
Marginal	5 (10.42)	3 (20)	1 (6.25)	1 (5.88)	
Enrolled Program (n, %)					0.2770
НСОР	38 (79.17)	12 (80)	12 (75)	14 (82.35)	
AUCON	4 (8.33)	0	2 (12.5)	2 (11.76)	
VCOM	6 (12.5)	3 (20)	2 (12.5)	1 (5.88)	
Enrolled in any other					0.5421
organized PA program					
(n, %)					
Yes	7 (14.58)	0	0	3 (17.65)	
No	41 (85.42)	15 (100)	16 (100)	14 (82.35)	
Mental Well-being (mean, Sl	D)	53.4 (19.0)	63.2	58.6 (12.48)	0.1188
			(14.22)		
Incentive Motivation (mean,	SD)	3.41 (0.95)	3.99 (0.39)	3.77 (0.65)	0.0925
Social Support (mean, SD)		2.4 (1.07)	2.36 (0.6)	2.22 (0.74)	0.7086
Self-Efficacy (mean, SD)		2.31 (0.89)	2.55 (0.62)	2.37 (0.57)	0.5012
Self-Regulation (mean, SD)		2.33 (0.82)	2.18 (0.6)	2.16 (0.76)	0.4636

Table 4.5. Participants' Baseline Characteristics

4.2.3 Path Analysis for Pretest Survey Data

A path analysis model was developed to understand the relationship between Self-Regulation, Incentive Motivation, Self-Efficacy, Social Support, and Physical Activity using the baseline survey data (Figure 4.5). In this model, the arrows indicate the direct effect of Social Support, Self-Regulation, Incentive Motivation, and Self-Efficacy on the endogenous variables (Self-Regulation, Incentive Motivation, Self-Efficacy, and Physical Activity). As shown in Table 4.6, this model has a mediocre but acceptable fit.

Figure 4.5. Path Analysis of the Effect of Framing on Self-Regulation, Incentive Motivation, Self-Efficacy, Social Support, and Physical Activity (Pre-test Model)



Table 4.6. Pre-test Model Fit Information

Chi-Square Test of Model Fit		
	Value	1.264
	Degrees of Freedom	1
	P-Value	0.2609
RMSEA (Root Mean Square Error Of Approximation)		
	Estimate	0.074
	90 Percent C.I.	0.000
		0.400
	Probability RMSEA	0.289
	<= .05	
CFI/TLI		
	CFI	0.996

	TLI	0.960
Chi-Square Test of Model Fit for the Baseline Model		
	Value	75.771
	Degrees of Freedom	10
	P-Value	0.0000
SRMR (Standardized Root Mean Square Residual)		
	Value	0.022
R-Square	Observed Variable	Estimate
	PA	0.325*
	SE	0.222*
	IM	0.043

Table 4.7 shows the standardized total, direct, and indirect effects between variables. Self-Efficacy and Incentive Motivation have a medium direct and significant total effect on Physical Activity, while Social Support and Self-Regulation have a small and non-significant direct and total effect on Physical Activity. Self-Efficacy and Social Support have a large and significant effect on Self-Regulation, and Incentive Motivation has a medium and significant effect on Self-Efficacy.

Table 4.7. Standardized Total and Direct Effects of Pre-Test Model

	Physical	Self-	Self-	Incentive
	Activity	Regulation	Efficacy	Motivation
	Standa	ardized Total Effe	cts	
Self-Efficacy	0.330*	0.570*	-	-
Incentive	0.429*	-	0.461*	-
Motivation				
Social Support	0.254	0.509*	0.135	0.208
Self-Regulation	0.008	-	-	-
	Standa	ardized Direct Effe	ects	
Self-Efficacy	0.325*	0.570*	-	-
Incentive	0.277*	-	0.461*	-
Motivation				
Social Support	0.148	0.433*	0.039	0.208
Self-Regulation	0.008	-	_	-

Table 4.8 shows the indirect effects between variables. Incentive Motivation has a small but significant indirect effect on Physical Activity through two paths: IM->SE->PA and IM->SE->SR->PA. In addition, Social Support has a small and non-significant indirect effect on Physical Activity and Self-Efficacy.

Table 4.8. Standardized Indirect Effects of Pre-Test Model

Standardized Indirect Effects								
	Physical Activity		Self-Regulation		Self-Ef	ficacy	Incentive Motivation	
Self-Efficacy	SE-SR-PA	0.005	0.00	0	-		-	
Incentive	IM-SE-PA	0.150	-		0.0	00	-	
Motivation	IM-SE-SR-	0.002						
	PA							
	Total	0.152*						
Social Support	SS-SE-PA	0.013	SS-SE-	SS-SE- SR 0.022		0.096	0.000	
	SS-IM-PA	0.058	SR					
	SS-SR-PA	0.004						
	SS-IM-SE-	0.031	SS-IM-	0.055				
	PA		SE- <mark>SR</mark>					
	SS-SE-SR-	0.000						
	PA							
	SS-IM-SE-	0.0000						
	SR-PA							
	Total	0.106	Total	0.077				

4.2.4 Effects of the DFFC and Text Message Interventions on PA

Figure 4.6 presents the weekly MVPA for all three groups from the 0 to 12 weeks of the DFFC. The week-0 data were self-reported as there were from the first survey, which was sent at the beginning of the DFFC, while the week 1–12 data were gathered via the ChallengeRunner app. From the start of the DFFC until the follow-up weeks (weeks 9-12), the Control Group's MVPA remained stable at around 100 minutes per week. However, with the commencement of the DFFC, both the DFFC Group and DFFC + Text Group experienced an increase in their

weekly MVPA. The DFFC + Text Group's MVPA increased until the 8th week, whereas the DFFC Group's MVPA decreased in the third week and again during week 8 of the DFFC.



Figure 4.6. Weekly MVPA Minutes Weeks 0-12

Repeated Measures ANOVA was conducted to test the equality of means between and within subjects. During the 8-week competition (Table 4.9), the effect of the DFFC was significant in weekly MVPA (p = 0.0006), indicating there was a significant difference in MVPA between the control group (who did not enroll in the competition) and the participants who enrolled in the competition (DFFC Group and DFFC + Text Group). However, the effect within the subject (Table 4.10) was not significant (p = 0.6917), meaning that there was no significant change observed within each group across the competition (week 1-8).

When considering both the 8-week competition *and follow-up weeks* (Table 4.11), the effects of both the DFFC (p = 0.1149) and text message interventions (p = 0.8439) were not significant in weekly MVPA, suggesting that there was no significant difference in MVPA between the control group (who did not enroll in the competition) and the participants who

enrolled in the competition (DFFC Group and DFFC + Text Group), as well as between the DFFC + Text Group and participants who did not receive text messages (control group and DFFC group). However, the MVPA changed significantly within each group over the entire 12-week period (Table 4.12, p = 0.0009), encompassing both the competition and follow-up phases. This indicates that within each group, there was a significant change in MVPA levels over time (week 1-12).

Also, it is worth noting that the change in mean MVPA across time depended on texting (week 1-8: p = 0.0107; week 1-12: p = 0.0410). Specifically, there was a statistically significant interaction between time and text message intervention on mean MVPA, indicating that the texting intervention had a differential effect on MVPA over time.

Table 4.9. Repeated Measures ANOVA for Between Subjects Effects (DFFC and text, week 1-8)

Source	DF	Type III SS	Mean Square	F Value	Pr > F
Text	1	3820.17	3820.17	0.01	0.9058
DFFC	1	3673605.41	3673605.41	13.64	0.0006
Error	42	11313484.12	269368.67		

Table 4.10. Repeated Measures ANOVA for Within Subject Effects (Time, week 1-8)

Source	DF	Type III SS	Mean Square	F Value	Pr > F	Adj Pr > F	
						G - G	H-F-L
time	7	83293.756	11899.108	0.68	0.6917	0.5938	0.6082
time*Text	7	329136.122	47019.446	2.67	0.0107	0.0395	0.0340
time*DFFC	7	118220.610	16888.659	0.96	0.4602	0.4246	0.4304
Error(time)	294	5168497.139	17579.922				

Table 4.11. Repeated Measures ANOVA for Between Subjects Effects (DFFC and text, week 1-

12)

Source	DF	Type III SS	Mean Square	F Value	Pr > F
Text	1	7209.707	7209.707	0.04	0.8439
DFFC	1	485228.340	485228.340	2.66	0.1149
Error	26	4740043.860	182309.379		

Table 4.12. Repeated Measures ANOVA for Within Subject Effects (Time, week 1-12)

Source	DF	Type III SS	Mean Square	F Value	Pr > F	Adj Pr > F	
						G - G	H-F-L
time	11	638502.424	58045.675	2.99	0.0009	0.0185	0.0115
time*Text	11	402797.512	36617.956	1.89	0.0410	0.1115	0.0953
time*DFFC	11	277952.035	25268.367	1.30	0.2232	0.2720	0.2646
Error(time)	286	5555641.483	19425.320				

One-way ANOVA was conducted for each week to test the mean of weekly MVPA between the three groups (Table 4.13). From 0-week to 8-week, the mean weekly MVPA was significantly different between the Control and DFFC Group, as well as between the Control and DFFC + Text Group. However, there was no statistically significant difference observed between the two intervention groups. During the follow-up weeks, there was no significant difference found among all three groups.

A paired t-test was performed to compare the weekly MVPA within each week to the first week of the competition ("Within Group" in Table 4.12). There was no statistically significant difference compared to the 1st week for the Control Group. In the DFFC Group, there was no significant difference observed until the 10th week, when weekly MVPA was significantly lower than week 1. In the DFFC + Text Group, the weekly MVPA was significantly
higher than that of the 1st week from week-4 to week-8, but it significantly decreased beginning in week-9.

Table 4.13.	Weekly	MVPA	Minutes	Weeks	0-12
-------------	--------	------	---------	-------	------

				Between Group			Within Group					
							Comparisons significant at the 0.05 level are indicated by ***					
Week	Control	SD	DFFC	SD	DFFC + Text	SD	Control -DFFC	Control -DFFC + Text	DFFC-DFFC + Text	Control	DFFC	DFFC + Text
0	80.00	124.27	157.188	81.5673	150.88	90.87	***	***				
1	75.07	103.47	293.32	215.69	203.38	125.92	***	***				
2	67.50	112.45	354.01	298.18	256.35	186.00	***	***		0.6205	0.3575	0.1615
3	83.43	107.06	365.72	294.40	264.15	194.04	***	***		0.2636	0.4262	0.192
4	68.46	122.42	317.22	279.37	304.42	215.15	***	***		0.795	0.9012	0.0451
5	67.29	120.38	284.47	311.02	311.19	241.94	***	***		0.5131	0.9014	0.049
6	64.43	104.16	282.06	210.92	315.31	241.65	***	***		0.5259	0.8023	0.0443
7	58.71	86.25	253.28	285.41	326.26	219.05	***	***		0.3319	0.5373	0.0337
8	30.57	117.56	371.06	358.43	355.38	251.81	***	***		0.0797	0.5208	0.0232
9	84.20	126.82	131.71	136.37	71.30	64.87				0.673	0.0534	0.0001
10	90.60	182.89	108.21	109.02	113.27	124.32				0.4518	0.0194	0.0041
11	103.40	25.04	170.21	165.01	95.16	93.40				0.4447	0.0329	0.0004
12	21.80	124.27	38.36	44.67	13.43	16.32				0.227	0.0005	<.0001

4.2.5 Effects of the DFFC and Text Message Interventions on Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation.

One-way ANOVA and mediation models were performed to determine if Mental Wellbeing, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation were significantly different between three groups using data from the second survey. Table 4.14 presents the posttest scores and the difference between posttest and pretest scores for the three groups. Only the Incentive Motivation score was significantly different between the three groups.

Mental Well-being							
	Control Group	DFFC Group	DFFC + Text	P-value			
			Group				
Posttest score	58.6 (22.83)	66.6 (18.83)	66.2 (13.84)	0.373			
Post test score minus pretest	5.2	3.4	7.6				
	Incentive Motivation						
Posttest score	3.24 (1.13)	4.17 (0.57)	3.82 (0.59)	0.006			
Post test score minus pretest	-0.17	0.18	0.05				
Social Support							
Posttest score	2.37 (0.84)	2.82 (1.01)	2.50 (1.03)	0.647			
Post test score minus pretest	-0.03	0.46	0.28				
Self-Efficacy							
Posttest score	2.05 (0.79)	2.78 (0.98)	2.65 (0.90)	0.058			
Post test score minus pretest	-0.26	0.23	0.23				
Self-Regulation							
Posttest score	2.18 (1.33)	2.58 (1.19)	2.40 (0.83)	0.559			
Post test score minus pretest	-0.15	0.40	0.24				

Table 4.14. Scores for Mental Well-being and Social Cognitive Outcomes After Intervention

Table 4.15 shows the results of one-way ANOVA, indicating a significant difference for Incentive Motivation between Control Group and DFFC Group (95% CI, 0.2561-1.6048), but not between DFFC Group and DFFC + Text Group (95% CI, -0.2984-0.9812) or between Control Group and DFFC + Text Group (95% CI, -0.0507-1.2288). Table 4.15. One-way ANOVA with Multiple Comparisons for Mental Well-being and Social

Cognitive Outcomes

Mental Well-being					
Group Comparison	n Difference Between Means 95% CI				
DFFC - DFFC + Text	0.0150	-0.7307	0.7607		
DFFC - Control	0.4000	-0.3860	1.1860		
DFFC + Text - Control	-0.0150	-0.7607	0.7307		
	Incentive Motivation				
DFFC - DFFC + Text	0.3414	-0.2984	0.9812		
DFFC - Control	0.9305	0.2561	1.6048		
DFFC + Text - Control	0.5891	-0.0507	1.2288		
Social Support					
DFFC - DFFC + Text	0.3127	-0.4986	1.1240		
DFFC - Control	0.2060	-0.6492	1.0611		
DFFC + Text - Control	-0.1067	-0.9180	0.7046		
Self-Efficacy					
DFFC - DFFC + Text	0.1295	-0.5956	0.8546		
DFFC - Control	0.7300	-0.0343	1.4943		
DFFC + Text - Control	0.6005	-0.1246	1.3256		
Self-Regulation					
DFFC - DFFC + Text	0.1851	-0.6714	1.0416		
DFFC - Control	0.4048	-0.4981	1.3077		
DFFC + Text - Control	0.2197	-0.6369	1.0762		

Figures 4.7-4.101present the mediation models for Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation. For each model, the independent variable is Group, the dependent variable is Physical Activity, and the mediator variables are Mental Well-being, Incentive Motivation, Self-Efficacy, Social Support, and Self-Regulation, respectively. Group impacts Physical Activity in all five models, as evidenced by the significant relationship between Group and PA in all models. The effect of Group is mediated by Self-Efficacy (Figure 4.8) and Incentive Motivation (Figure 4.9), but there is no evidence that this effect is mediated by Mental Well-being, Social Support, or Self-Regulation.





Figure 4.8. Mediation Model for Self-Efficacy



Figure 4.9. Mediation Model for Incentive Motivation



Figure 4.10. Mediation Model for Self-Regulation



Figure 4.11. Mediation Model for Social Support



4.3 Aim 2.2

4.3.1. Feedback and Satisfaction in DFFC

DFFC participants identified several positive and negative factors influencing their adherence and commitment to regular exercise (Figure 4.12). They were allowed to pick multiple or all factors that influenced their participation from the given list. A majority of participants (86.11%) in the DFFC Group and the DFFC + Text Group indicated that their exercise was motivated by the competitive nature of DFFC. Working with teammates (69.44%) and weekly activity logging (58.33%) were also reported by students to motivate their exercise, as well as prizes throughout the DFFC (50%), group fitness activities (47.22%), and announcement of previous week's winners and their data (27.78%). In addition, the weekly motivational email (19.44%) and marketing materials (2.78%) were mentioned by some students.

Figure 4.12. Motivators of Exercise Among DFFC Participants



As shown in Figure 4.13, the primary barriers to exercise identified by participants included insufficient time for exercising (91.67%), poor weather (47.22%), and decreased motivation (38.89%). Other barriers that made doing or scheduling exercise difficult during DFFC included lack of skills and knowledge (16.67%), lack of support (11.11%), lack of access to appropriate facilities (11.11%), negative PA experiences (11.11%), and unsafe neighborhoods (8.3%).





Responses to two open-ended questions provided insight into aspects of the 2022 DFFC participants enjoyed the most and which aspects they would like to change (Table 4.16). Many students described how they enjoyed the group fitness activities and working together with teammates, peers, and friends, which kept them motivated and engaged. For some students, participating in the fitness competition improved their awareness of their fitness and wellness. The ChallengeRunner leaderboard feature was also popular, as they were excited to see their

minutes and rankings compared to the other participants. However, some students were discouraged when they were unable to keep up with other teams' accomplishments, and some students doubted the methods of ensuring the authenticity and trustworthiness of the minutes data. Other suggestions included incorporating more activities (such as workout videos, strength exercises, and time management skills) into the DFFC and optimizing the timeframe of the DFFC to make it more manageable.

	Aspect of 2022 DFFC	Excerpt from responses
	Group fitness	I enjoyed the fitness classes. It was nice to try things I have never done before.
	activities	I enjoyed the fact that I had a teammate that kept me motivated and doing group fitness activities!
		It always motivates me when I have other people doing it with me.
	Working with teammates	I enjoyed the fact that I had a teammate that kept me motivated and doing group fitness activities!
		I enjoyed communicating with my team-mate on the exercise we had been doing
	Competitive atmosphere	I enjoyed seeing the leaderboard and seeing where other teams were at for motivation
Things		I enjoy tracking my minutes and seeing our team go up the rankings
Things they like		Having a partner and seeing the leaderboard over time.
ulcy like		I was able to see how much I actually work out
	Minutes logging	(I like) logging minutes with my partner/ registering minutes completed each day
		I enjoyed logging the minutes because it was rewarding/positive reinforcement!
	Amoronog of fitness	I enjoy the fitness challenge because it keeps me aware of my movement, even if I'm unable to do a physical activity for that day, I can recognize when I haven't moved enough during the day
	and health	I enjoyed the competition. I thought that I was going to do much better. I am however grateful for the challenge because it prompted me to visit my doctor when I wasn't losing any weight and realized I have issues with my thyroid causing me to gain excessive weight, so now it's

Table 4.16. Excerpts from Open-Ended Questions

		being treated appropriately. Even though I didn't lose
		weight I still enjoyed the fun of it!
		We got discouraged being able to see what the other teams
	Competitive	were doing each day or week. When we felt we couldn't
	atmosphere	catch up, we lost motivation to try.
		(I) was discouraging to see them put up such high numbers.
		Maybe include optional workout videos to follow
		Easier way to ensure people are truthful about minutes
		tracked.
		The app is very outdated
	Complaint about the	Some kind of accountability measure. I like being able to
	ChallengeRunner	enter in my own minutes because I wasn't always able to
	ann	wear my watch when I was doing physical activities.
	црр	However, people were abusing this because I found it hard
		to believe that 2 people were averaging 2 1/2 hours of
Ways to		exercise per day consistently over multiple weeks when at
improve		least one of the members was in school. This just didn't
mpiove		seem very realistic to me
		Offer time management skills on how to incorporate
		exercise in your daily routine
	More activities in	I think it would be nice to have a group fitness class on
		learning about the movements of strength exercises. Like
	DFFC	how to do exercises and when to do them for optimal
		benefits
		I think it would be interesting to somehow include the
-		hours slept in the Dean's Challenge.
		I think making the DFFC challenge a longer period of time
		would be a great idea. It kept us motivated so the longer the
	Timeframe	better.
		I would make it 6 weeks instead of 8 weeks. It is hard to
		keep up motivation and not get bored after that long!

Among 36 DFFC participants in DFFC Group and the DFFC + Text Group, 33 (91.67%)

students were very satisfied or somewhat satisfied with the 2022 DFFC (Figure 4.14), and 31

students stated that they were extremely likely to participate in the 2023 DFFC, if eligible

(Figure 4.15).





Figure 4.15. Intentions to Participant in 2023 DFFC



4.3.2 Feedback and Satisfaction of Text Message Interventions

Among the 20 participants in the DFFC + Text Group who received three text messages each week, 11 (55%) students reported that the messages were very helpful or helpful in *motivating them to start* exercising, but less than 40% of students reported that messages were very helpful or helpful in *motivating them to do more* exercise (35%) and helping them *achieve the recommended PA goal* (25%) (Figure 4.16).





Figure 4.17 indicates that the overall satisfaction with text message interventions was

55% of 20 participants. There were 16 (80%) students who were extremely or somewhat satisfied, while two (10%) students were extremely or somewhat unsatisfied with the frequency of text messages, which was three messages per week.

Figure 4.17. Overall Satisfaction of Text Message Interventions



4.4 Summary

In summary, both competition-based and text message interventions were found to be effective and acceptable in promoting physical activity among university students. During the DFFC competition, the mean weekly MVPA was significantly different between the Control Group and two intervention groups. There was statistically significant difference in postintervention Incentive Motivation among three groups, but not for Mental Well-being, Social Support, Self-Efficacy, and Self-Regulation. Self-Efficacy and Incentive Motivation have a medium direct and significant effect on Physical Activity. The effect of interventions was mediated by Self-Efficacy and Incentive Motivation.

Chapter 5. Discussion

The objectives of this study were 1) to conduct a systematic review and meta-analysis of text message interventions targeting physical activity among university students from published literature, and 2) to assess and compare the effectiveness of text message interventions and a competition-based intervention in improving physical activity among university students, and evaluate participants' acceptability and experiences of the text message interventions and the competition-based intervention. The findings of the first objective were used to develop text messages used in the second aim. Objective two was accomplished through a quasi-experimental study among HCOP, AUCON, and VCOM-Auburn students. The interventions were eight weeks long with four weeks of follow-up, and the primary outcome (weekly MVPA) was measured repeatedly for 12 weeks. This chapter discusses the findings and implications of each aim.

5.1 Aim 1

This aim involved a systematic review and meta-analysis of literature focusing on text message interventions in promoting PA among university students. The results indicated that text message interventions have a significant, medium effect on MET (SMD = 0.67, 95% CI = 0.49, 0.85, p < 0.01) according to Cohen's criteria.⁴³ However, positive effects of text message among university students have not been observed on BMI (SMD = -0.15, 95% CI = -0.32, 0.02, p = 0.08).

5.1.1 Discussion of Systematic Review and Meta-Analysis Findings

The findings of the current study provide a valuable extension of recent systematic reviews and meta-analysis. Previous studies of text message interventions predominantly focused on the general population¹⁷⁹, or a broader group of interventions.^{144,332} In 2013, Buchholz et al.

conducted a systematic review on PA text message interventions in adults. The median effect size for ten included studies was 0.5.¹⁷⁹ Maselli et al. synthesized the evidence from 27 controlled trials among university students in 2018, and 16 of included studies reported increased PA levels.¹⁴⁴ Peng et al. updated the search in 2022, focusing on e-health interventions on PA and sedentary behavior in college students.³³² There were 22 RCTs included, but only two of them included text message.^{326,330} A small-to-medium but significant improvement was found in PA for the experimental group at post-intervention compared to the control group (SMD = 0.32, 95% CI: 0.19, 0.45, p < 0.001). Specifically, positive effects were observed on total PA (SMD = 0.34, 95% CI: 0.10, 0.58, p = 0.005), MVPA (SMD = 0.17, 95% CI: 0.01, 0.32, p = 0.036), and steps (SMD = 0.75, 95% CI: 0.23, 1.28, p < 0.001). In addition, literature also suggested that text messaging also resulted in improvements in other health outcomes in university students, including decreased alcohol consumption³³³, mental problems^{334,335}, and nutritional behaviors.³³⁶ Compared with previous evidence, the current systematic review and meta-analysis focused specifically on text messaging and university students, narrowing the scope of interventions and the target population. While significant differences were not observed in some outcomes, the potential PA benefits of text message interventions should continue to be studied, such as potential reduction in BMI.

Because of the flexibility of text message interventions²²⁹, they can be implemented in a variety of PA program structures, including those designed for university students. The data suggested that university students possess a high level of electronic device ownership and enthusiasm for new technologies, making them a suitable population for the widespread implementation of text message interventions on campus.²²⁹ However, the results of previous reviews also identified several gaps in the existing evidence, such as limited information about

desired content of text messages, limited long-term evidence, and risk of bias of the studies.^{144,332,335} There are eleven studies^{133,137,140,269,270,321-323,327,328} in this systematic review that provided the content of texts used in their interventions, but the majority of them did not offer detailed information on how these text messages were developed. Informed by the COM-B model, Castro and colleagues developed a set of 24 text messages that encompassed four key domains: nudge messages, health-related messages, psychological well-being messages, and productivity messages.¹³³ Reese and colleagues developed the text messages through an iterative process involving participant input, ranking, group discussion, editing, and refinement based on feedback obtained from the nominal group technique sessions and subsequent focus groups.³²⁵ Out of the twenty studies included in this review, only two had a duration of over twelve months (one lasting eighteen months 324 and one lasting 24 months 319). Most studies lasted no more than eight weeks, which made it difficult to assess the engagement and retention of the interventions. Research demonstrates that the duration of the program may affect the program engagement and retention, as shorter interventions and higher level of completion were associated with better outcomes.337,338

5.1.2 Quality and Bias Assessment

The quality of included studies was assessed, indicating potential sources of bias, including lack of blinding and adherence in RCTs, and confounding factors in quantitative nonrandomized studies. These limitations in study design were also reported in previous reviews, highlighting the need for future research with well-controlled designs to address and improve the research quality.^{144,229}

Additionally, the method of measuring PA was not consistent across the included studies, limiting the comparability of some outcomes and the number of studies in meta-analysis. For

some outcomes, such as sitting time, walking time, standing time, and BMI, the measurements were standardized using the same units (minutes or hours per day; kg/m²). However, it is more complex to measure other variables, such as physical activity time. For example, the International Physical Activity Questionnaire (IPAQ)¹⁰⁰ were used in some RCTs to calculate the total MET, including LPA, MPA, and VPA. In contrast, Cotton et al.'s study measured LPA using a sedentary and light-intensity physical activity questionnaire^{269,339}, and MPA was measured using the seven-day physical activity recall questionnaire.³⁴⁰ IPAQ was also used in one quasi-experimental study, but the authors reported the time of LPA, MPA, and VPA instead of calculating the total MET. Keahey and colleagues used the Physical Activity and Sedentary Behavior Questionnaire³⁴¹ to measure the MVPA, which cannot be compared with MPA or VPA directly.²⁷⁰ Furthermore, some RCTs examined the effect of text message interventions using statistical models, without providing the numerical data on actual PA levels. For example, Figueroa and colleagues used the generalized estimating equations (GEE) models to evaluate the effect of several covariates (e.g., feedback and motivational messages) on steps.³²⁹ They estimated that a motivational message increased 717 steps after controlling interactions (p = .083, δ = .144).³²⁹ IPAQ was also used in this study but neither the PA minutes nor MET were reported.

Different methods of measurement, particularly for outcomes which were measured subjectively, may introduce bias. For instance, while the IPAQ demonstrated acceptable validity assessing PA in older adolescents³⁴² and healthy adults³⁴³, it may not be an accurate measure for PA or sedentary behavior in other populations. For example, Cleland and colleagues found that additional clarification is needed to ensure a more accurate and nuanced assessment of physical activity levels among older adults.³⁴⁴ Similarly, the modified version of the IPAQ and the IPAQ

Short Form were found to be inadequate for measuring total physical activity in young teens.^{342,345} Related, the 9-item IPAQ - Short Form (IPAQ-SF) was found to "overestimate" PA by an average of 84% compared to objective criteria.³⁴⁶

5.1.3 Implications and Future Directions

Text message interventions showed promising results in increasing PA among university students, and the success of these interventions is dependent on the contents and timing of text messages which should be tailored to the needs and preferences of participants. To address current limitations and challenges of text message interventions among university students, future studies should investigate the effects of longer-duration text message interventions with larger sample sizes and well-controlled designs. Also, characteristics of effective text messages among university students should be examined, including the interactional content, frequency, timing, and duration. Furthermore, studies could consider the potential benefits of combining text message interventions with other strategies, such as wearable fitness trackers, one-on-one consulting sessions, and group-based interventions. Table 5.1 below summarizes effective strategies for text messaging university students for physical activity purposes.

Personalization	Tailor and personalize messages to individual students, addressing		
	their specific needs, preferences, and goals.		
	Customize frequency and time to receive texts.		
Goal Setting	Help students set realistic and achievable PA goals, providing		
	guidance and support in developing exercise plans.		
Reminders	Timely reminders about scheduled PA and tracking progress		
Motivation	Share fitness tips and ideas to keep students engaged and prevent		
	boredom.		
	Introduce new PA opportunities or challenges		
Two-Way	Provide positive feedback and words of encouragement to boost		
Communication	students' confidence		
	Students can reply to messages, ask questions, seek guidance, or share		
	their experiences		

Table 5.1. Effective Strategies for Text Messaging University Students for PA Purposes

5.1.4 Strengths

This review is the first systematic review and meta-analysis to assess the effectiveness of text message interventions among university students in promoting physical activity, including both quantitative, qualitative, and mixed-method studies. It provides valuable insights and updated evidence on text message interventions for PA among university students, supporting the potential of text message as a tool for promoting PA. Considering the prevalence of physical inactivity in this population, the findings of the study strengthen the understanding of how text message programs can be implemented and utilized in academic settings. The review also informs the development of evidence-based interventions to improve PA in university students. Additionally, the study highlights the importance of standardized and objective measurements for PA to control the study bias. Overall, the study contributes to the growing body of evidence supporting the use of text message interventions as a feasible and effective strategy for promoting physical activity among university students.

5.1.5 Limitations

This review also has several limitations to consider. First, similar to all systematic reviews, our search might have missed relevant studies in other languages, published after our data extraction date, or not indexed in the search engines or databases. Second, despite narrowing the search to target PA interventions with text messages for university students, the heterogeneity of study designs, qualities, outcome measurements, and intervention length among the included articles may have impacted the synthesis of findings. Third, the number of studies that were eligible for meta-analysis was limited, making it difficult to estimate between-study heterogeneity and potentially leading to biased estimates and narrow confidence intervals.^{347,348}

Meanwhile, publication bias cannot be assessed for non-randomized studies or in meta-analyses of less than ten included studies because of confounders and the lack of power.³⁴⁹

5.2 Aim 2

The purpose of this quasi-experimental study was to assess and compare the effectiveness of a competition-based intervention and text message interventions among university students and evaluate participants' acceptability and experiences. Our findings suggest that both competition-based and text message interventions were effective and acceptable in increasing physical activity among university students, and Self-Efficacy and Incentive Motivation were important factors in promoting engagement in physical activity.

5.2.1 Effects, Experience, and Satisfaction with the DFFC

During weeks 0-8 of the DFFC, there were statistically significantly higher weekly MVPA minutes in the DFFC Group and DFFC + Text Group compared to the Control Group. These findings suggest an association between participation in the DFFC – either alone or in combination with text messages – and increased MVPA during the early and middle stages of the study. However, among the DFFC Group, no significant differences were observed in MVPA for weeks 1-8 compared to the first week. The DFFC Group and DFFC + Text Group showed increases in scores on the post-DFFC survey for Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation compared to the baseline. However, only the differences in Incentive Motivation scores between the Control Group and DFFC Group, as well as between the Control Group and DFFC + Text Group, were statistically significant.

The overall trend of MVPA minutes during this year's DFFC is comparable to the previous three years (2019-2021). The initial increases in MVPA in the DFFC Group and DFFC + Text Group during the first three weeks of DFFC likely reflect participants' motivation and

enthusiasm for the competition. According to participant feedback, the key strategies employed by the DFFC to motivate participants to exercise centered on the competitive nature of the fitness competition, followed by teamwork and tracking physical activity. These findings are consistent with existing literature, which suggests that competitive environments could encourage individuals to set goals and promote PA among competitive people.^{33,34,153,155,156}

However, the MVPA in the DFFC Group decreased beginning in week-4. Potential explanations for the observed decrease could relate to discouragement and self-doubt. Specifically, falling behind other teams in the first few weeks of DFFC may have a negative impact on participants' motivation to engage in the competition. Other potential explanations include concerns about data authenticity submitted by other participants, technical difficulties with the ChallengeRunner app, and competing demands for participants' time as the academic semester neared its conclusion. Previous studies found that competition could discourage individuals who did not meet their goals, feel overwhelmed, or are unable to win.^{159,161-163}

The maximum reported MVPA was in week-8 in both the DFFC Group and DFFC + Text Group, followed by a sharp drop. During the follow-up weeks (weeks 9-12), there was no statistically significant difference between all three groups in weekly MVPA. Compared to the week-1, both the DFFC Group and DFFC + Text Group experienced a statistically significant decrease in weekly MVPA in weeks 9-12. This could possibly be attributed to the assessment week for Doctor of Pharmacy students (week 9), Thanksgiving break (week 10), and final weeks for graduate students (week 11-12). According to the Fogg Behavior Model, three main components that explain human behavior: Motivation, Ability, and a Prompt. Motivation refers to an individual's willingness to perform a behavior; ability represents the individual's capability to behave; and a prompt is a cue that reminds a person to engage in a particular behavior.³⁵⁰

During weeks 9-12, participants' physical activity levels decreased compared to weeks 1-8. In the context of the Fogg Behavior Model, participants' motivation may have decreased because the competition concluded, leading to their reduced engagement in physical activity. Second, participants may not have been able to exercise due to semester exams, travel, and holiday breaks. Also, the lack of a prompt (email and text) after the intervention may have deprived them of a cue or reminder to engage in physical activity. Our apparent challenge of participant retention during the last few weeks of the semester was consistent with findings in previous research. A literature review on website-delivered physical activity interventions revealed that sample attrition rates varied from 7% to 69% across 15 studies, with an average attrition rate of 27%.³⁵¹ A 12-week study among 117 first-year students tested monetary incentives to boost fitness-center use and found that weekly incentives increased goal achievement.³⁵² However, attendance declined noticeably during weeks 10 to 12 in the presence of incentivs.³⁵² Compliance remained high in a 50-day online social networking physical activity intervention involving 51 adults, but engagement steadily declined throughout the duration.³⁵³ It is noteworthy that some participants in DFFC Group and DFFC + Text Group recommended extending the duration of the competition, while others suggested shortening it to maintain motivation. Additionally, it is important to consider the potential bias introduced by missing data due to sample attrition, given that the follow-up rate was about 70% in the DFFC Group and DFFC + Text Group and less than 30% in the Control Group.

Overall, satisfaction with the DFFC was high, and most participants indicated they were extremely likely or somewhat likely to participate in the 2023 DFFC. The popularity of the DFFC demonstrates the fitness tracking and competition programs in motivating participants and helping them maintain an active and healthy lifestyle. Likewise, Chung's study on tracking tools

in workplace wellness programs revealed that 66% of participants found these programs useful in supporting their health goals, and 78% of participants expressed their willingness to recommend health tracking programs to their colleagues.³⁵⁴

5.2.2 Effects, Experience, and Satisfaction with Text Message

Starting from week 4 of the DFFC, participants in the DFFC + Text Group demonstrated a statistically significant increase in weekly MVPA minutes compared to week 1. During the competition and follow-up weeks, there was a significant interaction between time and text message intervention on mean MVPA, indicating the impact of the text message interventions on MVPA varied across different time points. However, the one-way ANOVA analysis did not reveal a statistically significant difference in MVPA minutes between the DFFC Group and DFFC + Text Group during the competition and follow-up period. Additionally, there was no statistically significant difference in Mental Well-being, Incentive Motivation, Social Support, Self-Efficacy, and Self-Regulation between the DFFC Group and DFFC + Text Group in mean scores on the post-DFFC survey.

The MVPA minutes in DFFC + Text Group decreased in week 9 when text messages were no longer sent to participants. As shown in previous research, text message interventions could be an effective alternative in dealing with current challenges and barriers faced by university students when seeking assistance with their physical activity.^{36,170,171,179-183} Nevertheless, we also found that the text message interventions did not lead to significant increase MVPA or sustained maintenance in PA during the intervention and follow-up weeks. In addition to previously mentioned explanations such as holidays, end of semester deadlines, and exams, it could result from the relatively short duration of our text message interventions (7 weeks) and the potential inherent limitations of this approach. A systematic review of PA text

message interventions indicated that adults may benefit from the "newness" of using text messages to affect PA.¹⁷⁹ Compared to using texts alone, text message interventions featuring multiple components, such as educational sessions, phone/video calls, email messages, and selfmonitoring equipment, resulted in larger but non-significant effect sizes.²²⁹ The effectiveness of text messages also varies across different outcome measures. In the Aim 1 meta-analysis, we found evidence for a positive effect of text message interventions in MET, but not for other outcomes. However, our quasi-experimental study did not find evidence supporting the effectiveness of the text message intervention significantly increasing MVPA. One potential reason could be our limited sample size and short duration, which may not have been large enough or long enough to detect a significant effect. Also, the present study included a competition + text group instead of the text-only group. It is possible that the impact of the text message intervention could have been overshadowed by the influence of the fitness competition. Furthermore, in Aim 1, text messages were sent alongside other fitness programs in some studies, making it uncertain whether the observed effects were solely attributable to the text messages or influenced by other concurrent interventions. Additionally, our quasi-experimental study only recruited health professional students, whereas the systematic review included all university students, which limits the generalizability of our findings to the broader university student population. A recent meta-analysis also reported that text message interventions resulted in a significant increase in daily steps (d=0.38, n=10 studies) but a non-significant increase in MVPA (d=0.31, n=5 studies).²²⁹

In DFFC + Text Group, 80% of participants reported that they were satisfied with the understandability and motivational wording of the messages they received, and more than half found the texts helped motivate them to *start exercising*. While developing these messages, we

followed established guidelines from prior investigations, such as using positive framing,^{180,181} incorporating humor and emoticons, and avoiding delivery during early morning or late night hours.^{132,270,276,325} However, it is noteworthy that less than 40% of participants in the DFFC + Text Group found the messages helpful in motivating them to engage in more PA or achieve PA goals. One possible explanation for this negative experience could be a lack of personalization and contextualization in text message content.³²⁹ For example, Figueroa et al. interviewed participants in a PA intervention using a mobile application and found that some texts, such as "you can do better!", were annoying and unhelpful without any introduction to "do better." In our study, we also noted some similar contexts, like "You've got this!", which may not be very useful for university students to promote PA.³²⁹ Also, in our developed messages, we prioritized and emphasized the concept of "starting to exercise" over engaging in more exercise or achieving PA goals. However, it is important to acknowledge that participants in the DFFC may possess a high level of self-motivation for PA and seek guidance on how to exercise more and attain their PA goals. In addition, some messages we sent included more than two sentences, which was not recommended in the previous research. Overall, the satisfaction of text messages was acceptable.325

5.2.3 Roles of Incentive Motivation and Self-Efficacy

The results of this study revealed that both Self-Efficacy and Inventive Motivation exerted a moderate but significant impact on PA among university students, and these two factors mediated the effects of interventions on PA. In the context of physical activity, Self-Efficacy is the belief that one can successfully engage in adequate exercise and maintain an active lifestyle, and individuals with high PA Self-Efficacy are more likely to establish PA goals and exhibit perseverance in the face of obstacles.^{221,228,355} Research demonstrates that Self-

Efficacy is a strong predictor of PA in university students.^{124,223,228,356} A literature review published in Lancet in 2012 highlighted that both health status and self-efficacy show the strongest associations with physical activity among adults.²⁰⁶ In the mediation analysis of Aim 2, we found that the effect of interventions was mediated by Self-Efficacy, which is consistent with previous studies.^{355,357-359} Additionally, Self-Efficacy and Social Support were found to have a large and significant effect on Self-Regulation in the path analysis of Aim 2. This is consistent with previous research indicating that the impact of Self-Efficacy on physical activity was mediated by Self-Regulation.²²³

Incentive Motivation, which was operationalized as the use of rewards to modify PA behavior, had a medium and significant effect on Self-Efficacy and PA in this study. This finding is consistent with previous studies that demonstrated the effectiveness of incentives in improving PA behavior.^{352,360-363} A systematic review and meta-analysis found that incentives could also improve the attendance at PA interventions.³⁶¹ According to research by RAND Europe, participants in the Vitality Active Rewards program with Apple Watch reported an additional 4.8 activity days per month compared to non-participants.³⁶⁴ Similarly, the results of the ACHIEVE project demonstrated that an incentives-based program was a feasible method to improve participant retention leading to a significant increase on PA (p < 0.001).³⁶⁵ In 2015, Barte and Wendel-Vos conducted a systematic review of twelve PA interventions with financial incentives and found that compared to unconditional incentives (e.g., public facilities), conditional incentives (e.g., rewards based on achieving specific goals) had favorable impacts on PA.³⁶²

Previous research found that offering incentives - such as financial rewards or social recognition - can provide students with a tangible and immediate reward for their efforts.^{361,366} The nature of this type of incentive helps offset the short-term costs of exercise, such as time and effort. In the present study, we found that Incentive Motivation significantly affected Self-Efficacy, suggesting a stronger sense of Self-Efficacy when students experience positive outcomes and rewards from their exercise efforts. However, in this study, we did not find adequate evidence that supported the long-term effects of incentives.

5.2.4 Implications and Future Directions

Previous research underscored the significance of prioritizing the well-being of healthcare professionals, emphasizing the need to implement interventions that promote healthy habits among healthcare workers, including health professional students.³⁶⁷⁻³⁶⁹ However, it is essential to recognize that the effectiveness of wellness interventions is not uniform across populations or settings, and the implementation of wellness interventions should go beyond a "one-size-fits-all approach".²²⁹ Further research is needed to investigate the generalizability of the findings and explore the long-term effects of these interventions with more robust study designs. For future studies, we suggest exploring multimodal interventions grounded in behavior theories to target different aspects of behavior change. Moreover, there is a need to develop evidence-based guidelines for promoting the wellness of health professional students as well as other university students.

5.2.5 Strengths

This study is the first experimental study to quantitatively evaluate the DFFC competition and PA-related text messages among students in three health professional programs, providing evidence for the feasibility and effectiveness of competition-based interventions, text message interventions, and combined interventions of these two approaches among university students. Additionally, we collected quantitative and qualitative perceptions of participants and proposed future directions for the improvement of both interventions. Moreover, this study included a

follow-up period to assess the maintenance of behavior change and monitor participants' PA after the interventions. Furthermore, our findings revealed the relationships between Self-regulation, Incentive Motivation, Self-efficacy, Social Support, and Physical Activity, which is of great importance given that the mediation effect of Incentive Motivation on PA has never been investigated.

Overall, this quasi-experimental study provided first-hand evidence to support the implementation of fitness competition and text message interventions as feasible, acceptable, and effective strategies for promoting healthy lifestyles among university students, especially for health professional students.

5.2.6 Limitations

The primary limitation of this study was the sample population. First, due to limited sample size, we were unable to develop some hypothesized models, such as path analysis including both interventions, as well as a complicated mediation model including all dependent variables. Second, our samples were enrolled in health professional programs and may not generalize to all university students. Given the demands and high-pressure of their work, health professional students are known to experience higher levels of stress and anxiety compared to university students in other fields, which may introduce external confounders in assessing bias.³⁷⁰⁻³⁷² For example, we cannot consistently control their academic pressure and workload throughout the interventions and follow-up. Third, while participants were randomized to receive text messages or not, they were not randomized to participate in the DFFC or to a control group with no intervention, resulting in unbalanced baseline characteristics such as age and race. Due to self-selection for participation, students in the DFFC and DFFC + Text Group may exhibit a higher enthusiasm for exercise than students in the Control Group. One possible indicator

supporting this is the lower follow-up rate in the Control Group compared to the two intervention groups (30% vs. 70%).

Another important limitation is measurement bias. For the primary outcome, the accuracy of MVPA minutes could not be guaranteed. While PA minutes were collected through fitness trackers and the ChallengeRunner app, participants had the ability to manually enter data. The veracity of unusually high MVPA data of several teams was questioned. While these teams were not enrolled in this study, we were unable to validate any PA data for DFFC participants. Several participants reported technical issues connecting their trackers/phones with the ChallengeRunner app and had to report data manually, which could impact the intervention retention and the objectivity of the measurement. For the secondary outcomes, self-report measures were used for Self-regulation, Incentive Motivation, Self-efficacy, Social Support, and Mental Well-being, which is subject to recall bias and social desirability bias.³⁷³ Also, a limitation of our survey instruments is the lack of attention-tracking questions which help gauge respondents' attentiveness and ensure data accuracy.

Additionally, due to the setting of the DFFC competition, the Control Group in this study was not fully "controlled". During the competition, all students, faculty, and staff members in these three programs received the weekly email, including participants in the Control Group. Additionally, the group fitness classes were open to the Auburn community, meaning that students in the Control Group also had opportunities to participate. Because all participants in this study used the ChallengeRunner app to report their data, participants in the Control Group were using some competitive functions of this platform, such as the leader board. Also, the group activities during the competition were open to HCOP students and employees, meaning that participants from AUCON and VCOM-Auburn did not have organized group fitness classes.

Lastly, for participants in the DFFC + Text Group, although we had delivery reports for text message delivery, we were not able to confirm messages were read or any actions taken after reading the messages. In this study, the incorporation of text message interventions into a competition setting makes it challenging to isolate the specific effects of text message interventions. To accurately assess the effectiveness of text message interventions, it is crucial to introduce an additional intervention group, the Text Group, which would solely receive the text messages without the competition aspect, and compare outcomes of the Text Group to DFFC Group as well as the Control Group,

5.3 Implications and Conclusions

A comprehensive study was conducted to understand various components in promoting physical activity among university students, including a systematic review, meta-analyses, a fitness competition (DFFC) group, a combination of DFFC and text message interventions (DFFC + Text), and qualitative data analysis. Overall, both fitness competitions and text message interventions have been identified as valuable strategies for promoting engagement in physical activity among university students.

Updated evidence from the literature supports the positive effect of text message interventions on metabolic equivalents of tasks (MET) among university students but not for other outcomes. Effective text messages should be tailored to the specific needs and preferences of the target audience.

In the quasi-experimental study, both competition-based and text message interventions were found to be feasible and acceptable in promoting physical activity among university students. The fitness competition highlights the role of competitive environments and teamwork in motivating students to participate in physical activity. The quantitative analyses reveal that Self-Efficacy and Incentive Motivation are significant factors in influencing Physical Activity and mediating the effects of Physical Activity interventions. Future work needs to develop multimodal interventions based on behavior theories and explore long-term effects and generalizability with robust study designs. It is essential to evaluate the need of target populations for personalized interventions, and incorporate social support through teamwork, peer support, and group activities to enhance interpersonal interaction and increase physical activity engagement. Also, more resources are needed to enhance self-efficacy for physical activity, such as education sessions, one-on-one consulting, workouts, and more training opportunities. Moreover, to retain participation and maintain motivation throughout and after the competition, introducing new incentives, events, or challenges regularly could help provide ongoing support for participants and prevent plateauing in engagement. For example, implementing multiple reward tiers could increase physical activity motivation for both high achievers and beginners, providing everyone with a chance to win some award for their efforts.

References

1. Morris JN. Exercise in the prevention of coronary heart disease: today's best buy in public health. *Med Sci Sports Exerc*. 1994/07// 1994;26(7):807-814.

2. Center for Chronic Disease Prevention and Health Promotion. Why It Matters. Updated May 13. June 20. <u>https://www.cdc.gov/physicalactivity/about-physical-activity/why-it-matters.html</u>

3. Irwin ML, McTiernan A, Bernstein L, et al. Physical activity levels among breast cancer survivors. *Medicine and science in sports and exercise*. 2004;36(9):1484.

4. Rock CL, Doyle C, Demark-Wahnefried W, et al. Nutrition and physical activity guidelines for cancer survivors. <u>https://doi.org/10.3322/caac.21142</u>. *CA: A Cancer Journal for Clinicians*. 2012/07/01 2012;62(4):242-274. doi:<u>https://doi.org/10.3322/caac.21142</u>

5. Fong DY, Ho JW, Hui BP, et al. Physical activity for cancer survivors: meta-analysis of randomised controlled trials. *Bmj*. 2012;344:e70.

6. Speck RM, Courneya KS, Mâsse LC, Duval S, Schmitz KH. An update of controlled physical activity trials in cancer survivors: a systematic review and meta-analysis. *J Cancer Surviv*. Jun 2010;4(2):87-100. doi:10.1007/s11764-009-0110-5

7. Gill JMR, Cooper AR. Physical Activity and Prevention of Type 2 Diabetes Mellitus. *Sports Medicine*. 2008/10/01 2008;38(10):807-824. doi:10.2165/00007256-200838100-00002

8. Cornelissen VA, Smart NA. Exercise training for blood pressure: a systematic review and metaanalysis. *Journal of the American Heart Association*. 2013;2(1):e004473-e004473. doi:10.1161/JAHA.112.004473

9. Kohl HW, Craig CL, Lambert EV, et al. The pandemic of physical inactivity: global action for public health. *The Lancet*. 2012/07/21/ 2012;380(9838):294-305. doi:<u>https://doi.org/10.1016/S0140-6736(12)60898-8</u>

10. Wen CP, Wai JPM, Tsai MK, et al. Minimum amount of physical activity for reduced mortality and extended life expectancy: a prospective cohort study. *The Lancet*. 2011;378(9798):1244-1253. doi:10.1016/S0140-6736(11)60749-6

11. Mandolesi L, Polverino A, Montuori S, et al. Effects of Physical Exercise on Cognitive Functioning and Wellbeing: Biological and Psychological Benefits. *Frontiers in psychology*. 2018;9:509-509. doi:10.3389/fpsyg.2018.00509

12. Pucci GC, Rech CR, Fermino RC, Reis RS. Association between physical activity and quality of life in adults. *Rev Saude Publica*. Feb 2012;46(1):166-79. doi:10.1590/s0034-89102012000100021

13. White RL, Babic MJ, Parker PD, Lubans DR, Astell-Burt T, Lonsdale C. Domain-specific physical activity and mental health: a meta-analysis. *American journal of preventive medicine*. 2017;52(5):653-666.

14. Rebar AL, Taylor A. Physical activity and mental health; it is more than just a prescription. *Mental Health and Physical Activity*. 2017;13:77-82.

15. Center for Chronic Disease Prevention and Health Promotion. National Center for Chronic Disease Prevention and Health Promotion, Division of Nutrition, Physical Activity, and Obesity.Data, Trend and Maps [online]. Accessed May 08, 2022. <u>https://www.cdc.gov/nccdphp/dnpao/data-trends-maps/index.html</u>.

16. Erickson KI, Hillman C, Stillman CM, et al. Physical Activity, Cognition, and Brain Outcomes: A Review of the 2018 Physical Activity Guidelines. *Medicine and science in sports and exercise*. 2019;51(6):1242-1251. doi:10.1249/MSS.00000000001936

17. Centers for Disease Control and Prevention. CDC Maps America's High Levels of Inactivity. Updated January 16, 2020. Accessed May 16, 2022, <u>https://www.cdc.gov/media/releases/2020/0116-americas-</u>

inactivity.html#:~:text=Physical%20inactivity%20for%20adults%20is%20defined%20as%20not%20partici

pating%20in%20any%20leisure%2Dtime%20physical%20activities%20over%20the%20last%20month%2 0%E2%80%93%20activities%20such%20as%20running%2C%20walking%20for%20exercise%2C%20or%2 0gardening.

18. Santos SJ, Hurtado-Ortiz MT, Armendariz M, vanTwist V, Castillo Y. Obesity-Related Dietary Patterns and Health Status of Diabetes among At-Risk Latino College Students. *J Hispanic High Educ*. 2017;16(4):291-313. doi:10.1177/1538192716653504

19. McArthur LH, Riggs A, Uribe F, Spaulding TJ. Health Belief Model Offers Opportunities for Designing Weight Management Interventions for College Students. *Journal of Nutrition Education and Behavior*. 2018/05/01/ 2018;50(5):485-493. doi:<u>https://doi.org/10.1016/j.jneb.2017.09.010</u>

20. López-Valenciano A, Suárez-Iglesias D, Sanchez-Lastra MA, Ayán C. Impact of COVID-19 Pandemic on University Students' Physical Activity Levels: An Early Systematic Review. *Frontiers in psychology*. 2021;11:624567-624567. doi:10.3389/fpsyg.2020.624567

21. Huang TTK, Harris KJ, Lee RE, Nazir N, Born W, Kaur H. Assessing Overweight, Obesity, Diet, and Physical Activity in College Students. *Journal of American College Health*. 2003/09/01 2003;52(2):83-86. doi:10.1080/07448480309595728

22. Keating XD, Guan J, Piñero JC, Bridges DM. A meta-analysis of college students' physical activity behaviors. *J Am Coll Health*. Sep-Oct 2005;54(2):116-25. doi:10.3200/jach.54.2.116-126

23. Haase A, Steptoe A, Sallis JF, Wardle J. Leisure-time physical activity in university students from 23 countries: associations with health beliefs, risk awareness, and national economic development. *Prev Med*. 2004/07/01/ 2004;39(1):182-190. doi:<u>https://doi.org/10.1016/j.ypmed.2004.01.028</u>

24. Irwin JD. Prevalence of University Students' Sufficient Physical Activity: A Systematic Review. *Perceptual and Motor Skills*. 2004/06/01 2004;98(3):927-943. doi:10.2466/pms.98.3.927-943

25. Kljajević V, Stanković M, Đorđević D, et al. Physical Activity and Physical Fitness among University Students— A Systematic Review. *International Journal of Environmental Research and Public Health*. 2022;19(1):158.

26. Plotnikoff RC, Costigan SA, Williams RL, et al. Effectiveness of interventions targeting physical activity, nutrition and healthy weight for university and college students: a systematic review and metaanalysis. *International Journal of Behavioral Nutrition and Physical Activity*. 2015/04/01 2015;12(1):45. doi:10.1186/s12966-015-0203-7

27. Rose T, Barker M, Maria Jacob C, et al. A Systematic Review of Digital Interventions for Improving the Diet and Physical Activity Behaviors of Adolescents. *Journal of Adolescent Health*. 2017/12/01/ 2017;61(6):669-677. doi:<u>https://doi.org/10.1016/j.jadohealth.2017.05.024</u>

28. Atkin AJ, Gorely T, Biddle SJH, Cavill N, Foster C. Interventions to Promote Physical Activity in Young People Conducted in the Hours Immediately After School: A Systematic Review. *International Journal of Behavioral Medicine*. 2011/09/01 2011;18(3):176-187. doi:10.1007/s12529-010-9111-z

29. Kahn EB, Ramsey LT, Brownson RC, et al. The effectiveness of interventions to increase physical activity. A systematic review. *Am J Prev Med*. May 2002;22(4 Suppl):73-107. doi:10.1016/s0749-3797(02)00434-8

30. Sohaib Aslam A, van Luenen S, Aslam S, van Bodegom D, Chavannes NH. A systematic review on the use of mHealth to increase physical activity in older people. *Clinical eHealth*. 2020/01/01/2020;3:31-39. doi:https://doi.org/10.1016/j.ceh.2020.04.002

31. Yun L, Ori EM, Lee Y, Sivak A, Berry TR. A Systematic Review of Community-wide Media Physical Activity Campaigns: An Update From 2010. *J Phys Act Health*. Jul 2017;14(7):552-570. doi:10.1123/jpah.2016-0616

32. Triplett N. The Dynamogenic Factors in Pacemaking and Competition. *The American Journal of Psychology*. 1898;9(4):507-533. doi:10.2307/1412188

33. Frederick-Recascino CM, Schuster-Smith H. Competition and intrinsic motivation in physical activity: A comparison of two groups. *Journal of sport behaviour*. 2003;26(3):240-254.

34. Charles A, Girard A, Buckinx F, Mouton A, Reginster J-Y, Bruyère O. Senior physical activity contests in nursing homes: a feasibility study. *Aging Clinical and Experimental Research*. 2020/05/01 2020;32(5):869-876. doi:10.1007/s40520-020-01529-9

35. Marquez DX, Aguiñaga S, Vásquez PM, et al. A systematic review of physical activity and quality of life and well-being. *Translational Behavioral Medicine*. 2020;10(5):1098-1109. doi:10.1093/tbm/ibz198

36. Hall AK, Cole-Lewis H, Bernhardt JM. Mobile text messaging for health: a systematic review of reviews. *Annu Rev Public Health*. 2015;36:393-415. doi:10.1146/annurev-publhealth-031914-122855

37. Suffoletto B. Text Message Behavioral Interventions: From Here to Where? *Curr Opin Psychol*. Feb 1 2016;9:16-21. doi:10.1016/j.copsyc.2015.09.012

38. Stephens J, Allen J. Mobile phone interventions to increase physical activity and reduce weight: a systematic review. *J Cardiovasc Nurs*. Jul-Aug 2013;28(4):320-329.

doi:10.1097/JCN.0b013e318250a3e7

39. Siopis G, Chey T, Allman-Farinelli M. A systematic review and meta-analysis of interventions for weight management using text messaging. <u>https://doi.org/10.1111/jhn.12207</u>. *Journal of Human Nutrition and Dietetics*. 2015/02/01 2015;28(s2):1-15. doi:<u>https://doi.org/10.1111/jhn.12207</u>

40. Bock C, Jarczok MN, Litaker D. Community-based efforts to promote physical activity: a systematic review of interventions considering mode of delivery, study quality and population subgroups. *J Sci Med Sport*. May 2014;17(3):276-82. doi:10.1016/j.jsams.2013.04.009

41. Heath GW, Parra DC, Sarmiento OL, et al. Evidence-based intervention in physical activity: lessons from around the world. *Lancet*. Jul 21 2012;380(9838):272-81. doi:10.1016/s0140-6736(12)60816-2

42. van Sluijs EMF, Kriemler S, McMinn AM. The effect of community and family interventions on young people's physical activity levels: a review of reviews and updated systematic review. *British Journal of Sports Medicine*. 2011;45(11):914. doi:10.1136/bjsports-2011-090187

43. Favieri F, French MN, Casagrande M, Chen EY. Physical activity interventions have a moderate effect in increasing physical activity in university students—a meta-analysis. *Journal of American College Health*. 2022:1-12. doi:10.1080/07448481.2021.1998070

44. Deliens T, Deforche B, De Bourdeaudhuij I, Clarys P. Determinants of physical activity and sedentary behaviour in university students: a qualitative study using focus group discussions. *BMC Public Health*. 2015/02/28 2015;15(1):201. doi:10.1186/s12889-015-1553-4

45. Small M, Bailey-Davis L, Morgan N, Maggs J. Changes in eating and physical activity behaviors across seven semesters of college: living on or off campus matters. *Health Educ Behav*. 2013;40(4):435-441. doi:10.1177/1090198112467801

46. Hong QN, Pluye P, Fàbregues S, et al. Improving the content validity of the mixed methods appraisal tool: a modified e-Delphi study. *Journal of Clinical Epidemiology*. 2019;111:49-59.e1. doi:10.1016/j.jclinepi.2019.03.008

47. Hong QN, Fàbregues S, Bartlett G, et al. The Mixed Methods Appraisal Tool (MMAT) version 2018 for information professionals and researchers. *Education for Information*. 2018;34:285-291. doi:10.3233/EFI-180221

48. Sterne JAC, Sutton AJ, Ioannidis JPA, et al. Recommendations for examining and interpreting funnel plot asymmetry in meta-analyses of randomised controlled trials. *BMJ*. 2011;343:d4002. doi:10.1136/bmj.d4002

49. Valentine JC, Pigott TD, Rothstein HR. How Many Studies Do You Need?: A Primer on Statistical Power for Meta-Analysis. *Journal of Educational and Behavioral Statistics*. 2010/04/01 2010;35(2):215-247. doi:10.3102/1076998609346961

50. Ryan R, Hill S. Cochrane consumers and communication group: Meta-analysis. *Cochrane Consumers and Communication Review Group Available online:* <u>https://cccrq</u> cochrane org/sites/cccrg

cochrane org/files/public/uploads/meta-analysis_revised_ december_1st_1_2016 pdf (accessed on 15 June 2021). 2016;

51. Westerterp KR. Physical activity and physical activity induced energy expenditure in humans: measurement, determinants, and effects. *Frontiers in physiology*. 2013;4:90.

52. Ball JW, Bice MR, Parry T. Adults' motivation for physical activity: Differentiating motives for exercise, sport, and recreation. *Recreational Sports Journal*. 2014;38(2):130-142.

53. Caspersen CJ, Powell KE, Christenson GM. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. *Public Health Rep*. Mar-Apr 1985;100(2):126-131.

54. Cowell LL, Squires WG, Raven PB. Benefits of aerobic exercise for the paraplegic: a brief review. *Medicine and science in sports and exercise*. 1986/10// 1986;18(5):501-508.

55. Epstein LH, Wing RR. Aerobic exercise and weight. *Addictive Behaviors*. 1980/01/01/ 1980;5(4):371-388. doi:<u>https://doi.org/10.1016/0306-4603(80)90011-8</u>

56. Cassilhas RC, Tufik S, de Mello MT. Physical exercise, neuroplasticity, spatial learning and memory. *Cell Mol Life Sci*. Mar 2016;73(5):975-83. doi:10.1007/s00018-015-2102-0

57. Atha J. STRENGTHENING MUSCLE. *Exercise and Sport Sciences Reviews*. 1981;9(1)

58. Dehail P, Duclos C, Barat M. Electrical stimulation and muscle strengthening. *Annales de Réadaptation et de Médecine Physique*. 2008/07/01/ 2008;51(6):441-451. doi:https://doi.org/10.1016/j.annrmp.2008.05.001

59. Hootman JM. 2008 Physical Activity Guidelines for Americans: an opportunity for athletic trainers. *Journal of athletic training*. 2009;44(1):5-6.

60. Mendes MdA, Da Silva I, Ramires V, et al. Metabolic equivalent of task (METs) thresholds as an indicator of physical activity intensity. *PLoS One*. 2018;13(7):e0200701.

61. Alinia P, Saeedi R, Mortazavi B, Rokni A, Ghasemzadeh H. Impact of sensor misplacement on estimating metabolic equivalent of task with wearables. IEEE; 2015:1-6.

62. Piercy KL, Troiano RP, Ballard RM, et al. The physical activity guidelines for Americans. *Jama*. 2018;320(19):2020-2028.

63. Miyachi M, Yamamoto K, Ohkawara K, Tanaka S. METs in adults while playing active video games: a metabolic chamber study. *Med Sci Sports Exerc*. 2010;42(6):1149-53.

64. Schenkman M, Moore CG, Kohrt WM, et al. Effect of high-intensity treadmill exercise on motor symptoms in patients with de novo Parkinson disease: a phase 2 randomized clinical trial. *JAMA neurology*. 2018;75(2):219-226.

65. Pate RR, Pratt M, Blair SN, et al. Physical activity and public health. A recommendation from the Centers for Disease Control and Prevention and the American College of Sports Medicine. *Jama*. Feb 1 1995;273(5):402-7. doi:10.1001/jama.273.5.402

66. Fogli-Cawley JJ, Dwyer JT, Saltzman E, McCullough ML, Troy LM, Jacques PF. The 2005 dietary guidelines for Americans adherence index: development and application. *The Journal of nutrition*. 2006;136(11):2908-2915.

67. Office of the Surgeon G. Publications and Reports of the Surgeon General. *Step It Up! The Surgeon General's Call to Action to Promote Walking and Walkable Communities*. US Department of Health and Human Services; 2015.

68. Brooks GA, Butte NF, Rand WM, Flatt J-P, Caballero B. Chronicle of the Institute of Medicine physical activity recommendation: how a physical activity recommendation came to be among dietary recommendations. *The American Journal of Clinical Nutrition*. 2004;79(5):921S-930S. doi:10.1093/ajcn/79.5.921S

69. Haskell WL, Lee IM, Pate RR, et al. Physical activity and public health: updated recommendation for adults from the American College of Sports Medicine and the American Heart Association. *Med Sci Sports Exerc*. Aug 2007;39(8):1423-34. doi:10.1249/mss.0b013e3180616b27

70. Lee IM, Skerrett PJ. Physical activity and all-cause mortality: what is the dose-response relation? *Med Sci Sports Exerc*. Jun 2001;33(6 Suppl):S459-71; discussion S493-4. doi:10.1097/00005768-200106001-00016

71. Kraus WE, Powell KE, Haskell WL, et al. Physical Activity, All-Cause and Cardiovascular Mortality, and Cardiovascular Disease. *Medicine and science in sports and exercise*. 2019;51(6):1270-1281. doi:10.1249/MSS.00000000001939

72. 2018 Physical Activity Guidelines Advisory Committee. 2018 Physical Activity Guidelines Advisory Committee Scientific Report. 2018.

73. Zhao M, Veeranki SP, Magnussen CG, Xi B. Recommended physical activity and all cause and cause specific mortality in US adults: prospective cohort study. *BMJ*. 2020;370:m2031. doi:10.1136/bmj.m2031

74. Moore SC, Patel AV, Matthews CE, et al. Leisure Time Physical Activity of Moderate to Vigorous Intensity and Mortality: A Large Pooled Cohort Analysis. *PLOS Medicine*. 2012;9(11):e1001335. doi:10.1371/journal.pmed.1001335

75. Leitzmann MF, Park Y, Blair A, et al. Physical Activity Recommendations and Decreased Risk of Mortality. *Archives of Internal Medicine*. 2007;167(22):2453-2460. doi:10.1001/archinte.167.22.2453

76. Strain T, Brage S, Sharp SJ, et al. Use of the prevented fraction for the population to determine deaths averted by existing prevalence of physical activity: a descriptive study. *The Lancet Global Health*. 2020;8(7):e920-e930. doi:10.1016/S2214-109X(20)30211-4

77. Lieberman DE, Kistner TM, Richard D, Lee I-M, Baggish AL. The active grandparent hypothesis: Physical activity and the evolution of extended human healthspans and lifespans. *Proceedings of the National Academy of Sciences*. 2021;118(50):e2107621118. doi:doi:10.1073/pnas.2107621118

78. Pekkanen J, Nissinen A, Marti B, Tuomilehto J, Punsar S, Karvonen M. REDUCTION OF PREMATURE MORTALITY BY HIGH PHYSICAL ACTIVITY: A 20-YEAR FOLLOW-UP OF MIDDLE-AGED FINNISH MEN. *The Lancet*. 1987/06/27/ 1987;329(8548):1473-1477. doi:<u>https://doi.org/10.1016/S0140-6736(87)92218-5</u>

79. Tucker LA. Physical activity and telomere length in U.S. men and women: An NHANES investigation. *Preventive Medicine*. 2017/07/01/ 2017;100:145-151. doi:https://doi.org/10.1016/j.ypmed.2017.04.027

80. Mundstock E, Zatti H, Louzada FM, et al. Effects of physical activity in telomere length:
Systematic review and meta-analysis. *Ageing Res Rev.* Jul 2015;22:72-80. doi:10.1016/j.arr.2015.02.004
81. Lin X, Zhou J, Dong B. Effect of different levels of exercise on telomere length: A systematic

review and meta-analysis. *J Rehabil Med*. Jul 8 2019;51(7):473-478. doi:10.2340/16501977-2560 82. Reimers CD, Knapp G, Reimers AK. Does physical activity increase life expectancy? A review of the literature. *J Aging Res*. 2012;2012:243958. doi:10.1155/2012/243958

83. Wahid A, Manek N, Nichols M, et al. Quantifying the Association Between Physical Activity and Cardiovascular Disease and Diabetes: A Systematic Review and Meta-Analysis. *Journal of the American Heart Association*. 5(9):e002495. doi:10.1161/JAHA.115.002495

84. Pandey A, Garg S, Khunger M, et al. Dose–Response Relationship Between Physical Activity and Risk of Heart Failure. *Circulation*. 2015/11/10 2015;132(19):1786-1794.

doi:10.1161/CIRCULATIONAHA.115.015853

85. Cornelissen VA, Buys R, Smart NA. Endurance exercise beneficially affects ambulatory blood pressure: a systematic review and meta-analysis. *J Hypertens*. Apr 2013;31(4):639-48. doi:10.1097/HJH.0b013e32835ca964

86. Soroush A, Der Ananian C, Ainsworth BE, et al. Effects of a 6-Month Walking Study on Blood Pressure and Cardiorespiratory Fitness in U.S. and Swedish Adults: ASUKI Step Study. *Asian J Sports Med*. 2013;4(2):114-124. doi:10.5812/asjsm.34492 87. Lynch J, Helmrich SP, Lakka TA, et al. Moderately intense physical activities and high levels of cardiorespiratory fitness reduce the risk of non-insulin-dependent diabetes mellitus in middle-aged men. *Arch Intern Med*. Jun 24 1996;156(12):1307-14.

88. Hu FB, Sigal RJ, Rich-Edwards JW, et al. Walking compared with vigorous physical activity and risk of type 2 diabetes in women: a prospective study. *Jama*. Oct 20 1999;282(15):1433-9. doi:10.1001/jama.282.15.1433

89. Hu G, Lindström J, Valle TT, et al. Physical activity, body mass index, and risk of type 2 diabetes in patients with normal or impaired glucose regulation. *Arch Intern Med*. Apr 26 2004;164(8):892-6. doi:10.1001/archinte.164.8.892

90. Meyerhardt JA, Giovannucci EL, Holmes MD, et al. Physical Activity and Survival After Colorectal Cancer Diagnosis. *Journal of Clinical Oncology*. 2006;24(22):3527-3534. doi:10.1200/jco.2006.06.0855

91. Galvão DA, Taaffe DR, Spry N, Joseph D, Newton RU. Combined resistance and aerobic exercise program reverses muscle loss in men undergoing androgen suppression therapy for prostate cancer without bone metastases: a randomized controlled trial. *J Clin Oncol*. Jan 10 2010;28(2):340-7. doi:10.1200/jco.2009.23.2488

92. Meyerhardt JA, Heseltine D, Niedzwiecki D, et al. Impact of Physical Activity on Cancer Recurrence and Survival in Patients With Stage III Colon Cancer: Findings From CALGB 89803. *Journal of Clinical Oncology*. 2006;24(22):3535-3541. doi:10.1200/jco.2006.06.0863

93. Robertson MC, Lyons EJ, Song J, et al. Change in physical activity and quality of life in endometrial cancer survivors receiving a physical activity intervention. *Health and Quality of Life Outcomes*. 2019/05/27 2019;17(1):91. doi:10.1186/s12955-019-1154-5

94. O'Neill LM, Guinan E, Doyle SL, et al. The RESTORE Randomized Controlled Trial: Impact of a Multidisciplinary Rehabilitative Program on Cardiorespiratory Fitness in Esophagogastric cancer Survivorship. *Ann Surg.* Nov 2018;268(5):747-755. doi:10.1097/sla.00000000002895

95. Michaels C. The importance of exercise in lung cancer treatment. *Translational Lung Cancer Research*. 2016;5(3):235-238.

96. Ahlskog JE, Geda YE, Graff-Radford NR, Petersen RC. Physical exercise as a preventive or diseasemodifying treatment of dementia and brain aging. *Mayo Clin Proc.* 2011;86(9):876-884. doi:10.4065/mcp.2011.0252

97. Hillman CH, Belopolsky AV, Snook EM, Kramer AF, McAuley E. Physical activity and executive control: implications for increased cognitive health during older adulthood. *Research quarterly for exercise and sport*. 2004;75(2):176-185.

98. Biddle S, Mutrie N. *Psychology of physical activity: Determinants, well-being and interventions*. Routledge; 2007.

99. Anokye NK, Trueman P, Green C, Pavey TG, Taylor RS. Physical activity and health related quality of life. *BMC Public Health*. 2012/08/07 2012;12(1):624. doi:10.1186/1471-2458-12-624

100. Villalobos F, Vinuesa A, Pedret R, Basora T, Basora J, Arija V. Physical activity and health-related quality of life in adults: The "Pas a Pas" community intervention programme. *Mental Health and Physical Activity*. 2019/10/01/ 2019;17:100301. doi:<u>https://doi.org/10.1016/j.mhpa.2019.100301</u>

101. Marenus MW, Friedman K, Sanowski J, et al. Psychological predictors of physical activity intensity among college students. *Journal of American College Health*. 2022:1-10. doi:10.1080/07448481.2022.2037614

102. Hosker DK, Elkins RM, Potter MP. Promoting Mental Health and Wellness in Youth Through Physical Activity, Nutrition, and Sleep. *Child Adolesc Psychiatr Clin N Am*. Apr 2019;28(2):171-193. doi:10.1016/j.chc.2018.11.010

103. Biddle SJ, Asare M. Physical activity and mental health in children and adolescents: a review of reviews. *Br J Sports Med*. Sep 2011;45(11):886-95. doi:10.1136/bjsports-2011-090185
104. Sharma A, Madaan V, Petty FD. Exercise for mental health. *Prim Care Companion J Clin Psychiatry*. 2006;8(2):106. doi:10.4088/pcc.v08n0208a

105. Crombie KM, Sartin-Tarm A, Sellnow K, et al. Aerobic exercise and consolidation of fear extinction learning among women with posttraumatic stress disorder. *Behaviour Research and Therapy*. 2021/07/01/ 2021;142:103867. doi:<u>https://doi.org/10.1016/j.brat.2021.103867</u>

106. Chekroud SR, Gueorguieva R, Zheutlin AB, et al. Association between physical exercise and mental health in 1·2 million individuals in the USA between 2011 and 2015: a cross-sectional study. *The Lancet Psychiatry*. 2018;5(9):739-746. doi:10.1016/S2215-0366(18)30227-X

107. Benedetti TRB, Borges LJ, Petroski EL, Gonçalves LHT. Physical activity and mental health status among elderly people. *Revista de Saúde Pública*. 2008;42:302-307.

108. Sharma A, Madaan V, Petty FD. Exercise for mental health. *Prim Care Companion J Clin Psychiatry*. 2006;8(2):106-106. doi:10.4088/pcc.v08n0208a

109. Biddle S. Physical activity and mental health: evidence is growing. *World Psychiatry*. 2016;15(2):176.

110. Asztalos M, De Bourdeaudhuij I, Cardon G. The relationship between physical activity and mental health varies across activity intensity levels and dimensions of mental health among women and men. *Public health nutrition*. 2010;13(8):1207-1214.

111. Paffenbarger RS, Jr., Hyde RT, Wing AL, Lee IM, Jung DL, Kampert JB. The association of changes in physical-activity level and other lifestyle characteristics with mortality among men. *N Engl J Med*. Feb 25 1993;328(8):538-45. doi:10.1056/nejm199302253280804

112. Hu FB, Willett WC, Li T, Stampfer MJ, Colditz GA, Manson JE. Adiposity as compared with physical activity in predicting mortality among women. *N Engl J Med*. Dec 23 2004;351(26):2694-703. doi:10.1056/NEJMoa042135

113. Leon AS, Connett J, Jacobs DR, Jr., Rauramaa R. Leisure-time physical activity levels and risk of coronary heart disease and death. The Multiple Risk Factor Intervention Trial. *Jama*. Nov 6 1987;258(17):2388-95.

114. Booth FW, Roberts CK, Laye MJ. Lack of exercise is a major cause of chronic diseases. *Compr Physiol*. Apr 2012;2(2):1143-211. doi:10.1002/cphy.c110025

115. Lee IM, Shiroma EJ, Lobelo F, Puska P, Blair SN, Katzmarzyk PT. Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. *The Lancet*. 2012;380(9838):219-229. doi:10.1016/S0140-6736(12)61031-9

116. Lee IM, Shiroma EJ, Lobelo F, et al. Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. *Lancet*. 2012;380(9838):219-229. doi:10.1016/S0140-6736(12)61031-9

117. Centers for Disease Control and Prevention. Physical Inactivity. Updated June 7. Accessed June 10, 2022. <u>https://www.cdc.gov/chronicdisease/resources/publications/factsheets/physical-</u>

activity.htm#:~:text=Not%20getting%20enough%20physical%20activity%20can%20lead%20to%20heart %20disease,cholesterol%2C%20and%20type%202%20diabetes.

118. Ussery EN, Omura JD, McCain K, Watson KB. Change in Prevalence of Meeting the Aerobic Physical Activity Guideline Among US Adults, by States and Territories—Behavioral Risk Factor Surveillance System, 2011 and 2019. *Journal of Physical Activity and Health*. 01 Aug. 2021 2021;18(S1):S84-S85. doi:10.1123/jpah.2021-0181

119. 15 U.S.C. § 1637(r)(1).

120. 20 U.S. Code § 1001(a).

121. Calestine J, Bopp M, Bopp CM, Papalia Z. College Student Work Habits are Related to Physical Activity and Fitness. *Int J Exerc Sci.* 2017;10(7):1009-1017.

122. Buckworth J, Nigg C. Physical activity, exercise, and sedentary behavior in college students. *J Am Coll Health*. Jul-Aug 2004;53(1):28-34. doi:10.3200/jach.53.1.28-34

123. Dolan A. Predictors and associations of physical activity in college students: self-efficacy, self-regulation, social support and socioeconomic status. 2017;

124. Farren GL, Zhang T, Martin SB, Thomas KT. Factors related to meeting physical activity guidelines in active college students: A social cognitive perspective. *Journal of American College Health*. 2017/01/02 2017;65(1):10-21. doi:10.1080/07448481.2016.1229320

125. Kilpatrick M, Hebert E, Bartholomew J. College students' motivation for physical activity: differentiating men's and women's motives for sport participation and exercise. *J Am Coll Health*. Sep-Oct 2005;54(2):87-94. doi:10.3200/jach.54.2.87-94

126. Kulavic K, Hultquist CN, McLester JR. A comparison of motivational factors and barriers to physical activity among traditional versus nontraditional college students. *J Am Coll Health*. 2013;61(2):60-6. doi:10.1080/07448481.2012.753890

127. Zhang Z, He Z, Chen W. The relationship between physical activity intensity and subjective wellbeing in college students. *Journal of American College Health*. 2022/05/19 2022;70(4):1241-1246. doi:10.1080/07448481.2020.1790575

128. Marmo J. Applying Social Cognitive Theory to Develop Targeted Messages: College Students and Physical Activity. *Western Journal of Communication*. 2013/07/01 2013;77(4):444-465. doi:10.1080/10570314.2012.681101

129. Martinez YTS, Harmon BE, Nigg CR, Bantum EO, Strayhorn S. Diet and Physical Activity Intervention Strategies for College Students. *Health Behav Policy Rev.* 2016;3(4):336-347. doi:10.14485/HBPR.3.4.5

130. Nehl EJ, Blanchard CM, Kupperman J, et al. Exploring Physical Activity by Ethnicity and Gender in
College Students Using Social Cognitive Theory. *ICHPER-SD Journal of Research*. 2012;7(2):11-17.
131. Olofintuyi A, Roberson KB, Ilesanmi A, et al. The Use of Text Messaging to Promote Physical

Activity in African-American College Students: A Feasibility Study. *J Natl Soc Allied Health*. Fall 2018;15(1):30-44.

132. Prokhorov AV, Machado TC, Calabro KS, et al. Developing mobile phone text messages for tobacco risk communication among college students: a mixed methods study. *BMC Public Health*. 2017;17(1):137-137. doi:10.1186/s12889-017-4027-z

133. Castro O, Vergeer I, Bennie J, Biddle SJH. Feasibility of Reducing and Breaking Up University Students' Sedentary Behaviour: Pilot Trial and Process Evaluation. Original Research. *Frontiers in Psychology*. 2021-June-10 2021;12doi:10.3389/fpsyg.2021.661994

134. Burton NW, Barber BL, Khan A. A Qualitative Study of Barriers and Enablers of Physical Activity among Female Emirati University Students. *International journal of environmental research and public health*. 2021;18(7):3380. doi:10.3390/ijerph18073380

135. Alkhateeb SA, Alkhameesi NF, Lamfon GN, et al. Pattern of physical exercise practice among university students in the Kingdom of Saudi Arabia (before beginning and during college): a cross-sectional study. *BMC Public Health*. 2019/12/21 2019;19(1):1716. doi:10.1186/s12889-019-8093-2

136. Figueroa CA, Deliu N, Chakraborty B, et al. Daily Motivational Text Messages to Promote Physical Activity in University Students: Results From a Microrandomized Trial. *Ann Behav Med*. Feb 11 2022;56(2):212-218. doi:10.1093/abm/kaab028

137. Hardan-Khalil K, Costa CB, Fisher DG. Daily motivational text messages impact on college students' nutritional awareness and physical activity levels. *Journal of American College Health*. 2021:1-8. doi:10.1080/07448481.2020.1869748

138. Joseph RP, Pekmezi DW, Lewis T, Dutton G, Turner LW, Durant NH. Physical Activity and Social Cognitive Theory Outcomes of an Internet-Enhanced Physical Activity Intervention for African American Female College Students. *J Health Dispar Res Pract*. Summer 2013;6(2):1-18.

139. Romero-Blanco C, Rodríguez-Almagro J, Onieva-Zafra MD, Parra-Fernández ML, Prado-Laguna MDC, Hernández-Martínez A. Physical Activity and Sedentary Lifestyle in University Students: Changes

during Confinement Due to the COVID-19 Pandemic. *International journal of environmental research and public health*. 2020;17(18):6567. doi:10.3390/ijerph17186567

140. Sandrick J, Tracy D, Eliasson A, et al. Effect of a Counseling Session Bolstered by Text Messaging on Self-Selected Health Behaviors in College Students: A Preliminary Randomized Controlled Trial. *JMIR Mhealth Uhealth*. 2017/05/17 2017;5(5):e67. doi:10.2196/mhealth.6638

141. 38 U.S.C. § 3452(f).

142. American College Health Association. *American College Health Association-National College Health Assessment III: Reference Group Executive Summary Fall 2021*. American College Health Association; 2022.

143. Benjamin RM. The national prevention strategy: Shifting the nation's health-care system. *Public Health Reports*. 2011;126(6):774-776.

144. Maselli M, Ward PB, Gobbi E, Carraro A. Promoting Physical Activity Among University Students: A Systematic Review of Controlled Trials. *American Journal of Health Promotion*. 2018/09/01 2018;32(7):1602-1612. doi:10.1177/0890117117753798

145. Arzu D, Tuzun EH, Eker L. Perceived barriers to physical activity in university students. *Journal of sports science & medicine*. 2006;5(4):615-620.

146. Greendale GA, Huang MH, Wang Y, Finkelstein JS, Danielson ME, Sternfeld B. Sport and home physical activity are independently associated with bone density. *Med Sci Sports Exerc*. Mar 2003;35(3):506-12. doi:10.1249/01.Mss.0000056725.64347.C9

147. Steineke T. The Effect of Exercise on College Students' Overall Health. 2019;

148. Sparling PB, Snow TK. Physical activity patterns in recent college alumni. *Res Q Exerc Sport*. Jun 2002;73(2):200-5. doi:10.1080/02701367.2002.10609009

149. Wilson OWA, Matthews PJ, Duffey M, Papalia Z, Bopp M. Changes in Health Behaviors and Outcomes following Graduation from Higher Education. *Int J Exerc Sci*. 2020;13(5):131-139.

150. Driskell JA, Kim YN, Goebel KJ. Few differences found in the typical eating and physical activity habits of lower-level and upper-level university students. *J Am Diet Assoc*. May 2005;105(5):798-801. doi:10.1016/j.jada.2005.02.004

151. Ives JC, Neese K, Downs N, Root H, Finnerty T. The effects of competitive orientation on performance in competition. *The sport journal*. 2020;41(2)

152. Smith KG, Ferrier WJ, Ndofor H. Competitive dynamics research: Critique and future directions. *Handbook of strategic management*. 2001;315:361.

153. Zhang J, Brackbill D, Yang S, Becker J, Herbert N, Centola D. Support or competition? How online social networks increase physical activity: A randomized controlled trial. *Preventive Medicine Reports*. 2016/12/01/ 2016;4:453-458. doi:<u>https://doi.org/10.1016/j.pmedr.2016.08.008</u>

154. Festinger L. A Theory of Social Comparison Processes. *Human Relations*. 1954/05/01 1954;7(2):117-140. doi:10.1177/001872675400700202

155. González-Valero G, Ubago-Jiménez JL, Ramírez-Granizo IA, Puertas-Molero P. Association between Motivational Climate, Adherence to Mediterranean Diet, and Levels of Physical Activity in Physical Education Students. *Behavioral Sciences*. 2019;9(4):37.

156. DiMenichi BC, Tricomi E. The power of competition: Effects of social motivation on attention, sustained physical effort, and learning. Original Research. *Frontiers in Psychology*. 2015-September-01 2015;6doi:10.3389/fpsyg.2015.01282

157. Leahey TM, Kumar R, Weinberg BM, Wing RR. Teammates and social influence affect weight loss outcomes in a team-based weight loss competition. *Obesity (Silver Spring)*. 2012;20(7):1413-1418. doi:10.1038/oby.2012.18

158. Shakya HB, Christakis NA, Fowler JH. Self-comparisons as motivators for healthy behavior. *Obesity (Silver Spring)*. Dec 2015;23(12):2477-84. doi:10.1002/oby.21201

159. Prestwich A, Conner M, Morris B, Finlayson G, Sykes-Muskett B, Hurling R. Do web-based competitions promote physical activity? Randomized controlled trial. *Psychology of Sport and Exercise*. 2017/03/01/ 2017;29:1-9. doi:<u>https://doi.org/10.1016/j.psychsport.2016.11.003</u>

160. Patel MS, Small DS, Harrison JD, et al. Effectiveness of Behaviorally Designed Gamification Interventions With Social Incentives for Increasing Physical Activity Among Overweight and Obese Adults Across the United States: The STEP UP Randomized Clinical Trial. *JAMA Internal Medicine*. 2019;179(12):1624-1632. doi:10.1001/jamainternmed.2019.3505

161. Epstein JA, Harackiewicz JM. Winning is Not Enough: The Effects of Competition and Achievement Orientation on Intrinsic Interest. *Personality and Social Psychology Bulletin*. 1992/04/01 1992;18(2):128-138. doi:10.1177/0146167292182003

162. Reeve J, Deci EL. Elements of the Competitive Situation that Affect Intrinsic Motivation. *Personality and Social Psychology Bulletin*. 1996/01/01 1996;22(1):24-33. doi:10.1177/0146167296221003

163. Hibbert AW, Billaut F, Varley MC, Polman RCJ. Goal Orientation and the Presence of Competitors Influence Cycling Performance. *Frontiers in psychology*. 2018;9:1212-1212. doi:10.3389/fpsyg.2018.01212

164. Gill DL, Kelley BC, Martin JJ, Caruso CM. A comparison of competitive-orientation measures. *Journal of Sport & Exercise Psychology*. 1991;13(3):266-280.

165. Song H, Kim J, Tenzek KE, Lee KM. The effects of competition and competitiveness upon intrinsic motivation in exergames. *Computers in Human Behavior*. 2013/07/01/ 2013;29(4):1702-1708. doi:<u>https://doi.org/10.1016/j.chb.2013.01.042</u>

166. Rüth M, Kaspar K. Educational and Social Exergaming: A Perspective on Physical, Social, and Educational Benefits and Pitfalls of Exergaming at Home During the COVID-19 Pandemic and Afterwards. Perspective. *Frontiers in Psychology*. 2021-April-09 2021;12doi:10.3389/fpsyg.2021.644036

167. Snyder AL, Anderson-Hanley C, Arciero PJ. Virtual and live social facilitation while exergaming: competitiveness moderates exercise intensity. *J Sport Exerc Psychol*. Apr 2012;34(2):252-9. doi:10.1123/jsep.34.2.252

168. Anderson-Hanley C, Snyder AL, Nimon JP, Arciero PJ. Social facilitation in virtual realityenhanced exercise: competitiveness moderates exercise effort of older adults. *Clin Interv Aging*. 2011;6:275-280. doi:10.2147/CIA.S25337

169. Chen J-L, Guedes CM, Cooper BA, Lung AE. Short-Term Efficacy of an Innovative Mobile Phone Technology-Based Intervention for Weight Management for Overweight and Obese Adolescents: Pilot Study. *Interact J Med Res*. 2017/08/02 2017;6(2):e12. doi:10.2196/ijmr.7860

170. Ludwig K, Arthur R, Sculthorpe N, Fountain H, Buchan DS. Text Messaging Interventions for Improvement in Physical Activity and Sedentary Behavior in Youth: Systematic Review. *JMIR Mhealth Uhealth*. 2018/09/17 2018;6(9):e10799. doi:10.2196/10799

171. Head KJ, Noar SM, Iannarino NT, Grant Harrington N. Efficacy of text messaging-based interventions for health promotion: A meta-analysis. *Social Science & Medicine*. 2013/11/01/2013;97:41-48. doi:<u>https://doi.org/10.1016/j.socscimed.2013.08.003</u>

172. Eze CE, West BT, Dorsch MP, et al. Predictors of Smartphone and Tablet Use Among Patients With Hypertension: Secondary Analysis of Health Information National Trends Survey Data. *Journal of Medical Internet Research*. 2022;24(1):e33188.

173. Tindell DR, Bohlander RW. The Use and Abuse of Cell Phones and Text Messaging in the Classroom: A Survey of College Students. *College Teaching*. 2012/01/01 2012;60(1):1-9. doi:10.1080/87567555.2011.604802

174. Harrison MA, Gilmore AL. U txt WHEN? College students' social contexts of text messaging. *The Social Science Journal*. 2012/12/01 2012;49(4):513-518. doi:10.1016/j.soscij.2012.05.003

175. Hanson TL, Drumheller K, Mallard J, McKee C, Schlegel P. Cell Phones, Text Messaging, and Facebook: Competing Time Demands of Today's College Students. *College Teaching*. 2010/12/27 2010;59(1):23-30. doi:10.1080/87567555.2010.489078

176. Chou Y-L, Lin Y-H, Lin T-Y, You HY, Chang Y-J. Why Did You/I Read but Not Reply? IM Users' Unresponded-to Read-receipt Practices and Explanations of Them. 2022:1-15.

177. Morton K, Sutton S, Hardeman W, et al. A Text-Messaging and Pedometer Program to Promote Physical Activity in People at High Risk of Type 2 Diabetes: The Development of the PROPELS Follow-On Support Program. *JMIR mHealth* uHealth. 2015/12/15 2015;3(4):e105. doi:10.2196/mhealth.5026

178. Milton K, Bauman AE, Faulkner G, et al. Maximising the impact of global and national physical activity guidelines: the critical role of communication strategies. *British journal of sports medicine*. 2020;54(24):1463-1467. doi:10.1136/bjsports-2020-102324

179. Buchholz SW, Wilbur J, Ingram D, Fogg L. Physical activity text messaging interventions in adults: a systematic review. *Worldviews on Evidence -Based Nursing*. 2013;10(3):163-173.

180. Williamson C, Baker G, Mutrie N, Niven A, Kelly P. Get the message? A scoping review of physical activity messaging. *International Journal of Behavioral Nutrition and Physical Activity*. 2020/04/15 2020;17(1):51. doi:10.1186/s12966-020-00954-3

181. Latimer AE, Brawley LR, Bassett RL. A systematic review of three approaches for constructing physical activity messages: What messages work and what improvements are needed? *International Journal of Behavioral Nutrition and Physical Activity*. 2010/05/11 2010;7(1):36. doi:10.1186/1479-5868-7-36

182. Orr JA, King RJ. Mobile phone SMS messages can enhance healthy behaviour: a meta-analysis of randomised controlled trials. *Health Psychology Review*. 2015/11/06 2015;9(4):397-416. doi:10.1080/17437199.2015.1022847

183. Fanning J, Mullen SP, McAuley E. Increasing Physical Activity With Mobile Devices: A Meta-Analysis. *J Med Internet Res*. 2012/11/21 2012;14(6):e161. doi:10.2196/jmir.2171

184. Williamson C, Baker G, Tomasone JR, et al. The Physical Activity Messaging Framework (PAMF) and Checklist (PAMC): International consensus statement and user guide. *International Journal of Behavioral Nutrition and Physical Activity*. 2021/12/19 2021;18(1):164. doi:10.1186/s12966-021-01230-8

185. Davies CA, Spence JC, Vandelanotte C, Caperchione CM, Mummery WK. Meta-analysis of internet-delivered interventions to increase physical activity levels. *International Journal of Behavioral Nutrition and Physical Activity*. 2012/04/30 2012;9(1):52. doi:10.1186/1479-5868-9-52

186. Norman GJ, Zabinski MF, Adams MA, Rosenberg DE, Yaroch AL, Atienza AA. A Review of eHealth Interventions for Physical Activity and Dietary Behavior Change. *American Journal of Preventive Medicine*. 2007;33(4):336-345.e16. doi:10.1016/j.amepre.2007.05.007

187. Ferney SL, Marshall AL. Website physical activity interventions: preferences of potential users. *Health Education Research*. 2006;21(4):560-566. doi:10.1093/her/cyl013

188. van den Berg MH, Schoones JW, Vliet Vlieland TPM. Internet-Based Physical Activity Interventions: A Systematic Review of the Literature. *J Med Internet Res*. 2007/09/30 2007;9(3):e26. doi:10.2196/jmir.9.3.e26

189. Wantland DJ, Portillo CJ, Holzemer WL, Slaughter R, McGhee EM. The Effectiveness of Web-Based vs. Non-Web-Based Interventions: A Meta-Analysis of Behavioral Change Outcomes. *J Med Internet Res*. 2004/11/10 2004;6(4):e40. doi:10.2196/jmir.6.4.e40

190. Kuijpers W, Groen WG, Aaronson NK, van Harten WH. A Systematic Review of Web-Based Interventions for Patient Empowerment and Physical Activity in Chronic Diseases: Relevance for Cancer Survivors. *J Med Internet Res*. 2013/02/20 2013;15(2):e37. doi:10.2196/jmir.2281 191. Smith-Tran A. "Finally Something for Us": Black Girls Run! and Racialized Space-Making in Recreational Running. *Journal of Sport and Social Issues*. 2021/06/01 2020;45(3):235-250. doi:10.1177/0193723519899241

192. Driggers S. Alabama Providers Prescribe Physical Activity Via the Parks Rx Program. Accessed June 10, 2022. <u>https://nccd.cdc.gov/nccdsuccessstories/TemplateThree.aspx?s=13938&ds=1</u>

193. Centers for Disease Control and Prevention. AD/MATERIAL: Birmingham REACH for Better Health: Outdoor Physical Activity Accessed June 10, 2022.

https://nccd.cdc.gov/schmc/apps/searchdetails.aspx?CatalogID=5340

194. Garber MD, Flanders WD, Watkins KE, Lobelo F, Kramer MR, McCullough LE. Have Paved Trails and Protected Bike Lanes Led to More Bicycling in Atlanta?: A Generalized Synthetic-Control Analysis. *Epidemiology*. 2022;33(4):493-504.

195. Hanson M. College Enrollment & Student Demographic Statistics. Updated April 22. Accessed June 10, 2022. <u>https://educationdata.org/college-enrollment-statistics</u>

196. Ramirez O, Bernhardt V. Using Fitbit Competitions to Increase Physical Activity in College Students. *Johns Hopkins University*. 2020;1(1):21773.

197. Grim M, Hortz B, Petosa R. Impact evaluation of a pilot web-based intervention to increase physical activity. *American Journal of Health Promotion*. 2011;25(4):227-230.

198. Sáez I, Solabarrieta J, Rubio I. Motivation for Physical Activity in University Students and Its Relation with Gender, Amount of Activities, and Sport Satisfaction. *Sustainability*. 2021;13(6):3183.

199. Di Bartolomeo G, Papa S. The Effects of Physical Activity on Social Interactions: The Case of Trust and Trustworthiness. *Journal of Sports Economics*. 2019/01/01 2017;20(1):50-71. doi:10.1177/1527002517717299

200. Mullen SP, Olson EA, Phillips SM, et al. Measuring enjoyment of physical activity in older adults: invariance of the physical activity enjoyment scale (paces) across groups and time. *International Journal of Behavioral Nutrition and Physical Activity*. 2011/09/27 2011;8(1):103. doi:10.1186/1479-5868-8-103

201. Teixeira PJ, Carraça EV, Markland D, Silva MN, Ryan RM. Exercise, physical activity, and selfdetermination theory: A systematic review. *International Journal of Behavioral Nutrition and Physical Activity*. 2012/06/22 2012;9(1):78. doi:10.1186/1479-5868-9-78

202. Pressman SD, Matthews KA, Cohen S, et al. Association of enjoyable leisure activities with psychological and physical well-being. *Psychosom Med*. 2009;71(7):725-732. doi:10.1097/PSY.0b013e3181ad7978

203. Diehl K, Fuchs AK, Rathmann K, Hilger-Kolb J. Students' Motivation for Sport Activity and Participation in University Sports: A Mixed-Methods Study. *Biomed Res Int*. 2018;2018:9524861-9524861. doi:10.1155/2018/9524861

204. Zurita-Ortega F, Badicu G, Chacón-Cuberos R, Castro-Sánchez M. Motivational Climate and Physical Activity: A Multigroup Analysis in Romanian and Spanish University Students. *International journal of environmental research and public health*. 2019;16(11):2013. doi:10.3390/ijerph16112013
205. Rodenbaugh TA. The Barriers and Facilitators to Physical Activity on Campus: An Analysis of UA PhotoVoice Research. 2016:

206. Bauman AE, Reis RS, Sallis JF, Wells JC, Loos RJF, Martin BW. Correlates of physical activity: why are some people physically active and others not? *The Lancet*. 2012;380(9838):258-271. doi:10.1016/S0140-6736(12)60735-1

207. Hernández-Gallardo D, Arencibia-Moreno R, Linares-Girela D, Saca-Plua IJ, Linares-Manrique M. Energy Expenditure and Physical Activity in a University Population in the Coastal Region of Ecuador. *Sustainability*. 2020;12(23):10165.

208. Sogari G, Velez-Argumedo C, Gómez MI, Mora C. College Students and Eating Habits: A Study Using An Ecological Model for Healthy Behavior. *Nutrients*. 2018;10(12):1823.

209. von Sommoggy J, Rueter J, Curbach J, Helten J, Tittlbach S, Loss J. How Does the Campus Environment Influence Everyday Physical Activity? A Photovoice Study Among Students of Two German Universities. *Frontiers in public health*. 2020;8:561175-561175. doi:10.3389/fpubh.2020.561175

210. Hammond RA. Peer reviewed: complex systems modeling for obesity research. *Preventing chronic disease*. 2009;6(3)

211. Sallis JF, Bowles HR, Bauman A, et al. Neighborhood environments and physical activity among adults in 11 countries. *American journal of preventive medicine*. 2009;36(6):484-490.

212. Ding D, Sallis JF, Kerr J, Lee S, Rosenberg DE. Neighborhood environment and physical activity among youth: a review. *American journal of preventive medicine*. 2011;41(4):442-455.

213. Wood R, Bandura A. Social cognitive theory of organizational management. *Academy of management Review*. 1989;14(3):361-384.

214. LaMorte WW. The Social Cognitive Theory. Updated September 9. Accessed June 1, 2022. https://sphweb.bumc.bu.edu/otlt/mph-

modules/sb/behavioralchangetheories/behavioralchangetheories5.html#:~:text=The%20goal%20of%20 SCT%20is,the%20theory%20evolved%20into%20SCT.

215. Bandura A. Social Learning Theory of Aggression. *Journal of Communication*. 2006;28(3):12-29. doi:10.1111/j.1460-2466.1978.tb01621.x

216. Lantolf JP, Thorne SL, Poehner ME. Sociocultural theory and second language development. *Theories in second language acquisition*. Routledge; 2014:221-240.

217. Bandura A. Social cognitive theory. *Handbook of social psychological theories*. 2011;2012:349-373.

218. Bormann EG, Cragan JF, Shields DC. Three decades of developing, grounding, and using symbolic convergence theory (SCT). *Annals of the International Communication Association*. 2001;25(1):271-313.

219. Webber-Ritchey KJ, Taylor-Piliae RE, Loescher LJ. Physical Activity in Parents of Young African American Children: The Application of Social Cognitive Theory. *Res Theory Nurs Pract*. (1):63-81. doi:10.1891/1541-6577.32.1.63

220. Luszczynska A, Schwarzer R. Social cognitive theory. *Fac Health Sci Publ*. 2015:225-51.

221. Young MD, Plotnikoff RC, Collins CE, Callister R, Morgan PJ. Social cognitive theory and physical activity: a systematic review and meta-analysis. <u>https://doi.org/10.1111/obr.12225</u>. *Obesity Reviews*. 2014/12/01 2014;15(12):983-995. doi:<u>https://doi.org/10.1111/obr.12225</u>

222. Bandura A. Health promotion from the perspective of social cognitive theory. *Psychology & Health*. 1998/07/01 1998;13(4):623-649. doi:10.1080/08870449808407422

223. Rovniak LS, Anderson ES, Winett RA, Stephens RS. Social cognitive determinants of physical activity in young adults: a prospective structural equation analysis. *Ann Behav Med*. Spring 2002;24(2):149-56. doi:10.1207/s15324796abm2402_12

224. Petosa RL, Suminski R, Hortz B. Predicting vigorous physical activity using social cognitive theory. *American Journal of Health Behavior*. 2003;27(4):301-310.

225. Sriramatr S, Silalertdeskul S, Wachirathanin P. Social cognitive theory associated with physical activity in undergraduate students: A cross-sectional study. *Pacific Rim International Journal of Nursing Research*. 2016;20(2):95-105.

226. Choi JY, Chang AK, Choi E-J. Sex differences in social cognitive factors and physical activity in Korean college students. *Journal of Physical Therapy Science*. 2015;27(6):1659-1664.

227. Wallace LS, Buckworth J, Kirby TE, Sherman WM. Characteristics of Exercise Behavior among College Students: Application of Social Cognitive Theory to Predicting Stage of Change. *Prev Med*. 2000/11/01/ 2000;31(5):494-505. doi:<u>https://doi.org/10.1006/pmed.2000.0736</u>

228. Wang T, Ren M, Shen Y, et al. The Association Among Social Support, Self-Efficacy, Use of Mobile Apps, and Physical Activity: Structural Equation Models With Mediating Effects. *JMIR Mhealth Uhealth*. 2019/09/25 2019;7(9):e12606. doi:10.2196/12606

229. Smith DM, Duque L, Huffman JC, Healy BC, Celano CM. Text Message Interventions for Physical Activity: A Systematic Review and Meta-Analysis. *American Journal of Preventive Medicine*. 2020/01/01/2020;58(1):142-151. doi:<u>https://doi.org/10.1016/j.amepre.2019.08.014</u>

230. Buchholz SW, Wilbur J, Ingram D, Fogg L. Physical Activity Text Messaging Interventions in Adults: A Systematic Review. <u>https://doi.org/10.1111/wvn.12002</u>. *Worldviews on Evidence-Based Nursing*. 2013/08/01 2013;10(3):163-173. doi:https://doi.org/10.1111/wvn.12002

231. Lau PWC, Lau EY, Wong DP, Ransdell L. A Systematic Review of Information and Communication Technology–Based Interventions for Promoting Physical Activity Behavior Change in Children and Adolescents. *J Med Internet Res*. 2011/07/13 2011;13(3):e48. doi:10.2196/jmir.1533

232. He Z, Wu H, Yu F, et al. Effects of Smartphone-Based Interventions on Physical Activity in Children and Adolescents: Systematic Review and Meta-analysis. *JMIR Mhealth Uhealth*. 2021/2/1 2021;9(2):e22601. doi:10.2196/22601

233. Kwan RYC, Salihu D, Lee PH, et al. The effect of e-health interventions promoting physical activity in older people: a systematic review and meta-analysis. *European Review of Aging and Physical Activity*. 2020;17(1):1-17.

234. Muellmann S, Forberger S, Möllers T, Bröring E, Zeeb H, Pischke CR. Effectiveness of eHealth interventions for the promotion of physical activity in older adults: a systematic review. *Prev Med*. 2018;108:93-110.

235. Shariff-Ghazali S, Browning CJ, Yasin S. Interventions to promote physical activity in older people with type 2 diabetes mellitus: a systematic review. *Frontiers in public health*. 2013;1:71.

236. Connelly J, Kirk A, Masthoff J, MacRury S. The use of technology to promote physical activity in Type 2 diabetes management: a systematic review. <u>https://doi.org/10.1111/dme.12289</u>. *Diabetic Medicine*. 2013/12/01 2013;30(12):1420-1432. doi:<u>https://doi.org/10.1111/dme.12289</u>

237. Bort-Roig J, Gilson ND, Puig-Ribera A, Contreras RS, Trost SG. Measuring and Influencing Physical Activity with Smartphone Technology: A Systematic Review. *Sports Medicine*. 2014/05/01 2014;44(5):671-686. doi:10.1007/s40279-014-0142-5

238. Muntaner A, Vidal-Conti J, Palou P. Increasing physical activity through mobile device interventions: A systematic review. *Health Informatics Journal*. 2016/09/01 2015;22(3):451-469. doi:10.1177/1460458214567004

239. Müller AM, Alley S, Schoeppe S, Vandelanotte C. The effectiveness of e-& mHealth interventions to promote physical activity and healthy diets in developing countries: A systematic review. *International Journal of Behavioral Nutrition and Physical Activity*. 2016/10/10 2016;13(1):109. doi:10.1186/s12966-016-0434-2

240. Hutton B, Salanti G, Caldwell DM, et al. The PRISMA extension statement for reporting of systematic reviews incorporating network meta-analyses of health care interventions: checklist and explanations. *Ann Intern Med.* Jun 2 2015;162(11):777-84. doi:10.7326/m14-2385

241. Ben Van Den A. Cochrane Handbook for Systematic Reviews of Interventions. *International Coaching Psychology Review*. 2020;15(2):123-125.

242. Andrade C. Mean Difference, Standardized Mean Difference (SMD), and Their Use in Meta-Analysis: As Simple as It Gets. *J Clin Psychiatry*. Sep 22 2020;81(5)doi:10.4088/JCP.20f13681

243. Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *Bmj*. 2003;327(7414):557-560.

244. Ross A, Cooper C, Gray H, Umberham B, Vassar M. Assessment of Publication Bias and Systematic Review Findings in Top-Ranked Otolaryngology Journals. *JAMA Otolaryngology–Head & Neck Surgery*. 2019;145(2):187-188. doi:10.1001/jamaoto.2018.3301

245. Dubben H-H, Beck-Bornholdt H-P. Systematic review of publication bias in studies on publication bias. *BMJ*. 2005;331(7514):433-434. doi:10.1136/bmj.38478.497164.F7

246. Dwan K, Altman DG, Arnaiz JA, et al. Systematic review of the empirical evidence of study publication bias and outcome reporting bias. *PloS one*. 2008;3(8):e3081.

247. Dwan K, Gamble C, Williamson PR, Kirkham JJ, Group RB. Systematic review of the empirical evidence of study publication bias and outcome reporting bias—an updated review. *PloS one*. 2013;8(7):e66844.

248. Soeken KL, Sripusanapan A. Assessing Publication Bias in Meta-Analysis. *Nursing Research*. 2003;52(1)

249. Lin L, Chu H. Quantifying publication bias in meta-analysis. *Biometrics*. Sep 2018;74(3):785-794. doi:10.1111/biom.12817

Sterne JAC, Egger M. Funnel plots for detecting bias in meta-analysis: Guidelines on choice of axis. *Journal of Clinical Epidemiology*. 2001;54(10):1046-1055. doi:10.1016/S0895-4356(01)00377-8
Borenstein M, Hedges LV, Higgins JP, Rothstein HR. A basic introduction to fixed-effect and

random-effects models for meta-analysis. *Res Synth Methods*. 2010;1(2):97-111.

252. Higgins JP, Thomas J, Chandler J, et al. *Cochrane handbook for systematic reviews of interventions*. John Wiley & Sons; 2019.

253. Riley RD, Higgins JPT, Deeks JJ. Interpretation of random effects meta-analyses. *BMJ*. 2011;342:d549. doi:10.1136/bmj.d549

254. Collaboration C. 9.5. 4 Incorporating heterogeneity into random-effects models. *Cochrane Handbook for Systematic Reviews of Interventions*. 2020;5(0)

255. Daniel R, McKechnie T, Kruse CC, et al. Video-based coaching for surgical residents: a systematic review and meta-analysis. *Surgical Endoscopy*. 2022/06/23 2022;doi:10.1007/s00464-022-09379-4

256. McDonagh M PK, Raina P, et al. Avoiding Bias in Selecting Studies In: Methods Guide for Effectiveness and Comparative Effectiveness Reviews. Rockville (MD): Agency for Healthcare Research and Quality (US). Updated February 20. Accessed December 22, 2020.

https://www.ncbi.nlm.nih.gov/books/NBK126701/

257. Créquit P, Trinquart L, Yavchitz A, Ravaud P. Wasted research when systematic reviews fail to provide a complete and up-to-date evidence synthesis: the example of lung cancer. *BMC Medicine*. 2016/01/20 2016;14(1):8. doi:10.1186/s12916-016-0555-0

258. Tawfik GM, Dila KAS, Mohamed MYF, et al. A step by step guide for conducting a systematic review and meta-analysis with simulation data. *Tropical Medicine and Health*. 2019/08/01 2019;47(1):46. doi:10.1186/s41182-019-0165-6

259. Jin Z-C, Zhou X-H, He J. Statistical methods for dealing with publication bias in meta-analysis. *Statistics in Medicine*. 2015;34(2):343-360. doi:<u>https://doi.org/10.1002/sim.6342</u>

260. Debray TPA, Moons KGM, Riley RD. Detecting small-study effects and funnel plot asymmetry in meta-analysis of survival data: A comparison of new and existing tests. *Res Synth Methods*. 2018;9(1):41-50. doi:10.1002/jrsm.1266

Buchholz SW, Wilbur J, Ingram D, Fogg L. Physical activity text messaging interventions in adults:
a systematic review. *Worldviews Evid Based Nurs*. Aug 2013;10(3):163-73. doi:10.1111/wvn.12002
Abdi J, Eftekhar H, Estebsari F, Sadeghi R. Theory-based interventions in physical activity: a

systematic review of literature in Iran. *Glob J Health Sci.* 2014;7(3):215-229. doi:10.5539/gjhs.v7n3p215
263. Gourlan M, Bernard P, Bortolon C, et al. Efficacy of theory-based interventions to promote physical activity. A meta-analysis of randomised controlled trials. *Health Psychology Review*. 2016/01/02

2016;10(1):50-66. doi:10.1080/17437199.2014.981777

264. Dalgetty R, Miller CB, Dombrowski SU. Examining the theory-effectiveness hypothesis: A systematic review of systematic reviews. <u>https://doi.org/10.1111/bjhp.12356</u>. *British Journal of Health Psychology*. 2019/05/01 2019;24(2):334-356. doi:<u>https://doi.org/10.1111/bjhp.12356</u>

265. Bully P, Sánchez Á, Zabaleta-del-Olmo E, Pombo H, Grandes G. Evidence from interventions based on theoretical models for lifestyle modification (physical activity, diet, alcohol and tobacco use) in

primary care settings: A systematic review. *Prev Med*. 2015/07/01/ 2015;76:S76-S93. doi:<u>https://doi.org/10.1016/j.ypmed.2014.12.020</u>

266. Armanasco AA, Miller YD, Fjeldsoe BS, Marshall AL. Preventive Health Behavior Change Text Message Interventions: A Meta-analysis. *American Journal of Preventive Medicine*. 2017/03/01/ 2017;52(3):391-402. doi:<u>https://doi.org/10.1016/j.amepre.2016.10.042</u>

267. Dhand NKK, M. S. Statulator: An online statistical calculator. Sample Size Calculator for
Comparing Two Paired Means. Accessed 10 June 2022. <u>http://statulator.com/SampleSize/ss2PM.html</u>
268. ChallengeRunner. Free Online Fitness Challenge Management System. Accessed June 19, 2022.

https://www.challengerunner.com/

269. Cotten E, Prapavessis H. Increasing Nonsedentary Behaviors in University Students Using Text Messages: Randomized Controlled Trial. *JMIR Mhealth Uhealth*. 2016/08/19 2016;4(3):e99. doi:10.2196/mhealth.5411

270. Keahey R, White N, Duchesne A, Pelletier CA. A theory-grounded text message–based intervention to reduce sedentary behaviour in university students. *Health Education Journal*. 2021/10/01 2021;80(6):672-685. doi:10.1177/00178969211007163

271. Nam SJ, Cha C. Effects of a social-media-based support on premenstrual syndrome and physical activity among female university students in South Korea. *Journal of Psychosomatic Obstetrics & Gynecology*. 2020/01/02 2020;41(1):47-53. doi:10.1080/0167482X.2018.1559811

272. Hebden L, Cook A, van der Ploeg HP, King L, Bauman A, Allman-Farinelli M. A mobile health intervention for weight management among young adults: a pilot randomised controlled trial. <u>https://doi.org/10.1111/jhn.12155</u>. *Journal of Human Nutrition and Dietetics*. 2014/08/01 2014;27(4):322-332. doi:https://doi.org/10.1111/jhn.12155

273. Centers for Disease Control and Prevention. Fact Sheets & Infographics. Updated July 1. Accessed **June 18**, 2022. <u>https://www.cdc.gov/physicalactivity/resources/factsheets.html</u>

274. Segar ML, Updegraff JA, Zikmund-Fisher BJ, Richardson CR. Physical Activity Advertisements That Feature Daily Well-Being Improve Autonomy and Body Image in Overweight Women but Not Men. *Journal of Obesity*. 2012/06/04 2012;2012:354721. doi:10.1155/2012/354721

275. Cunningham GB, Woods J. FOR THE HEALTH OF IT: ADVERTISEMENT MESSAGE AND ATTRACTION TO FITNESS CLUBS. *American Journal of Health Studies*. 2011;26(1)

276. Jeminiwa R. Examining the Effects of Message Framing on Medication Beliefs, Intentions to Take Medication, Medication Adherence, and Asthma Control among College Students. Auburn University; 2020.

277. Jones KE, Evans R, Forbes L, Schoenberger Y-M, Heaton K, Snyder S. Research on freshman and sleeping habits: A text message-based sleep intervention. *Journal of American College Health*. 2020/11/16 2020;68(8):864-871. doi:10.1080/07448481.2019.1626860

278. Wilke DJ, Radey M, Langenderfer-Magruder L. Recruitment and retention of child welfare workers in longitudinal research: Successful strategies from the Florida Study of Professionals for Safe Families. *Children and Youth Services Review*. 2017/07/01/ 2017;78:122-128. doi:https://doi.org/10.1016/j.childyouth.2017.05.013

279. Seiter CR. Benefit, Barrier, and Self-Efficacy Messages in Advance Care Planning Education Materials. *Health Commun*. Oct 2021;36(11):1397-1404. doi:10.1080/10410236.2020.1754601

280. Yan AF, Stevens P, Wang Y, et al. mHealth Text Messaging for Physical Activity Promotion in College Students: A Formative Participatory Approach. *American Journal of Health Behavior*. // 2015;39(3):395-408. doi:10.5993/AJHB.39.3.12

281. García-Martín MÁ, Hombrados-Mendieta I, Gómez-Jacinto L. A multidimensional approach to social support: the Questionnaire on the Frequency of and Satisfaction with Social Support (QFSSS). *Anales De Psicología/Annals of Psychology*. 2016;32(2):501-515.

282. Stadler G, Oettingen G, Gollwitzer PM. Physical Activity in Women: Effects of a Self-Regulation Intervention. *American Journal of Preventive Medicine*. 2009/01/01/ 2009;36(1):29-34. doi:<u>https://doi.org/10.1016/j.amepre.2008.09.021</u>

283. Resnick B, Jenkins LS. Testing the Reliability and Validity of the Self-Efficacy for Exercise Scale. *Nursing Research*. 2000;49(3)

284. Frederick CM, Ryan RM. Differences in motivation for sport and exercise and their relations with participation and mental health. *Journal of sport behavior*. 1993;16(3):124-147.

285. Haughton McNeill L, Wyrwich KW, Brownson RC, Clark EM, Kreuter MW. Individual, social environmental, and physical environmental influences on physical activity among black and white Adults: A Structural Equation Analysis. *Annals of Behavioral Medicine*. 2006;31(1):36-44. doi:10.1207/s15324796abm3101_7

286. Hall T, Krahn GL, Horner-Johnson W, Lamb G. Examining functional content in widely used Health-Related Quality of Life scales. *Rehabil Psychol*. May 2011;56(2):94-99. doi:10.1037/a0023054

287. Topp CW, Østergaard SD, Søndergaard S, Bech P. The WHO-5 Well-Being Index: A Systematic Review of the Literature. *Psychotherapy and Psychosomatics*. 2015;84(3):167-176. doi:10.1159/000376585

288. Haun J, Luther S, Dodd V, Donaldson P. Measurement variation across health literacy assessments: implications for assessment selection in research and practice. *J Health Commun*. 2012;17 Suppl 3:141-59. doi:10.1080/10810730.2012.712615

289. Greenwood JL, Joy EA, Stanford JB. The physical activity vital sign: a primary care tool to guide counseling for obesity. *Journal of Physical Activity and Health*. 2010;7(5):571-576.

290. Fung S-f, Kong CYW, Liu Y-m, et al. Validity and Psychometric Evaluation of the Chinese Version of the 5-Item WHO Well-Being Index. Original Research. *Frontiers in Public Health*. 2022;10

291. Tennant R, Hiller L, Fishwick R, et al. The Warwick-Edinburgh Mental Well-being Scale (WEMWBS): development and UK validation. *Health and Quality of Life Outcomes*. 2007/11/27 2007;5(1):63. doi:10.1186/1477-7525-5-63

292. Blake H, Stanulewicz N, Mcgill F. Predictors of physical activity and barriers to exercise in nursing and medical students. *Journal of Advanced Nursing*. 2017;73(4):917-929. doi:https://doi.org/10.1111/jan.13181

293. Lee L-L, Perng S-J, Ho C-C, Hsu H-M, Lau S-C, Arthur A. A preliminary reliability and validity study of the Chinese version of the self-efficacy for exercise scale for older adults. *International Journal of Nursing Studies*. 2009/02/01/ 2009;46(2):230-238. doi:<u>https://doi.org/10.1016/j.ijnurstu.2008.09.003</u>

294. Wilcox S, Sharpe PA, Hutto B, Granner ML. Psychometric Properties of the Self-Efficacy for Exercise Questionnaire in a Diverse Sample of Men and Women. *Journal of Physical Activity and Health*. 01 Jul. 2005 2005;2(3):285-297. doi:10.1123/jpah.2.3.285

295. Molanorouzi K, Khoo S, Morris T. Validating the Physical Activity and Leisure Motivation Scale (PALMS). *BMC Public Health*. 2014/09/03 2014;14(1):909. doi:10.1186/1471-2458-14-909

296. Noroozi A, Ghofranipour F, Heydarnia AR, Nabipour I, Shokravi FA. Validity and Reliability of the Social Support Scale for Exercise Behavior in Diabetic Women. *Asia Pacific Journal of Public Health*. 2011/09/01 2010;23(5):730-741. doi:10.1177/1010539509357342

297. Gothe NP. Correlates of Physical Activity in Urban African American Adults and Older Adults: Testing the Social Cognitive Theory. *Annals of Behavioral Medicine*. 2018;52(9):743-751. doi:10.1093/abm/kax038

298. Arovah NI, Heesch KC. Social cognitive measures related to exercise behaviour: Validation in Indonesian middle-aged and older adults. *Clinical Epidemiology and Global Health*. 2022/03/01/2022;14:100975. doi:<u>https://doi.org/10.1016/j.cegh.2022.100975</u>

299. Liu J, Zeng M, Wang D, Zhang Y, Shang B, Ma X. Applying Social Cognitive Theory in Predicting Physical Activity Among Chinese Adolescents: A Cross-Sectional Study With Multigroup Structural

Equation Model. Original Research. *Frontiers in Psychology*. 2022-March-15 2022;12doi:10.3389/fpsyg.2021.695241

300. Park C-H, Elavsky S, Koo K-M. Factors influencing physical activity in older adults. *J Exerc Rehabil*. 02 2014;10(1):45-52. doi:10.12965/jer.140089

301. Walker TJ, Heredia NI, Reininger BM. Examining the Validity, Reliability, and Measurement Invariance of the Social Support for Exercise Scale among Spanish- and English- language Hispanics. *Hisp J Behav Sci*. 2019;41(3):427-443. doi:10.1177/0739986319854144

302. Kosteli M-C, Cumming J, Williams SE. Self-Regulatory Imagery and Physical Activity in Middle-Aged and Older Adults: A Social-Cognitive Perspective. *Journal of Aging and Physical Activity*. 01 Jan. 2018 2018;26(1):14-24. doi:10.1123/japa.2016-0024

303. Sriramatr S, Berry TR, Spence JC. An Internet-Based Intervention for Promoting and Maintaining Physical Activity: A Randomized Controlled Trial. *American Journal of Health Behavior*. // 2014;38(3):430-439. doi:10.5993/AJHB.38.3.12

304. Sriramatr S, Berry TR, Rodgers WM. Validity and Reliability of Thai Versions of Questionnaires Measuring Leisure-time Physical Activity, Exercise-Related Self-Efficacy, Outcome Expectations and Self-Regulation. *Pacific Rim International Journal of Nursing Research*. 08/28 2013;17(3):203-216.

305. Barrett P. Structural equation modelling: Adjudging model fit. *Personality and Individual differences*. 2007;42(5):815-824.

306. Stanley LM, Edwards MC. Reliability and Model Fit. *Educ Psychol Meas*. Dec 2016;76(6):976-985. doi:10.1177/0013164416638900

307. Kaplan D. Evaluating and modifying covariance structure models: A review and recommendation. *Multivariate Behavioral Research*. 1990;25(2):137-155.

308. Wen Z, Hau K-T, Herbert WM. Structural equation model testing: Cutoff criteria for goodness of fit indices and chi-square test. *Acta psychologica sinica*. 2004;36(02):186.

309. Kenny DA, Kaniskan B, McCoach DB. The performance of RMSEA in models with small degrees of freedom. *Sociological methods & research*. 2015;44(3):486-507.

310. DeCuir-Gunby JT, Marshall PL, McCulloch AW. Developing and using a codebook for the analysis of interview data: An example from a professional development research project. *Field Methods*. 2011;23(2):136-155. doi:10.1177/1525822X10388468

311. Moyce S, Velazquez M, Claudio D, et al. Exploring a rural Latino community's perception of the COVID-19 pandemic. *Ethnicity & Health*. 2021/01/02 2021;26(1):126-138.

doi:10.1080/13557858.2020.1838456

312. Newsome AN, Gilliard T, Phillips A, Dedrick R. Understanding the perceptions of sedentary college students' engagement in physical activity: application of the theory of planned behavior. *Journal of American College Health*. 2021:1-10. doi:10.1080/07448481.2021.1998069

313. Bengtsson M. How to plan and perform a qualitative study using content analysis. *NursingPlus Open*. 2016;2:8-14.

314. Robinson OC. Sampling in interview-based qualitative research: A theoretical and practical guide. *Qualitative research in psychology*. 2014;11(1):25-41.

315. Gucciardi E, Reynolds E, Karam G, Beanlands H, Sidani S, Espin S. Group-based storytelling in disease self-management among people with diabetes. *Chronic Illn*. 2021;17(3):306-320. doi:10.1177/1742395319859395

316. Tracy SJ. Qualitative quality: Eight "big-tent" criteria for excellent qualitative research. *Qualitative inquiry*. 2010;16(10):837-851.

317. Rodgers BL, Cowles KV. The qualitative research audit trail: A complex collection of documentation. *Research in nursing & health*. 1993;16(3):219-226.

318. Kiger ME, Varpio L. Thematic analysis of qualitative data: AMEE Guide No. 131. *Medical Teacher*. 2020/08/02 2020;42(8):846-854. doi:10.1080/0142159X.2020.1755030

Godino JG, Merchant G, Norman GJ, et al. Using social and mobile tools for weight loss in overweight and obese young adults (Project SMART): a 2 year, parallel-group, randomised, controlled trial. *Lancet Diabetes Endocrinol*. Sep 2016;4(9):747-755. doi:10.1016/s2213-8587(16)30105-x
Hayes JF, Balantekin KN, Graham AK, Strube MJ, Bickel WK, Wilfley DE. Implementation intentions for weight loss in college students with overweight and obesity: a proof-of-concept randomized controlled trial. *Transl Behav Med*. Mar 16 2021;11(2):359-368. doi:10.1093/tbm/ibaa038

321. Lua PL, Wan Dali W, Shahril M. Multimodal nutrition education intervention: a cluster randomised controlled trial study on weight gain and physical activity pattern among university students in Terengganu, Malaysia. *Malaysian Journal of Nutrition*. 2013;19(3)

322. St Quinton T, Morris B, Barwood MJ, Conner M. Promoting physical activity through text messages: the impact of attitude and goal priority messages. *Health Psychology and Behavioral Medicine*. 2021/01/01 2021;9(1):165-181. doi:10.1080/21642850.2021.1891073

323. Napolitano MA, Hayes S, Bennett GG, Ives AK, Foster GD. Using facebook and text messaging to deliver a weight loss program to college students. <u>https://doi.org/10.1002/oby.20232</u>. *Obesity*. 2013/01/01 2013;21(1):25-31. doi:<u>https://doi.org/10.1002/oby.20232</u>

324. Napolitano MA, Whiteley JA, Mavredes M, et al. Effect of tailoring on weight loss among young adults receiving digital interventions: an 18 month randomized controlled trial. *Translational Behavioral Medicine*. 2021;11(4):970-980. doi:10.1093/tbm/ibab017

325. Reese JM, Joseph RP, Cherrington A, et al. Development of Participant-Informed Text Messages to Promote Physical Activity Among African American Women Attending College: A Qualitative Mixed-Methods Inquiry. *Journal of Transcultural Nursing*. 2017;28(3):236-242. doi:10.1177/1043659616644959

326. Mbada C, Akinola O, Olumomi G, et al. Effect of structured short message service-based educational programme on physical activity and body weight status among Nigerian female undergraduates. Original Article. *Nigerian Journal of Health Sciences*. January 1, 2018 2018;18(1):37-42. doi:10.4103/njhs.njhs_16_19

327. Munoz LR, La France K, Dominguez D, et al. Text messaging as a tool to increase physical activity in college students. Report. *The Physical Educator*. 2014 Fall

// 2014;71:442+.

328. Stark MA, Elrod K, Haller S, Mahon A. Using technology to increase physical activity in health profession students. *The Open Journal of Occupational Therapy*. 2016;4(1):6.

329. Figueroa CA, Gomez-Pathak L, Khan I, Williams JJ, Lyles CR, Aguilera A. Ratings and experiences in using a mobile application to increase physical activity among university students: implications for future design. *Universal Access in the Information Society*. 2023/01/05 2023;doi:10.1007/s10209-022-00962-z

330. Dillon K, Rollo S, Prapavessis H. A combined health action process approach and mHealth intervention to reduce sedentary behaviour in university students – a randomized controlled trial. *Psychology & Health*. 2022/06/03 2022;37(6):692-711. doi:10.1080/08870446.2021.1900574

331. Prestwich A, Perugini M, Hurling R. Can the effects of implementation intentions on exercise be enhanced using text messages? *Psychol Health*. Jul 2009;24(6):677-87. doi:10.1080/08870440802040715

332. Peng S, Yuan F, Othman AT, Zhou X, Shen G, Liang J. The Effectiveness of E-Health Interventions Promoting Physical Activity and Reducing Sedentary Behavior in College Students: A Systematic Review and Meta-Analysis of Randomized Controlled Trials. *International Journal of Environmental Research and Public Health*. 2023;20(1):318.

333. Thomas K, Müssener U, Linderoth C, Karlsson N, Bendtsen P, Bendtsen M. Effectiveness of a Text Messaging–Based Intervention Targeting Alcohol Consumption Among University Students:

Randomized Controlled Trial. *JMIR Mhealth Uhealth*. 2018/06/25 2018;6(6):e146. doi:10.2196/mhealth.9642

334. Nolan C, Quinn S, MacCobb S. Use of Text Messaging in a Mental Health Service for University Students. *Occupational Therapy in Mental Health*. 2011/04/01 2011;27(2):103-125. doi:10.1080/0164212X.2011.565702

335. Harith S, Backhaus I, Mohbin N, Ngo HT, Khoo S. Effectiveness of digital mental health interventions for university students: an umbrella review. *PeerJ*. 2022;10:e13111.

336. Brown ON, O'Connor LE, Savaiano D. Mobile MyPlate: A Pilot Study Using Text Messaging to Provide Nutrition Education and Promote Better Dietary Choices in College Students. *Journal of American College Health*. 2014/07/04 2014;62(5):320-327. doi:10.1080/07448481.2014.899233

337. Lattie EG, Adkins EC, Winquist N, Stiles-Shields C, Wafford QE, Graham AK. Digital Mental Health Interventions for Depression, Anxiety, and Enhancement of Psychological Well-Being Among College Students: Systematic Review. *J Med Internet Res*. 2019/07/22 2019;21(7):e12869. doi:10.2196/12869

338. Harrer M, Adam SH, Baumeister H, et al. Internet interventions for mental health in university students: A systematic review and meta-analysis. <u>https://doi.org/10.1002/mpr.1759</u>. *International Journal of Methods in Psychiatric Research*. 2019/06/01 2019;28(2):e1759.

doi:https://doi.org/10.1002/mpr.1759

339. Tanaka R, Fujita K, Makimoto K, et al. Development and evaluation of the sedentary behavior and light-intensity physical activity questionnaire. *Journal of Physical Therapy Science*. 2021;33(2):125-131.

340. Garfield BE, Canavan JL, Smith CJ, et al. Stanford seven-day physical activity recall questionnaire in COPD. *European Respiratory Journal*. 2012;40(2):356-362.

341. Rosenberg DE, Norman GJ, Wagner N, Patrick K, Calfas KJ, Sallis JF. Reliability and validity of the Sedentary Behavior Questionnaire (SBQ) for adults. *Journal of physical activity and health*. 2010;7(6):697-705.

342. Hagströmer M, Bergman P, De Bourdeaudhuij I, et al. Concurrent validity of a modified version of the International Physical Activity Questionnaire (IPAQ-A) in European adolescents: The HELENA Study. *Int J Obes (Lond)*. Nov 2008;32 Suppl 5:S42-8. doi:10.1038/ijo.2008.182

343. Hagströmer M, Oja P, Sjöström M. The International Physical Activity Questionnaire (IPAQ): a study of concurrent and construct validity. *Public Health Nutr*. Sep 2006;9(6):755-62. doi:10.1079/phn2005898

344. Cleland C, Ferguson S, Ellis G, Hunter RF. Validity of the International Physical Activity Questionnaire (IPAQ) for assessing moderate-to-vigorous physical activity and sedentary behaviour of older adults in the United Kingdom. *BMC Medical Research Methodology*. 2018/12/22 2018;18(1):176. doi:10.1186/s12874-018-0642-3

345. Lavelle G, Noorkoiv M, Theis N, et al. Validity of the International Physical Activity Questionnaire Short Form (IPAQ-SF) as a measure of physical activity (PA) in young people with cerebral palsy: A crosssectional study. *Physiotherapy*. 2020/06/01/ 2020;107:209-215. doi:https://doi.org/10.1016/j.physio.2019.08.013

346. Lee PH, Macfarlane DJ, Lam TH, Stewart SM. Validity of the International Physical Activity Questionnaire Short Form (IPAQ-SF): a systematic review. *Int J Behav Nutr Phys Act*. Oct 21 2011;8:115. doi:10.1186/1479-5868-8-115

347. Mathes T, Kuss O. A comparison of methods for meta-analysis of a small number of studies with binary outcomes. *Res Synth Methods*. 2018;9(3):366-381. doi:<u>https://doi.org/10.1002/jrsm.1296</u>

348. Eysenck HJ. Meta-analysis and its problems. *Bmj*. Sep 24 1994;309(6957):789-92. doi:10.1136/bmj.309.6957.789

349. Dalton JE, Bolen SD, Mascha EJ. Publication Bias: The Elephant in the Review. *Anesth Analg*. Oct 2016;123(4):812-3. doi:10.1213/ane.00000000001596

350. Fogg B. Fogg behavior model. URL: <u>https://behaviormodel</u> org (visited on 12/14/2020). 2019;

351. Vandelanotte C, Spathonis KM, Eakin EG, Owen N. Website-Delivered Physical Activity Interventions: A Review of the Literature. *American Journal of Preventive Medicine*. 2007/07/01/ 2007;33(1):54-64. doi:<u>https://doi.org/10.1016/j.amepre.2007.02.041</u>

352. Pope L, Harvey-Berino J. Burn and earn: A randomized controlled trial incentivizing exercise during fall semester for college first-year students. *Prev Med*. 2013/03/01/ 2013;56(3):197-201. doi:<u>https://doi.org/10.1016/j.ypmed.2012.12.020</u>

353. Ryan J, Edney S, Maher C. Engagement, compliance and retention with a gamified online social networking physical activity intervention. *Translational Behavioral Medicine*. 2017/12/01 2017;7(4):702-708. doi:10.1007/s13142-017-0499-8

354. Chung C-F, Gorm N, Shklovski IA, Munson S. Finding the Right Fit: Understanding Health Tracking in Workplace Wellness Programs. presented at: Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems; 2017; Denver, Colorado, USA. <u>https://doi.org/10.1145/3025453.3025510</u>

355. Williams SL, French DP. What are the most effective intervention techniques for changing physical activity self-efficacy and physical activity behaviour—and are they the same? *Health Education Research*. 2011;26(2):308-322. doi:10.1093/her/cyr005

356. Mailey EL, Wójcicki TR, Motl RW, et al. Internet-delivered physical activity intervention for college students with mental health disorders: A randomized pilot trial. *Psychology, Health & Medicine*. 2010/12/01 2010;15(6):646-659. doi:10.1080/13548506.2010.498894

357. Burke V, Beilin LJ, Cutt HE, Mansour J, Mori TA. Moderators and mediators of behaviour change in a lifestyle program for treated hypertensives: a randomized controlled trial (ADAPT). *Health Education Research*. 2007;23(4):583-591. doi:10.1093/her/cym047

358. Darker CD, French DP, Eves FF, Sniehotta FF. An intervention to promote walking amongst the general population based on an 'extended' theory of planned behaviour: A waiting list randomised controlled trial. *Psychology & Health*. 2010/01/01 2010;25(1):71-88. doi:10.1080/08870440902893716 359. Dutton GR, Tan F, Provost BC, Sorenson JL, Allen B, Smith D. Relationship between self-efficacy and physical activity among patients with type 2 diabetes. *Journal of Behavioral Medicine*. 2009/06/01 2009;32(3):270-277. doi:10.1007/s10865-009-9200-0

360. Charness G, Gneezy U. Incentives to Exercise. *Econometrica*. 2009;77(3):909-931. doi:<u>https://doi.org/10.3982/ECTA7416</u>

361. Mitchell MS, Goodman JM, Alter DA, et al. Financial Incentives for Exercise Adherence in Adults: Systematic Review and Meta-Analysis. *American Journal of Preventive Medicine*. 2013/11/01/ 2013;45(5):658-667. doi:<u>https://doi.org/10.1016/j.amepre.2013.06.017</u>

362. Barte JCM, Wendel-Vos GCW. A Systematic Review of Financial Incentives for Physical Activity: The Effects on Physical Activity and Related Outcomes. *Behavioral Medicine*. 2017/04/03 2017;43(2):79-90. doi:10.1080/08964289.2015.1074880

363. Strohacker K, Galarraga O, Williams DM. The Impact of Incentives on Exercise Behavior: A Systematic Review of Randomized Controlled Trials. *Annals of Behavioral Medicine*. 2013;48(1):92-99. doi:10.1007/s12160-013-9577-4

364. Hafner M, Pollard J, Van Stolk C. Incentives and Physical Activity: An Assessment of the Association Between Vitality's Active Rewards with Apple Watch Benefit and Sustained Physical Activity Improvements. *Rand Health Q.* Jun 2020;9(1):4.

365. Ball K, Hunter RF, Maple J-L, et al. Can an incentive-based intervention increase physical activity and reduce sitting among adults? the ACHIEVE (Active Choices IncEntiVE) feasibility study. *International Journal of Behavioral Nutrition and Physical Activity*. 2017/03/21 2017;14(1):35. doi:10.1186/s12966-017-0490-2

366. Marteau TM, Ashcroft RE, Oliver A. Using financial incentives to achieve healthy behaviour. *BMJ*. 2009;338:b1415. doi:10.1136/bmj.b1415

367. Rotenstein LS, Berwick DM, Cassel CK. Addressing Well-being Throughout the Health Care Workforce: The Next Imperative. *JAMA*. 2022;328(6):521-522. doi:10.1001/jama.2022.12437

368. Sigurdsson EL. The wellbeing of health care workers. *Scandinavian Journal of Primary Health Care*. 2021/10/02 2021;39(4):389-390. doi:10.1080/02813432.2021.2012352

369. Søvold LE, Naslund JA, Kousoulis AA, et al. Prioritizing the Mental Health and Well-Being of Healthcare Workers: An Urgent Global Public Health Priority. Perspective. *Frontiers in Public Health*. 2021;9

370. Hinrichs LA, Seidler KJ, Morrison K, Coats H. "We're not broken. We're human." A qualitative meta-synthesis of health-care providers' experiences participating in well-being programs. *Physiotherapy Theory and Practice*. 2022:1-26. doi:10.1080/09593985.2022.2103052

371. Melnyk BM, Kelly SA, Stephens J, et al. Interventions to Improve Mental Health, Well-Being, Physical Health, and Lifestyle Behaviors in Physicians and Nurses: A Systematic Review. *American Journal of Health Promotion*. 2020/11/01 2020;34(8):929-941. doi:10.1177/0890117120920451

372. Bischoff LL, Otto A-K, Hold C, Wollesen B. The effect of physical activity interventions on occupational stress for health personnel: A systematic review. *International Journal of Nursing Studies*. 2019/09/01/ 2019;97:94-104. doi:<u>https://doi.org/10.1016/j.ijnurstu.2019.06.002</u>

373. Grimm P. Social desirability bias. *Wiley international encyclopedia of marketing*. 2010;

Appendix 1. 1st DFFC Email for Students

1st DFFC Email - HCOP

We kick off our fifth annual **Deans' Fit Family Challenge** next week!

What is the Deans' Fit Family Challenge? The Deans' Fit Family Challenge is a health and wellness competition between HCOP vs. AUCON vs. VCOM-Auburn. Participants representing each college compete in teams of two. The competition is based on two metrics: cumulative minutes of exercise logged and number of total pounds lost from baseline to endpoint.

What does it cost to participate? Nothing!

What about prizes? Weekly prizes will be awarded to the team with the most exercise minutes logged that week and to a randomly selected participant from HCOP, AUCON, and VCOM-Auburn (i.e., 3 winners per college per week). The overall winning college gets the coveted <u>Deans' Cup</u>. Let's get to moving, HCOP, and win that cup this year! We have not won since Nursing and VCOM-Auburn joined the Challenge.

Why should I participate? We all work hard to do our best each day. But, do we devote enough time to being physically active? The goal of the Challenge is to encourage you to get out there and get active. What better way to relieve today's stress than being physically active?!

How do I get started? Find a partner and think of a fun team name! One member of each team must be a current HCOP employee or student.

What do I need to do next week? Watch for an email with the link to sign up, submit your team name, and complete your weigh-in. Everyone wanting to participate must be signed up by September x, including weighing in. Weigh-ins begin September X and conclude September X. The weigh-in process will consist of you weighing yourself and submitting an image of the scale to a Qualtrics survey. Weigh-outs will be the same process.

Is my weight going to be posted for everyone to see? No way! Weights will be submitted through a Qualtrics link that is only accessible by Dr. B. Fox. He will use weight data to determine the winner and to submit the overall weight loss by the HCOP.

When do I submit exercise data? Exercise/activity logging will begin September x, which is the official kick-off for the challenge. Minutes are due every Sunday by 11:59pm. The Challenge concludes November x.

How do I submit my exercise data? We will use Challenge Runner to <u>automatically record</u> <u>activity</u> using activity trackers. This means that participants with compatible devices do not have to manually submit their data! Challenge Runner is compatible with Android, Apple Health, Samsung Health, Fitbit, Garmin, Google Fit, Misfit, Withings (Nokia), Oura, Suunto, and MapMyFitness. Data can also be submitted manually for those who do not have a compatible device. Stay tuned for more details on the app.

What activities are included in the Challenge? Any intentional, moderate or vigorous physical activity (i.e., that increases your heart rate) counts!

Do I have to lose weight to win? NO! Number of exercise minutes logged is also a huge component of this competition. You could win the individual prize for total minutes logged!

What about the group fitness Village competition? The popular Village competition returns this year. We will have several group fitness activities throughout the Challenge. The Village with the highest participation rate during group fitness activities wins a prize, <u>including points</u> toward the Hargreaves' Day competition.

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research and Policy. Each participant will be compensated with a total of \$15 Amazon gift card for two surveys, and there are chances to win a \$20, a \$30, or \$50 Amazon gift card! This research project is open to all students from HCOP, AUCON, and VCOM-Auburn, regardless of whether you sign up for the DFFC 2022. Additional details will be provided in a future email.

We are so excited for the Challenge this year and we encourage you to participate!

If you have any questions, don't hesitate to email Madi Brown (<u>meb0219@auburn.edu</u>) or Bekcy Woodruff (<u>rlw0067@auburn.edu</u>).

1st DFFC Email - VCOM-Auburn

We kick off our fifth annual **Deans' Fit Family Challenge** next week!

What is the Deans' Fit Family Challenge? The Deans' Fit Family Challenge is an 8-week health and wellness competition between VCOM-Auburn vs. HCOP vs. AUCON. Participants representing each college compete in teams of two. The competition is based on two metrics: cumulative minutes of exercise logged and number of total pounds lost from baseline to endpoint.

What does it cost to participate? Nothing!

What about prizes? Weekly prizes will be awarded to the team with the most exercise minutes logged that week and to a randomly selected participant from VCOM-Auburn, HCOP, and AUCON (i.e., 3 winners per college per week). The overall winning college gets the coveted <u>Deans' Cup</u>. Let's get to moving, VCOM-Auburn, and win that cup this year - again!

Why should I participate? We all work hard to do our best each day. But, do we devote enough time to being physically active? The goal of the Challenge is to encourage you to get out there and get active. What better way to relieve today's stress than being physically active?!

How do I get started? Find a partner and think of a fun team name! One member of each team must be a current VCOM-Auburn employee or student.

What do I need to do this week? Watch for an email with the link to sign up tomorrow (9/15), submit your team name, and complete your weigh-in. Everyone wanting to participate must be signed up by September 17, including weighing in. The weigh-in process will consist of you weighing yourself and submitting an image of the scale to a Qualtrics survey. Weigh-outs will be the same process.

Is my weight going to be posted for everyone to see? No way! Weights will be submitted through a Qualtrics link that is only accessible by Dr. Hollingsworth. He will use weight data to determine the winner and to submit the overall weight loss by VCOM-Auburn.

When do I submit exercise data? Exercise/activity logging will begin September 19, which is the official kick-off for the challenge. Minutes are due every Sunday by 11:59pm. The Challenge concludes November 13.

How do I submit my exercise data? We will use Challenge Runner to <u>automatically record</u> <u>activity</u> using activity trackers. This means that participants with compatible devices do not have to manually submit their data! Challenge Runner is compatible with Apple Health, Fitbit, Garmin, Google Fit Online and Mobile, Misfit, Omron, Oura, Polar, Samsung Health, Strava, Suunto, Under Armour MapMyFitness, and Withings. Data can also be submitted manually for those who do not have a compatible device. Stay tuned for more details on the app.

What activities are included in the Challenge? Any intentional, moderate or vigorous physical activity (i.e., that increases your heart rate) counts!

Do I have to lose weight to win? NO! Number of exercise minutes logged is also a huge component of this competition. You could win the individual prize for total minutes logged!

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research and Policy. Each participant will be compensated with a total of \$15 Amazon gift card for two surveys, and there are chances to win a \$20, a \$30, or \$50 Amazon gift card! This research project is open to all students from HCOP, AUCON, and VCOM-Auburn, regardless of whether you sign up for the DFFC 2022. Additional details will be provided in a future email.

We are so excited for the Challenge this year and we encourage you to participate!

If you have any questions, don't hesitate to email Dr. Hollingsworth (jhollingsworth@auburn.vcom.edu).

Let's go, VCOM-Auburn!

1st DFFC Email -AUCON

We kick off our fifth annual **Deans' Fit Family Challenge** next week!

What is the Deans' Fit Family Challenge? The Deans' Fit Family Challenge is a health and wellness competition between AUCON vs. VCOM-Auburn vs. HCOP. Participants representing each college compete in teams of two. The competition is based on two metrics: cumulative minutes of exercise logged and number of total pounds lost from baseline to endpoint.

What does it cost to participate? Nothing!

What about prizes? Weekly prizes will be awarded to the team with the most exercise minutes logged that week and to a randomly selected participant from AUCON, VCOM-Auburn, and HCOP (i.e., 3 winners per college per week). The overall winning college gets the coveted <u>Deans' Cup</u>. Let's get to moving, AUCON, and win that cup this year - again!

Why should I participate? We all work hard to do our best each day. But, do we devote enough time to being physically active? The goal of the Challenge is to encourage you to get out there and get active. What better way to relieve today's stress than being physically active?!

How do I get started? Find a partner and think of a fun team name! One member of each team must be a current AUCON employee or student.

What do I need to do next week? Watch for an email with the link to sign up, submit your team name, and complete your weigh-in. Everyone wanting to participate must be signed up by September x, including weighing in. Weigh-ins begin September X and conclude September X. The weigh-in process will consist of you weighing yourself and submitting an image of the scale to a Qualtrics survey. Weigh-outs will be the same process.

Is my weight going to be posted for everyone to see? No way! Weights will be submitted through a Qualtrics link that is only accessible by Dr. Robin Gosdin Farrell. She will use weight data to determine the winner and to submit the overall weight loss by AUCON.

When do I submit exercise data? Exercise/activity logging will begin September x, which is the official kick-off for the challenge. Minutes are due every Sunday by 11:59pm. The Challenge concludes November x.

How do I submit my exercise data? We will use Challenge Runner to <u>automatically record</u> <u>activity</u> using activity trackers. This means that participants with compatible devices do not have to manually submit their data! Challenge Runner is compatible with Android, Apple Health, Samsung Health, Fitbit, Garmin, Google Fit, Misfit, Withings (Nokia), Oura, Suunto, and MapMyFitness. Data can also be submitted manually for those who do not have a compatible device. Stay tuned for more details on the app.

What activities are included in the Challenge? Any intentional, moderate or vigorous physical activity (i.e., that increases your heart rate) counts!

Do I have to lose weight to win? NO! Number of exercise minutes logged is also a huge component of this competition. You could win the individual prize for total minutes logged!

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research and Policy. Each participant will be compensated with a total of \$15 Amazon gift card for two surveys, and there are chances to win a \$20, a \$30, or \$50 Amazon gift card! This research project is open to all students from HCOP, AUCON, and VCOM-Auburn, regardless of whether you sign up for the DFFC 2022. Additional details will be provided in a future email.

We are so excited for the Challenge this year and we encourage you to participate!

If you have any questions, don't hesitate to email Dr. Farrell (rgf0001@auburn.edu).

Appendix 2. 2nd DFFC Email for Students

2nd DFFC Email – HCOP and AUCON

It's time!!!

Sign-up for the Deans' Fit Family Challenge is open and closes September 17th at 10pm. The link is below.

This is a <u>free</u> 8-week weight loss and exercise-minutes challenge between HCOP vs. AUCON vs. VCOM-Auburn with <u>weekly prizes</u>. Click here for more information: <u>https://drive.google.com/file/d/1fl0psRGV07GaTW08L58i7vmCGoEmpUOn/view?usp=sharing</u>

To sign up, you will need:

- A teammate. Your teammate does not have to be affiliated with the HCOP.
- A team name. Be creative but professional.
- Your weight, captured in a photo with your feet standing on the scale. This will be submitted using the link below.
 - Use the scale of your choice. Must use same scale when weighing out.

Weigh-in information is submitted individually, so teammates register/weigh-in separately.

It's easiest to register for the Challenge on your phone (to submit the weight photo).

Go here to weigh-in and register for the

Challenge: https://auburn.qualtrics.com/jfe/form/SV_9WBG3dKnmmkRMaO

Once registration closes, an email will be sent with instructions to submit physical activity minutes via Challenge Runner.

• This email will also contain the link to register for the optional research study mentioned in our first email, provided below.

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research

and Policy. Each participant will be compensated with a total of \$15 Amazon gift card for two surveys, and there are chances to win a \$20, a \$30, or \$50 Amazon gift card! This research project is open to all students from HCOP, AUCON, and VCOM-Auburn, regardless of whether you sign up for the DFFC 2022.

If you are **NOT** participating in the Challenge but would like to participate in the research study, click <u>https://auburn.qualtrics.com/jfe/form/SV_0PTxF2SPgFXP0rQ</u>.

If you ARE participating in the Deans' Fit Family Challenge and would like to also participate in the research study, we will send another email with the study link after you register for the Challenge. Stay tuned!

2nd DFFC Email – VCOM-Auburn

Sign-up for the Deans' Fit Family Challenge **is open and closes at 11:59 PM on Saturday September** 17th. The sign-up link is below.

This is a **free** 8-week weight loss and exercise-minutes challenge between HCOP vs. AUCON vs. VCOM-Auburn with **weekly prizes**. <u>Click here for more info</u>.

To sign up, you will need:

- A teammate. Your teammate does not have to be affiliated with VCOM.
- A team name. Be creative but professional.
- Your weight, captured in a photo with your feet standing on the scale. This will be submitted using the link below.
 - Use the scale of your choice. You must use same scale when weighing out in 8 weeks.

Weigh-in information is submitted individually, so teammates register and weigh-in separately.

It's easiest to register for the Challenge on your phone (to submit the weight photo).

Go here to weigh-in and register for the Challenge:

https://auburn.qualtrics.com/jfe/form/SV_9WBG3dKnmmkRMaO

Once registration closes, an email will be sent to participants with instructions to submit physical activity minutes via ChallengeRunner. This email will also contain the link to register for the optional research study mentioned in our first email and described below.

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research and Policy. Compensation will be pro-rated, with \$5 for completion and submission of the first survey and \$10 after completion and submission of the second survey. Individuals who have completed both surveys will be entered into a drawing for one of three prizes - a \$20, a \$30, or a \$50 Amazon gift card. The chances of winning one of these prizes is 1/500. This research project is open to all students from HCOP, AUCON, and VCOM-Auburn, regardless of whether you sign up for the Challenge.

If you are **NOT** participating in the Deans' Fit Family Challenge **but would like to participate in the** research study, click here: https://auburn.qualtrics.com/jfe/form/SV_0PTxF2SPgFXP0rO

If you **ARE** participating in the Deans' Fit Family Challenge **and would like to also participate in the research study**, we will send another email with the study link after you register for the Challenge. Stay tuned!

If you have any questions, please contact Dr. Hollingsworth by replying to this email.

Let's go VCOM, Let's go! 🕅

Appendix 3. Sign-up Surveys for Control Group

Q7 Select your academic affiliation. If you are not affiliated with these programs, please indicate your teammate's affiliation.

 \bigcirc HCOP (1)

 \bigcirc AUCON (2)

 \bigcirc VCOM-Auburn (3)

Display This Question:

If Select your academic affiliation. If you are not affiliated with these programs, please indicate... = HCOP Or Select your academic affiliation. If you are not affiliated with these programs, please indicate... = AUCON

Q9

(NOTE: DO NOT SIGN THIS DOCUMENT UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.)

INFORMATION LETTER for a Research Study entitled

"The effect of text message and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing"

SUMMARY: The purpose of this study is to assess and compare the effectiveness of competitionbased and text message interventions in improving physical activity among university students and evaluate participants' acceptability and experiences. Participants need to download the "Challenge Runner" app to automatically submit their weekly physical activity data. Participants will be asked to answer two anonymous online surveys about their physical activity levels, self-efficacy, selfregulation, social support, incentive motivation, and mental wellbeing. Each participant will be compensated with one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. Participants will be entered into a drawing for a chance to win a \$20, a \$30, or \$50 Amazon gift card after completing both surveys. Participants' emails will be discarded once we complete data collection and incentive delivery. No identifiable information will be linked to survey responses.

You are invited to participate in a research study to assess and compare the effectiveness of competition-based and text message interventions on physical activity among university students and to evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Auburn University Department of Health Outcomes Research and Policy. You are invited to participate because you are an Auburn University student and are not participating in the Deans' Fit Family Challenge.

What will be involved if you participate? Your participation is completely voluntary. To be eligible to

participate, you should be age 18 or older, and you will need a smartphone and a fitness tracker or smartwatch. If you decide to participate in this research study, you will be asked to download the "Challenge Runner" app on your smartphone and connect this app to your fitness tracker or smartwatch to automatically submit your weekly physical activity minutes (or you can submit weekly activity minutes on the Challenge Runner website). You will also be asked to answer two online surveys (will be sent in September and November). Both surveys will take approximately 20 minutes to complete. Your total time commitment for surveys will be approximately 45 minutes.

Are there any risks or discomforts? Breach of confidential information is possible. To minimize these risks, we will store your weekly activity minutes recordings and survey responses in a secured server; furthermore, your contact information will be kept on a password-protected computer, separate from questionnaire data. Participants' emails will be discarded once data collection is complete and incentives have been delivered. For the purposes of this research project, no identifiable information will be linked to survey responses or physical activity data. You may skip any items you choose not to answer because of potential discomfort answering questions of a personal nature.

Are there any benefits to yourself or others? There are no direct benefits associated with participating in this project. The study findings will inform future studies regarding physical activity interventions. We cannot promise you that will receive any or all of the benefits described.

Will you receive compensation for participating? To thank you for your time you will be offered one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. You will be entered into a drawing for a chance to win a \$20, \$30, or \$50 Amazon gift card at the conclusion of the DFFC.

Are there any costs? There will be no cost to participate in this study. The Challenge Runner app is free to download. You will be responsible for your own expenses if you experience any injury during the study period.

If you change your mind about participating, you can withdraw at any time during the study. Your participation is completely voluntary. You can withdraw at any time by emailing Chenyu Zou (czz0065@auburn.edu) or Dr. Brent I. Fox (foxbren@auburn.edu). Once submitted, data cannot be withdrawn. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Health Outcomes Research and Policy, or the study organizers.

Your privacy will be protected. Your identity will remain confidential. Information obtained through your participation may be used to fulfill an educational requirement, published in a professional journal, or presented at a professional meeting.

If you have questions about this study, please contact Chenyu Zou at <u>czz0063@auburn.edu</u> or 513-206-2455. This letter was approved by the VCOM and Auburn University Institutional Review Boards for use starting 9/12/22 as protocol 22-380 EX 2209.

If you have questions about your rights as a research participant, and you are an AU student, you may contact the Auburn University Office of Research Compliance or the Institutional Review Board by phone (334) 844-5966 or e-mail at IRBadmin@auburn.edu or IRBChair@auburn.edu. If you are a

VCOM student, please contact the VCOM IRB Chair, Dr. Gunnar Brolinson at pbrolinson@vcom.vt.edu or (540) 231-4981.

HAVING READ THE INFORMATION ABOVE, YOU MUST DECIDE IF YOU WANT TO PARTICIPATE IN THIS RESEARCH PROJECT. IF YOU DECIDE TO PARTICIPATE, PLEASE SELECT "YES" TO THE QUESTION BELOW. YOUR COMPLETION OF THIS ONLINE SURVEY INDICATES YOUR WILLINGNESS TO PARTICIPATE IN THIS PROJECT. YOU MAY PRINT A COPY OF THIS LETTER TO KEEP.

Display This Question:

If Select your academic affiliation. If you are not affiliated with these programs, please indicate... = VCOM-Auburn

Q8

(NOTE: DO NOT SIGN THIS DOCUMENT UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.)

INFORMATION LETTER for a Research Study entitled

"Effect of text message and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing"

SUMMARY: The purpose of this study is to assess and compare the effectiveness of competitionbased and text message interventions in improving physical activity among university students and evaluate participants' acceptability and experiences. Participants need to download the "Challenge Runner" app to automatically submit their weekly physical activity data. Participants will be asked to answer two anonymous online surveys about their physical activity levels, self-efficacy, selfregulation, social support, incentive motivation, and mental wellbeing. All participants will be compensated with one \$5 Amazon gift card after completing the first survey and another \$10 Amazon gift card after completing the second survey. Participants will be entered into a drawing for a \$20, a \$30, or \$50 Amazon gift card after completing both surveys, with a 1 in 500 chance of winning one of these prizes. Participants' emails will be discarded once we complete data collection and incentive delivery. No identifiable information will be linked to survey responses.

You are invited to participate in a research study to assess and compare the effectiveness of competition-based and text message interventions on physical activity among university students and to evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Auburn University Department of Health Outcomes Research and Policy. You are invited to participate because you are an Auburn University student and are not participating in the Deans' Fit Family Challenge.

What will be involved if you participate? Your participation is completely voluntary. To be eligible to participate, you should be age 18 or older, and you will need a smartphone and a fitness tracker or smartwatch. If you decide to participate in this research study, you will be asked to download the "Challenge Runner" app on your smartphone and connect this app to your fitness tracker or smartwatch to automatically submit your weekly physical activity minutes (or you can submit weekly activity minutes on the Challenge Runner website). You will also be asked to answer two online surveys (will be sent in September and November). Both surveys will take approximately 20 minutes to complete. Your total time commitment for surveys will be approximately 45 minutes.

Are there any risks or discomforts? Breach of confidential information is possible. To minimize these risks, we will store your weekly activity minutes recordings and survey responses in a secured server; furthermore, your contact information will be kept on a password-protected computer, separate from questionnaire data. Participants' emails will be discarded once data collection is complete and incentives have been delivered. For the purposes of this research project, no identifiable information will be linked to survey responses or physical activity data. You may skip any items you choose not to answer because of potential discomfort answering questions of a personal nature.

Are there any benefits to yourself or others? There are no direct benefits associated with participating in this project. The study findings will inform future studies regarding physical activity interventions. We cannot promise you that will receive any or all of the benefits described.

Will you receive compensation for participating? To thank you for your time, you will be offered one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. After completing both surveys, you will be entered into a drawing for a \$20, a \$30, or \$50 Amazon gift card, with a 1 in 500 chance of winning one of these prizes.

Are there any costs? There will be no cost to participate in this study. The Challenge Runner app is free to download. You will be responsible for your own expenses if you experience any injury during the study period.

If you change your mind about participating, you can withdraw at any time during the study. Your participation is completely voluntary. You can withdraw at any time by emailing Chenyu Zou (czz0065@auburn.edu) or Dr. Brent I. Fox (foxbren@auburn.edu). Once submitted, data cannot be withdrawn. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Health Outcomes Research and Policy, or the study organizers.

Your privacy will be protected. Your identity will remain confidential. Information obtained through your participation may be used to fulfill an educational requirement, published in a professional journal, or presented at a professional meeting.

If you have questions about this study, please contact Chenyu Zou at czz0063@auburn.edu or 513-206-2455. This letter was approved by the VCOM Institutional Review Board for use starting 9/12/22 as protocol 2022-061.

158

If you have questions about your rights as a research participant, please contact the VCOM IRB Chair, Dr. Gunnar Brolinson at pbrolinson@vcom.vt.edu or (540) 231-4981.

HAVING READ THE INFORMATION ABOVE, YOU MUST DECIDE IF YOU WANT TO PARTICIPATE IN THIS RESEARCH PROJECT.IF YOU DECIDE TO PARTICIPATE, PLEASE SELECT "YES" TO THE QUESTION BELOW. YOUR COMPLETION OF THIS ONLINE SURVEY INDICATES YOUR WILLINGNESS TO PARTICIPATE IN THIS PROJECT. YOU MAY PRINT A COPY OF THIS LETTER TO KEEP.

This research has been reviewed and approved by the VCOM Institutional Review board. (Record #2022-061)

Q2 Do you agree to participate in this project?

 \bigcirc YES (1)

O NO (2)

Skip To: End of Survey If Do you agree to participate in this project? = NO

End of Block: Information Letter

Start of Block: Inclusion Criteria

Q3 Are you at least 18 years of age?

 \bigcirc YES (1)

O NO (2)

Skip To: End of Survey If Are you at least 18 years of age ? = NO

Q8 Do you have any health issues that prevent you from being physically active, including but not limited to pregnancy, disabilities, and chronic diseases?

YES (1)NO (2)

Skip To: End of Survey If Do you have any health issues that prevent you from being physically active, including but not li... = YES

Q5 To be eligible to participate, you will need a smartphone and a fitness tracker or smartwatch. Please indicate which app you would like to use to synchronize activity minute to the Challenge Runner app.

O Apple Health (1)
O Fitbit (2)
O Garmin (3)
O Google Fit (online and mobile) (4)
Omron (5)
Oura (6)
O Polar (7)
O Samsung Health (8)
O Strava (9)
O Suunto (10)
O Under Armour MapMyFitness (11)
O Withings (12)
\bigcirc I don't have a fitness tracker or smartwatch, or my devices do not support any of these apps (13)

Skip To: End of Survey If To be eligible to participate, you will need a smartphone and a fitness tracker or smartwatch. PI... = I don't have a fitness tracker or smartwatch, or my devices do not support any of these apps

Q6 Please provide your AU/VCOM email address. Please watch this email. We will send important information and links for your participation in this project. Thank you for your participation!

End of Block: Inclusion Criteria

Appendix 4. 3rd DFFC Email for Students

<Only those students who signed up for the Challenge will receive this email.>

Your registration for the Challenge is almost complete.

The next step is to create your Challenge Runner account, and to download the Challenge Runner app.

The Challenge Runner instructions are attached.

As previously mentioned, you can sign up for the optional research study, described below.

Students are also invited to participate in a research study to assess and compare the effectiveness of text message interventions on improving physical activity among university students and evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Department of Health Outcomes Research and Policy. Each participant will be compensated with a total of \$15 Amazon gift card for two surveys, and there are chances to win a \$20, \$30, or \$50 Amazon gift card!

Click here to sign up for the research project.

Appendix 5. Sign-up Surveys for DFFC and DFFC + Text Group

Q8 Select your academic affiliation. If you are not affiliated with these programs, please indicate your teammate's affiliation.

 \bigcirc HCOP (1)

 \bigcirc AUCON (2)

 \bigcirc VCOM-Auburn (3)

Display This Question:

If Select your academic affiliation. If you are not affiliated with these programs, please indicate... = HCOP Or Select your academic affiliation. If you are not affiliated with these programs, please indicate... = AUCON

Q13

(NOTE: DO NOT SIGN THIS DOCUMENT UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.)

INFORMATION LETTER for a Research Study entitled

"The effect of text message and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing"

SUMMARY: The purpose of this study is to assess and compare the effectiveness of competitionbased and text messaging interventions in improving physical activity among health professional students and evaluate participants' acceptability and experiences. We will collect weekly physical activity data via the Challenge Runner app. Participants will be asked to answer two anonymous online surveys about their physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing. Each participant will be compensated with one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. Participants will be entered into a drawing for a chance to win a \$20, a \$30, or \$50 Amazon gift card after completing both surveys. Participants' emails will be discarded once we complete data collection and incentive delivery. No identifiable information will be linked to survey

responses.

You are invited to participate in a research study to assess and compare the effectiveness of competition-based and text messaging interventions on physical activity among health professional students and to evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD,
in the Auburn University Department of Health Outcomes Research and Policy. You are invited to participate because you are enrolled in HCOP, AUCON, or VCOM-Auburn.

What will be involved if you participate? Your participation is completely voluntary. To be eligible to participate, you should be age 18 or older and willing to receive text messages (at your expense, if it is not free). You will need a smartphone and a fitness tracker or smartwatch. If you decide to participate in this research study, your physical activity minutes collected for the 2022 Deans' Fit Family Challenge (DFFC) will be used in this research study. In addition, you may receive weekly text messages. You will be asked to answer two online surveys (will be sent in September and November). Both surveys will take approximately 20 minutes to complete. Your total time commitment for surveys will be approximately 45 minutes.

Are there any risks or discomforts? Breach of confidential information is possible. To minimize these risks, we will store weekly physical activity minutes recordings and survey responses in a secured server; furthermore, your contact information will be kept on a password-protected computer, separate from questionnaire data. Participants' phone numbers and emails will be discarded once data collection is complete and incentives have been delivered. For the purposes of this research project, no identifiable information will be linked to survey responses or physical activity data. You may skip any items you choose not to answer because of potential discomfort answering questions of a personal nature.

Are there any benefits to yourself or others? There are no direct benefits associated with participating in this project. The study findings will inform future studies regarding physical activity interventions. You may find it enjoyable to receive supportive messages and share your experience with us. We cannot promise that you will receive any or all of the benefits described.

Will you receive compensation for participating? To thank you for your time, you will be offered one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. You will be entered into a drawing for a chance to win a \$20, \$30, or \$50 Amazon gift card at the conclusion of the DFFC.

Are there any costs? There will be no cost to participate in this study. The Challenge Runner app is free to download. Your decision regarding participation in this study will NOT affect your competition in 2022 Deans' Fit Family Challenge. You will be responsible for any carrier fees for text messages, if applicable. You will be responsible for your own expenses, if you experience any injury during the DFFC.

If you change your mind about participating, you can withdraw at any time during the study. Your participation is completely voluntary. To remain in the Deans' Fit Family Challenge but withdraw from the study, please email Chenyu Zou (czz0063@auburn.edu) or Dr. Brent I. Fox (foxbren@auburn.edu). Once submitted, data cannot be withdrawn. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Health Outcomes Research and Policy, or the study organizers.

Your privacy will be protected. Your identity will remain confidential. Information obtained through your participation may be used to fulfill an educational requirement, published in a professional journal, or presented at a professional meeting.

If you have questions about this study, please contact Chenyu Zou at czz0063@auburn.edu or 513-206-2455. This letter was approved by the VCOM and Auburn University Institutional Review Boards for use starting 9/18/22 as protocol 22-380 EX 2209.

If you have questions about your rights as a research participant, and you are an AU student, you may contact the Auburn University Office of Research Compliance or the Institutional Review Board by phone (334) 844-5966 or e-mail at <u>IRBadmin@auburn.edu</u> or <u>IRBChair@auburn.edu</u>. If you are a **VCOM student**, please contact the VCOM IRB Chair, Dr. Gunnar Brolinson at pbrolinson@vcom.vt.edu or (540) 231-4981.

HAVING READ THE INFORMATION ABOVE, YOU MUST DECIDE IF YOU WANT TO PARTICIPATE IN THIS RESEARCH PROJECT. IF YOU DECIDE TO PARTICIPATE, PLEASE SELECT "YES" TO THE QUESTION BELOW. YOUR COMPLETION OF THIS ONLINE SURVEY INDICATES YOUR WILLINGNESS TO PARTICIPATE IN THIS PROJECT. YOU MAY PRINT A COPY OF THIS LETTER TO KEEP.

Display This Question:

If Select your academic affiliation. If you are not affiliated with these programs, please indicate... = VCOM-Auburn

Q9

(NOTE: DO NOT SIGN THIS DOCUMENT UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.)

INFORMATION LETTER for a Research Study entitled

"Effect of text message and competition-based interventions on university students' physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing"

SUMMARY: The purpose of this study is to assess and compare the effectiveness of competitionbased and text messaging interventions in improving physical activity among health professional students and evaluate participants' acceptability and experiences. We will collect weekly physical activity data via the Challenge Runner app. Participants will be asked to answer two anonymous online surveys about their physical activity levels, self-efficacy, self-regulation, social support, incentive motivation, and mental wellbeing. All participants will be compensated with one \$5 Amazon gift card after completing the first survey and another \$10 Amazon gift card after completing the second survey. Participants will be entered into a drawing for a \$20, a \$30, or \$50 Amazon gift card after completing both surveys, with a 1 in 500 chance of winning one of these prizes. Participants' emails will be discarded once we complete data collection and incentive

delivery. No identifiable information will be linked to survey responses.

Yo

u are invited to participate in a research study to assess and compare the effectiveness of competitionbased and text messaging interventions on physical activity among health professional students and to evaluate participants' acceptability and experiences. The study is being conducted by Chenyu Zou, MS, as her doctorate dissertation project under the direction of Brent I. Fox, PharmD, PhD, in the Auburn University Department of Health Outcomes Research and Policy. You are invited to participate because you are enrolled in HCOP, AUCON, or VCOM-Auburn.

What will be involved if you participate? Your participation is completely voluntary. To be eligible to participate, you should be age 18 or older and willing to receive text messages (at your expense, if it is not free). You will need a smartphone and a fitness tracker or smartwatch. If you decide to participate in this research study, your physical activity minutes collected for the 2022 Deans' Fit Family Challenge (DFFC) will be used in this research study. In addition, you may receive weekly text messages. You will be asked to answer two online surveys (will be sent in September and November). Both surveys will take approximately 20 minutes to complete. Your total time commitment for surveys will be approximately 45 minutes.

Are there any risks or discomforts? Breach of confidential information is possible. To minimize these risks, we will store weekly physical activity minutes recordings and survey responses in a secured server; furthermore, your contact information will be kept on a password-protected computer, separate from questionnaire data. Participants' phone numbers and emails will be discarded once data collection is complete and incentives have been delivered. For the purposes of this research project, no identifiable information will be linked to survey responses or physical activity data. You may skip any items you choose not to answer because of potential discomfort answering questions of a personal nature. There are inherent risks of physical injury or discomfort from moderate or vigorous exercise. However, the exercises/activities you will be performing are part of the Dean's Fit Family Challenge, and the research is not requiring you to do any additional exercise beyond what is included in that program. Neither the researchers nor their institutions have funds set aside should you require medical intervention following an injury, and you or your insurer would be responsible for any associated costs of that care.

Are there any benefits to yourself or others? There are no direct benefits associated with participating in this project. The study findings will inform future studies regarding physical activity interventions. You may find it enjoyable to receive supportive messages and share your experience with us. We cannot promise that you will receive any or all of the benefits described.

Will you receive compensation for participating? To thank you for your time, you will be offered one \$5 Amazon gift card after completing the first survey, and another \$10 Amazon gift card after completing the second survey. After completing both surveys, you will be entered into a drawing for a \$20, a \$30, or \$50 Amazon gift card, with a 1 in 500 chance of winning one of these prizes.

Are there any costs? There will be no cost to participate in this study. The Challenge Runner app is free to download. Your decision regarding participation in this study will NOT affect your competition in 2022 Deans' Fit Family Challenge. You will be responsible for any carrier fees for text messages, if applicable. You will be responsible for your own expenses, if you experience any injury during the DFFC.

If you change your mind about participating, you can withdraw at any time during the study. Your participation is completely voluntary. To remain in the Deans' Fit Family Challenge but withdraw from the study, please email Chenyu Zou (czz0063@auburn.edu) or Dr. Brent I. Fox (foxbren@auburn.edu). Once submitted, data cannot be withdrawn. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Health Outcomes Research and Policy, or the study organizers.

Your privacy will be protected. Your identity will remain confidential. Information obtained through your participation may be used to fulfill an educational requirement, published in a professional journal, or presented at a professional meeting.

If you have questions about this study, please contact Chenyu Zou at czz0063@auburn.edu or 513-206-2455. This letter was approved by the VCOM Institutional Review Board for use starting 9/12/22 as protocol 2022-

061.

If you have questions about your rights as a research participant, please contact the VCOM IRB Chair, Dr. Gunnar Brolinson at pbrolinson@vcom.vt.edu or (540) 231-4981.

HAVING READ THE INFORMATION ABOVE, YOU MUST DECIDE IF YOU WANT TO PARTICIPATE IN THIS RESEARCH PROJECT. IF YOU DECIDE TO PARTICIPATE, PLEASE SELECT "YES" TO THE QUESTION BELOW. YOUR COMPLETION OF THIS ONLINE SURVEY INDICATES YOUR WILLINGNESS TO PARTICIPATE IN THIS PROJECT. YOU MAY PRINT A COPY OF THIS LETTER TO KEEP.

This research has been reviewed and approved by the VCOM Institutional Review board. (Record #2022-061)

Q2 Do you agree to participate in this project?

 \bigcirc YES (1)

O NO (2)

Skip To: End of Survey If Do you agree to participate in this project? = NO

End of Block: Information Letter

Start of Block: Inclusion Criteria

Q3 Are you at least 18 years of age?

YES (1)NO (2)

Skip To: End of Survey If Are you at least 18 years of age ? = NO

Q9 Do you have any health issues that prevent you from being physically active, including but not limited to pregnancy, disabilities, and chronic disease?

 \bigcirc Yes (2)

O No (9)

Skip To: End of Survey If Do you have any health issues that prevent you from being physically active, including but not li... = Yes

Q8 The physical activity minutes you submit for the DFFC will be used for this study. How do you plan to submit your physical activity minutes?

 \bigcirc I do not possess a compatible fitness tracker, smartwatch, or smartphone. I will submit my minutes to ChallengeRunner.com manually. (1)

 \bigcirc I will submit my minutes to ChallengeRunner automatically through my fitness tracker or smartphone app. (2)

Skip To: End of Survey If The physical activity minutes you submit for the DFFC will be used for this study. How do you pla... = I do not possess a compatible fitness tracker, smartwatch, or smartphone. I will submit my minutes to ChallengeRunner.com manually.

Q7 Please provide a cellphone number for you. We may use this number to send text messages about physical activity during this study.

Q6 Please provide your AU or VCOM email address. Please watch this email. We will send important information and links for your participation in this project. Thank you for your participation!

End of Block: Inclusion Criteria

	Monday	Wednesday	Friday
	Self-efficacy	Social support	Self-regulation and
			incentive motivation
WEEK- 2 WEEK- 2	The first step is often the	Thank you for agreeing to participate in this research project affiliated with the Deans' Challenge! P You'll receive a supportive text Mon, Wed, and Fri for the next 7 weeks. Now's the time to work on increasing your physical activity. You've got this! September 28, 2022, 12.40pm If working out isn't working for you, try a fun activity with friends. Throw a fricker	Happy Friday! Exercise minutes will be collected this Sunday. Exercise at the same time and place every day to make it a habit. Cheers to health! September 30, 2022, 1.21pm Don't forget to track your
3	nardest. Focus on getting out there and getting started. Consistency is key. You've got this!	have a dance party, or go for bike ride or walk. I'll make you feel so much better to be active!	And when reflecting on your performance, remember: Progress over perfection. You don't have to be perfect. Just keep moving in the right direction Σ
	October 3, 2022, 11.30am	October 5, 2022, 11.12 am	October 7, 2022, 9.15 am
4	Look for small ways to increase your physical activity. Park in the back of the lot, take the stairs, pace when on the phone, and add short walks to your day. Small sustainable changes lead to big results!	increase their activity? Consider being partners in progress. Support and encourage each other. Be better together!	no worries. Don't beat yourself up. Instead of viewing it as a "fail," view it as FAIL: "First Attempts. I'm Learning." Take the lesson and focus forward.
	October 10, 2022, 11 am	October 12, 2022, 11am	October 14, 2022, 7pm
WEEK- 5	Spending a lot of time on the computer today? Consider going for a short walk. Five minutes is better than none, and more is generally better. Focus small and do more when you're motivated and able!	Your friends can help keep you stay motivated to reach your goal! S Reach out to friends today and do some exercise or activity y'all enjoy!	It's time to ask yourself: How'd I do this week? No matter how slow you go, you're still lapping everyone on the couch! Don't forget to track your minutes this week!
	October 17, 2022, 4.30pm	October 19, 2022, 11.20am	October 21, 2022, 1pm
WEEK- 6	If you hit any barriers to increasing your physical activity, don't give up! Ask yourself, "What's making this hard to do?" and then, "How can I	Got a break between classes? Go for a walk w friends around campus. Got a longer break? Explore a new park or hiking trail! <u>https://bit.ly/WalkAuburnAL</u>	It's almost the weekend. To celebrate, go for a nice, long, 40-minute walk tomorrow. Don't forget to track your minutes this week!

Appendix 6. 20 Texts for DFFC + Text Group

	make it easier?" You've		
	got this! 😈		
	October 24, 2022, 11am	October 26, 2022, 2pm	October 28, 2022, 2pm
WEEK- 7	Breaking up your sitting time can reduce your risk for certain types of cancer. You have more control over your health than you think!	Even if you're glued to your phone, you don't have to be glued to your seat! Go and meet up with your friends. Working out with a group can be fun and motivating!	Here is another reason to sit less: taking a break from sitting to walk around or doing some light stretching can help strengthen your bones. Reminder to track your minutes if you have not
	10 21 2022 11:20 AM	11 02 2022 11:20 AM	Nevershar 4, 2022
	10-31-2022 11:29 AM	11-02-2022 11:29 AM	November 4, 2022, 2.30pm
WEEK- 8	How's your exercise plan going? Make sure to get some walking in today and avoid sitting for too long. You got this!	Let's replace an hour of sitting time with walking or exercising with your friends! Do something fun!	Hopefully these goals have become habits by now, and if not, just keep practicing them! "Your health is worth the effort. Don't forget to track your minute this week!
	11-07-2022 11:30 AM	11-09-2022 11:30 AM	11-11-2022 02:00 PM

Appendix 7. Week-0 Survey for All Three Groups

Start of Block: Introduce

Q16 Thank you for participating in this research project. Please answer the following questions. The survey takes approximately 20 minutes to complete. Upon completion of this survey, you will be given the option to provide your contact information for a \$5 Amazon gift card, to be delivered to your AU/VCOM email.

End of Block: Introduce

Start of Block: Physical activity levels

Q5 On average, how many days per week do you engage in moderate to vigorous physical activity (like a brisk walk)? You are encouraged to look at the data collected by your fitness tracker/smartphone.

Mod	erate Intensity Activities	Vigorous Intensity Activities				
• `	Walking briskly (2.5 miles per hour or faster)	•	Jogging or running			
•	Recreational swimming	•	Swimming laps			
•	Bicycling slower than 10 miles per hour on level	•	Tennis (singles)			
1	terrain	•	Vigorous dancing			
• ·	Tennis (doubles)	•	Bicycling faster than 10 miles per hour			
• /	Active forms of yoga (for example, Vinyasa or	•	Jumping rope			
1	power yoga)	•	Heavy yard work (digging or shoveling, with heart			
•	Ballroom or line dancing		rate increases)			
• (General yard work and home repair work	•	Hiking uphill or with a heavy backpack			
•	Exercise classes like water aerobics	•	High-intensity interval training			
		•	Exercise classes like vigorous step aerobics or			
			kickboxing			

▼ 0 ... 7

Q20 On average, how many minutes do you engage in physical activity at this level? You are encouraged to look at the data collected by your fitness tracker/smartphone.

Q17 How many days a week do you perform muscle strengthening exercises? A strength exercise involves using your body weight or working against a resistance, such as lifting weights, working with resistance bands, climbing stairs, hill walking, cycling, dancing, push-ups, sit-ups and squats, and yoga.

▼ 0....7

Q11 Are you currently participating or will you participate in any other organized program to increase your physical activity during the study period (Sep. X - Nov. X)? If yes, pleas briefly describe the program.

• Yes. Please briefly describe. (1)

O No (2)

End of Block: Physical activity levels

Start of Block: Mental wellbeing

Q6 Over the last two weeks:

	At no time (1)	Some of the time (2)	More than half of the time (3)	Most of the time (4)	All the time (5)
I have felt cheerful and in good spirits (1)	0	\bigcirc	0	\bigcirc	\bigcirc
I have felt calm and relaxed (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I have felt active and vigorous (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I woke up feeling fresh and rested (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
My daily life has been filled with things that interest me (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
End of Block: Mental wellbeing					

Start of Block: Self-Efficacy

Q7 According to Physical Activity Guidelines for Americans, adults should perform at least 150 minutes a week of moderate-intensity, or 75 minutes a week of vigorous-intensity aerobic physical activity, or an equivalent combination of moderate- and vigorous-intensity activity (1 minute of vigorous activity is equal to 2 minutes of moderate activity). In addition, strengthening exercises on 2 or more days a week,

	Not at all confident (1)	Slightly confident (2)	Moderately confident (3)	Very confident (4)	Extremely confident (5)			
The weather was bothering you (1)	0	0	\bigcirc	\bigcirc	0			
You were bored by the program or activity (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You felt pain when exercising (3)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You had to exercise alone (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You did not enjoy it (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You were too busy with other activities (6)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You felt tired (7)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You felt stressed (8)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
You felt depressed (9)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc			
End of Block: Self-Efficacy								

such as bodyweight exercises or resistance training, are recommended. How confident are you right now that you could meet these guidelines if:

Start of Block: Incentive Motivation

Q8 I do physical activity because...

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
it's fun (1)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I think it's interesting (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
it makes me happy (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like to do this activity (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I find this activity stimulating (5)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I enjoy this activity (6)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to be physically fit (7)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to lose or maintain weight (8)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to better cope with stress (9)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my appearance (10)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to have more energy (11)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to define my muscles (12)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to be attractive to others (13)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my body shape (14)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I will feel ugly if I don't (15)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my cardio fitness (16)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc

I want to get better at my activity (17)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to obtain new skills (18)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my existing skills (19)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to keep up my current skill level (20)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc
I like the competition (21)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like the challenge (22)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like the excitement of participation (23)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

End of Block: Incentive Motivation

Start of Block: Social support

Q9 During the past month, my family (or members of my household) or friends or teammates:

	1 None (1)	2 Rarely (2)	3 A few times (3)	4 Often (4)	5 Very often (5)	Does not apply (6)
exercised with me (1)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
offered to exercise with me (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me helpful reminders to exercise ("Are you going to exercise tonight?") (3)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me encouragement to stick with my exercise program (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
changed their schedule so we could exercise together (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
complained about the time I spend exercising (6)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
criticized me or made fun of me for exercising. (7)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me rewards for exercising (bought me something or gave me something I like). (8)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
planned for exercise on recreational outings. (9)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
helped plan activities around my exercise. (10)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
asked me for ideas on how they can get more exercise. (11)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
talked about how much they like to exercise. (12)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

End of Block: Social support

Start of Block: Self-regulation

	1 Does not describe (1)	2 (2)	3 Describes moderately (3)	4 (4)	5 Describes Completely (5)
I never seem to have enough time to exercise (1)	0	(0	(0
Exercise is generally a high priority when I plan my schedule (2)	\bigcirc	(\bigcirc	(\bigcirc
Finding time for exercise is not difficult for me (3)	\bigcirc	(\bigcirc	(\bigcirc
I schedule all events in my life around my exercise routine (4)	\bigcirc	(\bigcirc	(\bigcirc
I schedule my exercise at specific times each week (5)	\bigcirc	(\bigcirc	(\bigcirc
I plan my weekly exercise schedule (6)	\bigcirc	(\bigcirc	(\bigcirc
When I am very busy, I don't do much exercise (7)	\bigcirc	(\bigcirc	(\bigcirc
Everything is scheduled around my exercise routine—both classes and work (8)	\bigcirc	(0	(\bigcirc
I try to exercise at the same time and same day each week to keep a routine going (9)	\bigcirc	(\bigcirc	(\bigcirc
I write my planned activity sessions in an appointment book or calendar (10)	\bigcirc	(\bigcirc	(\bigcirc

Q10 The following questions refer to how you fit exercise into your lifestyle. Please indicate the extent to which each of the statements below describes you:

End of Block: Self-regulation

Start of Block: Demographic questions

Q15 The following questions ask you to describe yourself

Q1 What is your gender?

Male (1)
Female (2)
Non-binary / third gender (3)
Prefer not to say (4)

Q2 What is your age?

Q3 Are you a person of Hispanic, Latino, or Spanish origin?

 \bigcirc Mexican, Mexican American, Chicano(a) (1)

 \bigcirc Puerto Rican (2)

O Cuban (3)

Other Hispanic, Latino(a), or Spanish origin (4)

 \bigcirc Not of Hispanic, Latino(a), or Spanish origin (5)

Q4 What is your race?

 \bigcirc American Indian or Alaska Native (1)

O Asian (2)

O Black, African American (3)

 \bigcirc Native Hawaiian or Pacific Islander (4)

 \bigcirc White, Caucasian (5)

 \bigcirc More than one race (6)

Other, please specify (7)

Never (11)	Sometimes (12)	About half the time (13)	Most of the time (14)	Always (15)
0	0	0	0	0
\bigcirc	\bigcirc	0	\bigcirc	\bigcirc
0	\bigcirc	\bigcirc	0	\bigcirc
	Never (11)	Never (11) Sometimes (12) O O O O O O O O	Never (11) Sometimes (12) About half the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of the time (13) Image: Constraint of time (13) Image: Constraint of time (13) Image: Constraint of time (13) Image: Constraint of time (13) Image: Constraint of time (13) Image: Constrain	Never (11) Sometimes (12) About half the time (13) Most of the time (14) Image: I

Q21 Please click the circle that represents your response to the following questions.

Q22 How confident are you filling out medical forms by yourself?

 \bigcirc Not at all (1)

 \bigcirc Slightly (2)

 \bigcirc Moderately (3)

O Very (4)

 \bigcirc Extremely (5)

Q18 Which program are your enrolled in?

O Harrison College of Pharmacy (1)

 \bigcirc College of Nursing (2)

O The Edward Via College of Osteopathic Medicine-Auburn Campus (3)

End of Block: Demographic questions

Start of Block: Resources and Contact Info

Q12 Some of the questions contained in the survey are related to sensitive mental health information. There is a possibility that after thinking more about the behavior you engage in and experiences you have had, you may experience psychological distress, or wonder if there is a resource available that could benefit you. A referral list of mental health providers is included <u>Referral list</u> for your use. If you need immediate assistance, please call the National Crisis Line, which is available 24 hours a day, 7 days a week free of charge at 1-800-273-TALK (8255). In addition, we invite you to search for a local counselor or therapist in your area at the following link: <u>https://therapists.psychologytoday.com/rms/</u>, noting that any cost involved in seeking treatment is at your own expense.

You have now completed the survey. Thank you for your responses. Click on the arrow below and you will be taken to a separate survey to enter your Auburn or VCOM Email address to receive a \$5 Amazon gift card. You will receive another \$10 Amazon gift card once you complete the second survey in November! Your contact information will not be linked in any way to your responses to this survey.

End of Block: Resources and Contact Info

Appendix 8. Week-8 Survey for Control Group

Start of Block: intro

Q16 Thank you for participating in this research project. Please answer the following questions. The survey takes approximately 20 minutes to complete. Upon completion of this survey, you will be given the option to provide your contact information for a \$10 Amazon gift card, to be delivered to your Auburn/VCOM email.

Q17 How many days a week do you perform muscle strengthening exercises? A strength exercise involves using your body weight or working against a resistance, such as lifting weights, working with resistance bands, climbing stairs, hill walking, cycling, dancing, push-ups, sit-ups and squats, and yoga.

▼ 0 ... 7

End of Block: intro

Start of Block: Mental wellbeing

Q6 Over the last two weeks:

	At no time (1)	Some of the time (2)	More than half of the time (3)	Most of the time (4)	All the time (5)
I have felt cheerful and in good spirits (1)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I have felt calm and relaxed (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I have felt active and vigorous (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I woke up feeling fresh and rested (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
My daily life has been filled with things that interest me (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
End of Block: Mental wellbeing					

Start of Block: Self-Efficacy

Q7 According to Physical Activity Guidelines for Americans, adults should perform at least 150 minutes a week of moderate-intensity, or 75 minutes a week of vigorous-intensity aerobic physical activity, or an equivalent combination of moderate- and vigorous-intensity activity (1 minute of vigorous activity is equal to 2 minutes of moderate activity). In addition, strengthening exercises on 2 or more days a week,

	Not at all confident (1)	Slightly confident (2)	Moderately confident (3)	Very confident (4)	Extremely confident (5)
The weather was bothering you (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You were bored by the program or activity (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You felt pain when exercising (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You had to exercise alone (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You did not enjoy it (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You were too busy with other activities (6)	0	0	\bigcirc	\bigcirc	\bigcirc
You felt tired (7)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You felt stressed (8)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
You felt depressed (9)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc

such as bodyweight exercises or resistance training, are recommended. How confident are you right now that you could meet these guidelines if:

End of Block: Self-Efficacy

Start of Block: Motivational incentives

Q8 I do physical activity because...

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
it's fun (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I think it's interesting (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
it makes me happy (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like to do this activity (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I find this activity stimulating (5)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I enjoy this activity (6)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to be physically fit (7)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to lose or maintain weight (8)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to better cope with stress (9)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my appearance (10)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to have more energy (11)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to define my muscles (12)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to be attractive to others (13)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my body shape (14)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I will feel ugly if I don't (15)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my cardio fitness (16)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

I want to get better at my activity (17)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to obtain new skills (18)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to improve my existing skills (19)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I want to keep up my current skill level (20)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like the competition (21)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like the challenge (22)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I like the excitement of participation (23)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc

End of Block: Motivational incentives

Start of Block: Social support

	1 None (1)	2 Rarely (2)	3 A few times (3)	4 Often (4)	5 Very often (5)	Does not apply (6)
exercised with me (1)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
offered to exercise with me (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me helpful reminders to exercise ("Are you going to exercise tonight?") (3)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me encouragement to stick with my exercise program (4)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
changed their schedule so we could exercise together (5)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
complained about the time I spend exercising (6)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
criticized me or made fun of me for exercising. (7)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
gave me rewards for exercising (bought me something or gave me something I like). (8)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
planned for exercise on recreational outings. (9)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
helped plan activities around my exercise. (10)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
asked me for ideas on how they can get more exercise. (11)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
talked about how much they like to exercise. (12)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q9 During the past month, my family (or members of my household) or friends or teammates:

End of Block: Social support

Start of Block: Self-regulation

	1 Does not describe (1)	2 (2)	3 Describes moderately (3)	4 (4)	5 Describes Completely (5)
I never seem to have enough time to exercise (1)	\bigcirc	(0	(0
Exercise is generally a high priority when I plan my schedule (2)	\bigcirc	(\bigcirc	(\bigcirc
Finding time for exercise is not difficult for me (3)	\bigcirc	(\bigcirc	(\bigcirc
I schedule all events in my life around my exercise routine (4)	\bigcirc	(\bigcirc	(\bigcirc
I schedule my exercise at specific times each week (5)	\bigcirc	(\bigcirc	(\bigcirc
I plan my weekly exercise schedule (6)	\bigcirc	(\bigcirc	(\bigcirc
When I am very busy, I don't do much exercise (7)	\bigcirc	(\bigcirc	(\bigcirc
Everything is scheduled around my exercise routine—both classes and work (8)	\bigcirc	(\bigcirc	(\bigcirc
I try to exercise at the same time and same day each week to keep a routine going (9)	\bigcirc	(\bigcirc	(\bigcirc
I write my planned activity sessions in an appointment book or calendar (10)	\bigcirc	(\bigcirc	(\bigcirc

Q10 The following questions refer to how you fit exercise into your lifestyle. Please indicate the extent to which each of the statements below describe you:

End of Block: Self-regulation	End	of Blo	ock:	Self-ro	egulation	1
-------------------------------	-----	--------	------	---------	-----------	---

Start of Block: Resources and Contact Info

Q12 Some of the questions contained in the survey are related to sensitive mental health information. There is a possibility that after thinking more about the behavior you engage in and experiences you have had, you may experience psychological distress, or wonder if there is a resource available that could benefit you. A referral list of mental health providers is included <u>Referral list</u> for your use. If you need immediate assistance, please call the National Crisis Line, which is available 24 hours a day, 7 days a week free of charge at 1-800-273-TALK (8255). In addition, we invite you to search for a local counselor

or therapist in your area at the following link: <u>https://therapists.psychologytoday.com/rms/</u>, noting that any cost involved in seeking treatment is at your own expense.

You have now completed the survey. Thank you for your responses. Click on the arrow below and you will be taken to a separate survey to enter your Auburn or VCOM Email address to receive a \$10 Amazon gift card. If you also completed the first survey at the beginning of this study, you will be entered into the drawing for an opportunity to win an additional \$20/\$30/\$50 Amazon gift card in approximately 4 weeks. We will contact you if you win! Your contact information will not be linked in any way to your responses to this survey.

End of Block: Resources and Contact Info

Appendix 9. Week-8 Survey for DFFC Group

Start of Block: intro

Q16 Thank you for participating in the 2022 Deans' Fit Family Challenge and this research project. Please answer the following questions. The survey takes approximately 20 minutes to complete. Upon completion of this survey, you will be given the option to provide your contact information for a \$10 Amazon gift card, to be delivered to you Auburn/VCOM email.

Q17 How many days a week do you perform muscle strengthening exercises? A strength exercise involves using your body weight or working against a resistance, such as lifting weights, working with resistance bands, climbing stairs, hill walking, cycling, dancing, push-ups, sit-ups and squats, and yoga.

▼ 0 ... 7

End of Block: intro

Start of Block: Mental wellbeing

Q6 Over the last two weeks:

	At no time (1)	Some of the time (2)	More than half of the time (3)	Most of the time (4)	All the time (5)
I have felt cheerful and in good spirits (1)					
I have felt calm and relaxed (2)					
I have felt active and vigorous (3)					
I woke up feeling fresh and rested (4)					
My daily life has been filled with things that interest me (5)					

End of Block: Mental wellbeing

Start of Block: Self-Efficacy

Q7 According to Physical Activity Guidelines for Americans, adults should perform at least 150 minutes a week of moderate-intensity, or 75 minutes a week of vigorous-intensity aerobic physical activity, or an equivalent combination of moderate- and vigorous-intensity activity (1 minute of vigorous activity is equal to 2 minutes of moderate activity). In addition, strengthening exercises on 2 or more days a week, such as bodyweight exercises or resistance training, are recommended. How confident are you right now that you could meet these guidelines if:

The weather was bothering you (1) You were bored by the program or activity (2) You felt pain when exercising (3) You had to exercise alone (4) You did not enjoy it (5) You were too busy with				Not at all confident (1)	Slightly confident (2)	Moderately confident (3)	Very confident (4)	Extremely confident (5)
You were bored by the program or activity (2) You felt pain when exercising (3) You had to exercise alone (4) You did not enjoy it (5) You were too busy with	The weather was bothering you (1)	e weather was hering you (1)	The weather was othering you (1)					
You felt pain when exercising (3) You had to exercise alone (4) You did not enjoy it (5) You were too busy with	You were bored by the program or activity (2)	vere bored by the am or activity (2)	were bored by the gram or activity (2					
You had to exercise alone (4) You did not enjoy it (5) You were too busy with	You felt pain when exercising (3)	felt pain when kercising (3)	ou felt pain when exercising (3)					
You did not enjoy it (5) You were too busy with	You had to exercise alone (4)	had to exercise alone (4)	ou had to exercise alone (4)					
You were too busy with	You did not enjoy it (5)	d not enjoy it (5)	did not enjoy it (5)				
other activities (b)	You were too busy with other activities (6)	ere too busy with er activities (6)	were too busy wit ther activities (6)	n				
You felt tired (7)	You felt tired (7)	u felt tired (7)	You felt tired (7)					
You felt stressed (8)	You felt stressed (8)	felt stressed (8)	ou felt stressed (8)					
You felt depressed (9)	You felt depressed (9)	elt depressed (9)	1 felt depressed (9					

End of Block: Self-Efficacy

Start of Block: Motivational incentives

Q8 I do physical activity because...

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
it's fun (1)					
I think it's interesting (2)					
it makes me happy (3)					
I like to do this activity (4)					
I find this activity stimulating (5)					
I enjoy this activity (6)					
I want to be physically fit (7)					
I want to lose or maintain weight (8)					
I want to better cope with stress (9)					
I want to improve my appearance (10)					
I want to have more energy (11)					
I want to define my muscles (12)					
I want to be attractive to others (13)					
I want to improve my body shape (14)					
I will feel ugly if I don't (15)					
I want to improve my cardio fitness (16)					
I want to get better at my activity (17)					
I want to obtain new skills (18)					
I want to improve my existing skills (19)					
I want to keep up my current skill level (20)					

I like the competition (21) I like the challenge (22) I like the excitement of participation (23)

End of Block: Motivational incentives

Start of Block: Social support

Q9 During the past month, my family (or members of my household) or friends or teammates:

	1 None (1)	2 Rarely (2)	3 A few times (3)	4 Often (4)	5 Very often (5)	Does not apply (6)
exercised with me (1)						
offered to exercise with me (2)						
gave me helpful reminders to exercise ("Are you going to exercise tonight?") (3)						
gave me encouragement to stick with my exercise program (4)						
changed their schedule so we could exercise together (5)						
complained about the time I spend exercising (6)						
criticized me or made fun of me for exercising. (7)						
gave me rewards for exercising (bought me something or gave me something I like). (8)						
planned for exercise on recreational outings. (9)						
helped plan activities around my exercise. (10)						
asked me for ideas on how they can get more exercise. (11)						
talked about how much they like to exercise. (12)						

End of Block: Social support

Start of Block: Self-regulation

Q10 The following questions refer to how you fit exercise into your lifestyle. Please indicate the extent to which each of the statements below describes you:

	1 Does not describe (1)	2 (2)	3 Describes moderately (3)	4 (4)	5 Describes Completely (5)
I never seem to have enough time to exercise (1)					
Exercise is generally a high priority when I plan my schedule (2)					
Finding time for exercise is not difficult for me (3)					
I schedule all events in my life around my exercise routine (4)					
I schedule my exercise at specific times each week (5)					
I plan my weekly exercise schedule (6)					
When I am very busy, I don't do much exercise (7)					
Everything is scheduled around my exercise routine—both classes and work (8)					
I try to exercise at the same time and same day each week to keep a routine going (9)					
I write my planned activity sessions in an appointment book or calendar (10)					

	C	TO L - L -	C-1C	1 - 4 •
нла	OT	KIOCK	seit r	eomanon
	UL.	DIUCIN		<i>c</i> <u>c</u> ululul

Start of Block: Participants' experiences DFFC (delete for control group)

The competitive nature of the Challenge (1)
Weekly activity minute logging (2)
Weekly motivational email (3)
Working with my teammate (4)
Group fitness activities (5)
Prizes/awards throughout the Challenge (6)
Announcement of previous week's winners and their data (7)
Marketing materials about physical activity (8)
Other, please describe (9)

Q11 What aspects of the Deans' Fit Family Challenge motivate you to exercise?

Q12 Please describe an aspect of 2022 Deans' Fit Family Challenge you enjoyed the most

Q13 What are the barriers you encounter while doing or scheduling your exercise during 2022 Deans' Fit Family Challenge?

Insufficient time (1)
Lack of skills and knowledge (2)
Lack of motivation (3)
Lack of support (4)
Lack of access to appropriate facilities (5)
Unsafe neighborhoods (6)
Poor weather (7)
Negative experiences while engaging in physical activity (8)
Other, please describe (9)

Q14 Please describe one thing/aspect of the 2022 Deans' Fit Family Challenge you would change to make it better.

End of Block: Participants' experiences DFFC (delete for control group)

Start of Block: Acceptability (delete if they in control group)

Q25 Overall, how satisfied were you with the 2022 Deans' Fit Family Challenge

Very dissatisfied (1) Somewhat dissatisfied (2) Neither satisfied nor dissatisfied (3) Somewhat satisfied (4) Very satisfied (5)

Q16 How likely are you to participate in next year's 2023 Deans' Fit Family Challenge if you are still eligible?

Extremely unlikely (1) Somewhat unlikely (2) Neither likely nor unlikely (3) Somewhat likely (4) Extremely likely (5)

End of Block: Acceptability (delete if they in control group)

Start of Block: Resources and Contact Info

Q12 Some of the questions contained in the survey are related to sensitive mental health information. There is a possibility that after thinking more about the behavior you engage in and experiences you have had, you may experience psychological distress, or wonder if there is a resource available that could benefit you. A referral list of mental health providers is included <u>Referral list</u> for your use. If you need immediate assistance, please call the National Crisis Line, which is available 24 hours a day, 7 days a week free of charge at 1-800-273-TALK (8255). In addition, we invite you to search for a local counselor or therapist in your area at the following link: <u>https://therapists.psychologytoday.com/rms/</u>, noting that any cost involved in seeking treatment is at your own expense.

You have now completed the survey. Thank you for your responses. Click on the arrow below and you will be taken to a separate survey to enter your Auburn or VCOM Email address to receive a \$10 Amazon gift card. If you also completed the first survey at the beginning of this study, you will be entered into the drawing for an opportunity to win an additional \$20/\$30/\$50 Amazon gift card in approximately 4 weeks. We will contact you if you win! Your contact information will not be linked in any way to your responses to this survey.

End of Block: Resources and Contact Info
Start of Block: intro

Q16 Thank you for participating in the 2022 Deans' Fit Family Challenge and this research project. Please answer the following questions. The survey takes approximately 20 minutes to complete. Upon completion of this survey, you will be given the option to provide your contact information for a \$10 Amazon gift card, to be delivered to you Auburn/VCOM email.

Q17 How many days a week do you perform muscle strengthening exercises? A strength exercise involves using your body weight or working against a resistance, such as lifting weights, working with resistance bands, climbing stairs, hill walking, cycling, dancing, push-ups, sit-ups and squats, and yoga.

▼ 0 ... 7

End of Block: intro

Start of Block: Mental wellbeing

Q6 Over the last two weeks:

	At no time (1)	Some of the time (2)	More than half of the time (3)	Most of the time (4)	All the time (5)
I have felt cheerful and in good spirits (1)					
I have felt calm and relaxed (2)					
I have felt active and vigorous (3)					
I woke up feeling fresh and rested (4)					
My daily life has been filled with things that interest me (5)					

End of Block: Mental wellbeing

Start of Block: Self-Efficacy

Q7 According to Physical Activity Guidelines for Americans, adults should perform at least 150 minutes a week of moderate-intensity, or 75 minutes a week of vigorous-intensity aerobic physical activity, or an equivalent combination of moderate- and vigorous-intensity activity (1 minute of vigorous activity is equal to 2 minutes of moderate activity). In addition, strengthening exercises on 2 or more days a week, such as bodyweight exercises or resistance training, are recommended. How confident are you right now that you could meet these guidelines if:

	Not at all confident (1)	Slightly confident (2)	Moderately confident (3)	Very confident (4)	Extremely confident (5)
The weather was bothering you (1)					
You were bored by the program or activity (2)					
You felt pain when exercising (3)					
You had to exercise alone (4)					
You did not enjoy it (5)					
You were too busy with other activities (6)					
You felt tired (7)					
You felt stressed (8)					
You felt depressed (9)					

End of Block: Self-Efficacy

Start of Block: Motivational incentives

Q8 I do physical activity because...

	Strongly disagree (1)	Somewhat disagree (2)	Neither agree nor disagree (3)	Somewhat agree (4)	Strongly agree (5)
it's fun (1)					
I think it's interesting (2)					
it makes me happy (3)					
I like to do this activity (4)					
I find this activity stimulating (5)					
I enjoy this activity (6)					
I want to be physically fit (7)					
I want to lose or maintain weight (8)					
I want to better cope with stress (9)					
I want to improve my appearance (10)					
I want to have more energy (11)					
I want to define my muscles (12)					
I want to be attractive to others (13)					
I want to improve my body shape (14)					
I will feel ugly if I don't (15)					
I want to improve my cardio fitness (16)					
I want to get better at my activity (17)					
I want to obtain new skills (18)					
I want to improve my existing skills (19)					
I want to keep up my current skill level (20)					

I like the competition (21) I like the challenge (22) I like the excitement of participation (23)

End of Block: Motivational incentives

Start of Block: Social support

Q9 During the past month, my family (or members of my household) or friends or teammates:

	1 None (1)	2 Rarely (2)	3 A few times (3)	4 Often (4)	5 Very often (5)	Does not apply (6)
exercised with me (1)						
offered to exercise with me (2)						
gave me helpful reminders to exercise ("Are you going to exercise tonight?") (3)						
gave me encouragement to stick with my exercise program (4)						
changed their schedule so we could exercise together (5)						
complained about the time I spend exercising (6)						
criticized me or made fun of me for exercising. (7)						
gave me rewards for exercising (bought me something or gave me something I like). (8)						
planned for exercise on recreational outings. (9)						
helped plan activities around my exercise. (10)						
asked me for ideas on how they can get more exercise. (11)						
talked about how much they like to exercise. (12)						

End of Block: Social support

Start of Block: Self-regulation

Q10 The following questions refer to how you fit exercise into your lifestyle. Please indicate the extent to which each of the statements below describes you:

	1 Does not describe (1)	2 (2)	3 Describes moderately (3)	4 (4)	5 Describes Completely (5)
I never seem to have enough time to exercise (1)					
Exercise is generally a high priority when I plan my schedule (2)					
Finding time for exercise is not difficult for me (3)					
I schedule all events in my life around my exercise routine (4)					
I schedule my exercise at specific times each week (5)					
I plan my weekly exercise schedule (6)					
When I am very busy, I don't do much exercise (7)					
Everything is scheduled around my exercise routine—both classes and work (8)					
I try to exercise at the same time and same day each week to keep a routine going (9)					
I write my planned activity sessions in an appointment book or calendar (10)					

Fnd	of	RIC	ok	C.	Jf_	ro	ml	oti	on
L'IIU	UL.	DIU	JUK.	00		16	gui	au	UII

Start of Block: Participants' experiences DFFC

The competitive nature of the Challenge (1)
Weekly activity minute logging (2)
Weekly motivational email (3)
Working with my teammate (4)
Group fitness activities (5)
Prizes/awards throughout the Challenge (6)
Announcement of previous week's winners and their data (7)
Marketing materials about physical activity (8)
Other, please describe (9)

Q11 What aspects of the 2022 Deans' Fit Family Challenge motivate you to exercise?

Q12 Please describe an aspect of 2022 Deans' Fit Family Challenge you enjoyed the most

Q13 What are the barriers you encounter while doing or scheduling your exercise during 2022 Deans' Fit Family Challenge?

Insufficient time (1)
Lack of skills and knowledge (2)
Lack of motivation (3)
Lack of support (4)
Lack of access to appropriate facilities (5)
Unsafe neighborhoods (6)
Poor weather (7)
Negative experiences while engaging in physical activity (8)
Other, please describe (9)

Q14 Please describe one thing/aspect of the 2022 Deans' Fit Family Challenge you would change to make it better.

End of Block: Participants' experiences DFFC

Start of Block: Acceptability

Q25 Overall, how satisfied were you with the 2022 Deans' Fit Family Challenge

Very dissatisfied (1) Somewhat dissatisfied (2) Neither satisfied nor dissatisfied (3) Somewhat satisfied (4) Very satisfied (5)

Q16 How likely are you to participate in next year's 2023 Deans' Fit Family Challenge if you are still eligible?

Extremely unlikely (1) Somewhat unlikely (2) Neither likely nor unlikely (3) Somewhat likely (4) Extremely likely (5)

End of Block: Acceptability

Start of Block: Satisfaction SMS

Unhelpful Very unhelpful Neutral (3) Helpful (4) Very helpful (5) (1)(2)motivating me to start exercising (1) motivating me to do more exercise (2) helping me achieve the 150minute/week goal (3)

Q25 During the Challenge, how would you rate the text messages you received in terms of the activities below?

	Extremely dissatisfied (1)	Somewhat dissatisfied (2)	Neither satisfied nor dissatisfied (3)	Somewhat satisfied (4)	Extremely satisfied (5)
Frequency of text messages received (1)					
Satisfaction with the understandability of the content of the text messages (2)					
Satisfaction with motivational wording in the text messages (3)					
Overall satisfaction with the text messages (4)					

Q24 How satisfied are you with the physical activity text messages you received during the Challenge?

End of Block: Satisfaction SMS

Start of Block: Resources and Contact Info

Q12 Some of the questions contained in the survey are related to sensitive mental health information. There is a possibility that after thinking more about the behavior you engage in and experiences you have had, you may experience psychological distress, or wonder if there is a resource available that could benefit you. A referral list of mental health providers is included <u>Referral list</u> for your use. If you need immediate assistance, please call the National Crisis Line, which is available 24 hours a day, 7 days a week free of charge at 1-800-273-TALK (8255). In addition, we invite you to search for a local counselor or therapist in your area at the following link: <u>https://therapists.psychologytoday.com/rms/</u>, noting that any cost involved in seeking treatment is at your own expense.

You have now completed the survey. Thank you for your responses. Click on the arrow below and you will be taken to a separate survey to enter your Auburn or VCOM Email address to receive a \$10 Amazon gift card. If you also completed the first survey at the beginning of this study, you will be entered into the drawing for an opportunity to win an additional \$20/\$30/\$50 Amazon gift card in approximately 4 weeks. We will contact you if you win! Your contact information will not be linked in any way to your responses to this survey.

Appendix 11. Week -0 Contact Information Survey

Q1 Please enter your Auburn or VCOM Email address to receive a \$5 Amazon gift card. Your contact information will not be linked in any way to your responses to this survey.

Appendix 12. Week -8 Contact Information Survey

Q1 Please enter your Auburn or VCOM Email address to receive a \$10 Amazon gift card. Your contact information will not be linked in any way to your responses to this survey.

Appendix 13. Referral List for Mental Wellbeing

Referral List

If you have experienced distress as a result of your participation in this study, a referral list of mental health providers is included below for your use. (Please remember that any cost in seeking medical assistance is at your own expense.)

On Campus Services:

Provider	Phone Number	Cost/Hour
Auburn University Student Counseling and Psychological Services (SCPS)	334-844-5123	Services are no charge for students. Up to 10 sessions allotted per academic year. Crisis walk-in/call-in service during business hours; crisis call-in service after-hours &weekends.
Auburn University Psychological Services Center (AUPSC)	334-844-4889	First appointment \$80 Subsequent appointments \$30-\$60 based on income, or \$30 for AU students
Auburn University Marriage & Family Center	334-844-4478	First appointment \$20 Subsequent appointments \$50 based on income or \$20 for AU Students
Tiger Education Screening Intervention	334-844-1311	No charge for self-referred AU students \$125 for mandated referrals
Auburn University Public Safety	334-844-8888	
Auburn University Medical Clinic	334-844-4416	

Off Campus Services:

Provider	Phone Number	Cost/Hour
Clinical Psychologists, PC	334-821-3350	\$150, first appointment \$130, other appointments Insurance Accepted
Auburn Psychology Group, LLC	334-887-4343	\$160, first appointment\$140, other appointmentsInsurance Accepted
East Alabama Mental Health	800-815-0630 334-742-2877 334-742-2700	Help available 24/7 \$8-\$80 based on income
East Alabama Alcoholism and Substance Abuse Services	334-742-2130	Outpatient and court-ordered referrals; call for a quote

Emergency Services:

Provider	Phone Number	Cost/Hour
National Suicide Prevention Lifeline	1-800-273-TALK	No charge, This lifeline is FREE, confidential and
	(8255)	always available.
Emergency Services	911	Dependent on insurance
East Alabama Medical Center	334-528-1150	
Emergency Department		